

	2 LEVEL II
AD A 0 97 76 0	PROL RAM CRITERIA SPECIFICATIONS DOCUMENT. COMPUTER PROGRAM TWDA FOR DESIGN AND ANALYSIS OF INVERTED-T RETAINING WALLS AND FLOODWALLS. Victor M/Agostinelli Victor M/Agostinelli William A. /Price U. S. Army Engineer Division, Lower Mississippi Valley P. O. Box 80, Vicksburg, Miss. 31980 and William A. Price, Carl E. Pace U. S. Army Engineer Waterways Experiment Station P. O. Box 631, Vicksburg, Miss. 39180
COPY	Approved For Public Release; Distribution Unlimited Prepared for U. S. Army Engineer Division, Lower Mississippi Valley P. O. Box 80, Vicksburg, Miss. 39180
te faith	
	038/00

038/00 81 4 13 266 met Destroy this report when no longer needed. Do not return it to the originator.

The findings in this report are not to be construed as an official Department of the Army position unless so designated. by other authorized documents.

> The contents of this report are not to be used for advertising, publication, or promotional purposes. Citation of trade names does not constitute an official endorsement or approval of the use of such commercial products.

PREFACE

This document presents criteria prepared as the basis for developing TWDA, a computer program for design and analysis of inverted-T retaining walls and floodwalls. Development of the program is a joint effort of the Computer-Aided Structural Design (CASD) Project of the U. S. Army Engineer Division, Lower Mississippi Valley (LMVD), and of the Computer-Aided Structural Engineering (CASE) Project of the Office, Chief of Engineers, U. S. Army (OCE).

Mr. William A. Price, Chief, Computer-Aided Design Group (CADG), Automatic Data Processing (ADP) Center, U. S. Army Engineer Waterways Experiment Station (WES), provided the overall design of the program and led the program development team.

Engineering criteria for TWDA were provided by the members of LMVD's CASD Committee and of the CASE Task Group on T-Walls and by other Corps personnel:

LMVD's CASD Committee

Victor M. Agostinelli, LMVD (Chairman) Arvis R. Dennis, Vicksburg District Clifton C. Hamby, Vicksburg District Sefton B. Lucas, Memphis District Joseph Barber, Memphis District James G. Bigham, New Orleans District Joseph V. Milliorn, formerly with the New Orleans District James J. Smith, St. Louis District Thomas J. Mudd, St. Louis District James Cronin, St. Louis District Carlton Smith, St. Louis District

CASE Task Group on T-Walls

Victor M. Agostinelli, LMVD (Chairman) Terry C. Cox, LMVD Alvis Eikstrems, North Atlantic Division Stacey Anastos, North Atlantic Division Joseph V. Milliorn, formerly with the New Orleans District Raymond Veselka, Galveston District

OCE (DAEN-CWE-DS) Personnel

Keith O. O'Donnell, former Chief, Structural Engineering Section (retired)

i

OCE (DAEN-CWE-DS) Personnel (Continued)

Donald R. Dressler Lucian G. Guthrie

A

Other Corps Personnel

William A. Price, WES Carl E. Pace, WES James D. Wall, South Atlantic Division

This document was compiled by Mr. Agostinelli, Mr. Price, and Dr.

Pace. It was published for LMVD.

A basic user's guide, a user's reference manual, and a validation report will also be published on TWDA by WES.

Acce	ssion For					
NTIS	NTIS GRA&I					
DITC	TAB					
Usan	benurse	ā l				
Just.	Cleation					
	·					
B						
Distribution/						
Availability Codes						
	Avail and	'or				
Dist Special						
		1				
H						
//		1				

CONTENTS

				Page
PREI	FACE .			i
MAR	GINAL	NOTES	•••••••••••••••••••••••••••••••••••••••	viii
1.	PROGE	RAM PURI	POSE AND ORGANIZATION	1-1
	1.1	PURPOSI	E OF PROGRAM TWDA	1-1
		1.1.1	Computer-Aided Design	1-1
		1.1.2	Criteria Document Scope (Structural, not Soils)	1-1
	1.2	ORGANI	ZATION	1-1
		1.2.1	Program Structure	1-1
		1.2.2	Data Entry Procedure	1-1
		1.2.3	Data Review Procedure	1-2
		1.2.4	Restart Capability	1-3
		1.2.5	Volume of Printout	1-3
		1.2.6	Calculation Modules	1-3
2.	DATA			2-1
	2.1	GENERA	L	2-1
	2.2	BASIC I	DATA	2-1
		2.2.1	Criteria Data	2-1
		2.2.2	Wall Basic Data for Design	2-1
		2.2.3	Soils	2-1
		2.2.4	Loads	2-1
	2.3	LOAD CA	ASE DATA	2-1
		2.3.1	Possible Factors for Describing Any ONE Load Case	2-1
		2.3.2	Typical Application of Load Cases	2-4
3.	STAB	ILITY .		3-1
	3.1	INTROD	UCTION TO STABILITY CRITERIA	3-1
	3.2	CONFIG	URATIONS	3-1
		3.2.1	Design Load Configuration References	3-1
		3.2.2	Base Contact Configurations	3-1
	3.3	GENERA	L STABILITY CAPABILITIES	3-3
		3.3.1	Wall Dimension Variations	3-3

······

				Page
		3.3.2	Quantities For Cost Comparison	3-3
		3.3.3	General Criteria	3-6
4	. APP	LIED PRE	SSURES	4-1
	4.1	GENERA	L BASIS OF EARTH PRESSURES	4-1
	4.2	EARTH	PRESSURE APPLICATION LOCATION	4-1
	4.3	METHOD	OF COMPUTATION OF EARTH PRESSURES	4-1
		4.3.1	Active Earth Pressure	4-1
		4.3.2	Fully Passive Pressures	4-3
		4.3.3	At Rest Pressures	4-3
		4.3.4	Forces Along Soil-Stem Interface	4-3
	4.4	RESIST	ING FORCE DISCUSSION FOR OVERTURNING AND	
		SLIDIN	G	4-4
		4.4.1	Precaution	4-4
		4.4.2	Distribution for Overturning Calculations	4-4
		4.4.3	Limiting Value for Sliding	4-5
		4.4.4	Reduction of Excessive Required Passive Pressure	4-5
		4.4.5	An Alternate Resisting Strut Force	4-5
	4.5	HYDROS	TATIC PRESSURES	4-7
	4.6	SURCHA	RGE PRESSURES	4-7
		4.6.1	Vertical Pressures	4-7
		4.6.2	Horizontal Pressures	4-8
	4.7	EFFECT WEIGHT	OF SEEPAGE PRESSURES ON EFFECTIVE EARTH UNIT	4-9
5	. OVE	RTURNING	STABILITY	5-1
	5.1	GENERA	L	5-1
		5.1.1	Terminology	5 -1
		5.1.2	Stability Requirements	5-1
		5.1.3	Horizontal Pressures	5-1
	5.2	ANALYS	IS PROCEDURE	5-1
		5.2.1	The Moment Center	5-1
		5.2.2	Force Summations	5-1
		5.2.3	Moment Summation	5-1
		5.2.4	Sample Calculations	5-1

Line Rain

ويعذرهم

2

					Page
	5.3	DESIGN	PROCEDUR	Ε	5-1
		5.3.1	The Basi	c Program Procedure	5-1
		5.3.2	Verifica	tion	5-2
		5.3.3	Setting	Desired Values	5-2
		5.3.4	Bearing	Pressure Reference	5-2
	5.4	RESULT	ANT RATIO	<i>.</i>	5-2
		5.4.1	Definiti	on	5-2
		5.4.2	Limiting	Values	5-2
		5.4.3	Relation	ship to Percent Effective Area	5-3
	5.5	UPLIFT			5-3
5.	SLID	ING			6-1
	6.1	AVAILA	BLE METHO	DS OF ANALYSIS	6-1
		6.1.1	Shear Fr	iction Method	6-1
		6.1.2	Allowabl	e Strength Equilibrium Method (I)	6-1
		6.1.3	Allowabl	e Strength Equilibrium Method (II)	6-1
		6.1.4	Modified	Shear Friction Sliding Method	6-1
	6.2	DISCUS	SION OF A	NALYSIS METHODS	6-1
		6.2.1	Shear Fr	iction Method (ETL 1110-2-184)	6-1
			6.2.1.1	Resistance Force for Uphill Sliding	6-2
			6.2.1.2	Resisting Force for Downhill Sliding	6-2
			6.2.1.3	Safety Factor	6-3
			6.2.1.4	Apparent Deficiencies in ETL 1110-2-184 Shear Friction Formulas	6-3
			6.2.1.5	Passive Resistance of the Wedge Adjacent to the Wall	6-3
			6.2.1.6	Values for the Minimum Shear Friction Safety Factor	6-4
		6.2.2	Allowabl	e Strength Equilibrium Method (I)	6-4
		6.2.3	Allowab1	e Strength Equilibrium Method (II)	6-6
	6.3	DISCUS	SIONS COM	MON TO ALL SLIDING METHODS	6-6
		6.3.1	Assumed	Failure Surfaces	6-6
		6.3.2	Effect D Compress	ue to Some of the Base Not Being in ion With the Foundation	6-7
			6.3.2.1	Portion of Base in Compression With Foundation (Percent Effective Base)	6-7

Server and the server

v

			Page
		6.3.2.2 Uplift	6-8
		6.3.2.3 Coefficient of Friction ϕ and Cohesion	
_		c to be Used in the Analysis	6-9
7.	BEAR	RING PRESSURES	7-1
	7.1	CALCULATION OF BEARING PRESSURES	7-1
		/.1.1 The Procedure	7-1
		7.1.2 Entire Base in Compression	7-1
		7.1.3 Part of Base in Compression	7-1
	7.2	MAXIMUM ALLOWABLE BEARING PRESSURES	7-3
		7.2.1 Selection of Maximum Allowable Pressure	7-3
		7.2.2 Data Structure	7-3
		7.2.3 Method	7-3
8.	EART	HQUAKE ANALYSIS	8-1
	8.1	INTRODUCTION	8-1
		8.1.1 Seismic Coefficient	8-1
		o.1.2 T-Wall (ER 1110-2-1806)	8-1
	8.2	SELECTION OF SEISMIC COEFFICIENT	8-1
	8.3	LOCAL SOIL CONDITIONS	8-2
	8.4	INERTIA FORCE OF WALL	8-2
	8.5	LATERAL COHESIONLESS EARTH PRESSURES DUE TO EARTHQUAKES	8-3
		8.5.1 Active Earth Pressure Conditions	8-3
		8.5.2 Passive Earth Pressure Conditions	8-5
		8.5.3 At-Rest Earth Pressure Conditions	8-7
	8.6	LATERAL COHESIVE EARTH PRESSURES DUE TO EARTHQUAKES	8-8
	8.7	WATER PRESSURE DUE TO EARTHQUAKES	8-8
		8.7.1 Method	8-8
		8.7.2 Westergaard Theory	8-8
9.	STRU	CTURAL DESIGN CRITERIA	9-1
	9.1	GENERAL	9-1
		9.1.1 Critical Sections and Alternate Pressure Diagrams	9-1
		9.1.2 The Method of Analysis	9-1
		9.1.2.1 Working Stress Design	9-1

ĥ

		Page
	9.1.2.2 Ultimate Stress Design	9-1
9.	1.3 Pressures for Structural Design	9-2
9.2 RE	INFORCING STEEL DETAILS	9-2
9.	2.1 The Primary Reference	9-2
9.	2.2 Default Values for Concrete Clear Cover	9-2
9.	2.3 Compressive Reinforcement	9-2
9.	2.4 Placement of Reinforcing Steel	9-2
9.3 ST	EM LOADINGS	9-3
9.	3.1 General	9-3
9.	3.2 Alternate Loading	9-3
EXHIBIT A:	COMPARISON OF COULOMB AND INCREMENTAL WEDGE EARTH	_
	PRESSURE VALUES	A-1
	Procedures Followed	A-1
	Discussion of Comparisons	••• A-2
	Addendum A: Procedure for Obtaining Coulomb Active Earth Pressure at a Given Depth.	A-4
	Addendum B: Lateral Earth Pressure on Walls Using Theory of Elasticity	A-6
	Addendum C: Active Force for One Wedge (Generic)	A-7
	Addendum D: Pressure Comparisons	A-9
	Addendum E: Calculation Worksheets for Addendum D	A-15
	Addendum F: Errors Inherent in Trial Wedge Method	A-63
EXHIBIT B:	DERIVATION OF EQUATIONS FOR SLIDING	B-1
EXHIBIT C:	EQUATIONS FOR EARTH PRESSURE UNDER A TRAPEZOIDAL BAS	E. C-1
EXHIBIT D:	COMPUTATION OF AT REST EARTH PRESSURE	D-1
EXHIBIT E:	FORMULAS FOR WORKING STRESS DESIGN OF CANTILEVER SLABS WITH AXIAL FORCE (ACCORDING TO APPENDIX B OF ACI 318-77)	E-1
EXHIBIT F:	FORMULAS FOR ULTIMATE STRENGTH DESIGN OF NONHYDRAULIC STRUCTURES WITH CANTILEVER SLABS WITH AXIAL FORCE (ACCORDING TO ACI STANDARD NO. 318-77)	F-1
EXHIBIT G:	COMPARISON OF BOUSSINESQ AND BOWLES SOLUTIONS FOR PRESSURES ON RETAINING WALLS DUE TO SURCHARGE LOADS.	G-1
EXHIBIT H:	SAMPLE RETAINING WALL SHEAR FRICTION SLIDING ANALYSI	S. H-1
EXHIBIT I:	SAMPLE FLOODWALL SLIDING ANALYSIS	I-1

vii

		Page
EXHIBIT J:	SAMPLE FLOODWALL SLIDING ANALYSIS	J-1
EXHIBIT K:	SAMPLE OVERTURNING ANALYSIS FOR FLOODWALLS AND RETAINING WALLS	K-1
EXHIBIT L:	PASSIVE RESISTANCE FOR T-WALLSEQUIVALENT RELATION FOR COULOMB THEORY AND ETL 1110-2-184	L-1

MARGINAL NOTES

Special marginal notes, in the form of a letter or group of letters enclosed in brackets, are used to identify the principal sources of the key items of criteria. These symbols include:

- [F] EM 1110-2-2501, "Flood Walls," Jan 1948, with Change 3, 18 Jun 1962.
- [FD] Draft manual for floodwalls (EC 1110-2-156, 17 Jun 1975).
- [R] EM 1110-2-2502, "Retaining Walls," 29 May 1961, with Change 3, 25 Jan 1965.
- [D] EM 1110-2-2200, "Gravity Dam Design," 25 Sep 1958.
- [EL] ETL 1110-2-22, "Lock Gravity Walls," 19 Apr 1967.
- [ED] ETL 1110-2-184, "Gravity Dam Design--Stability."
- [E] ER 1110-2-1806, "Earthquake Design . . . Dams," 30 Apr 1977.
- [ACI] ACI 318-71, "ACI Building Code . . . Concrete," with 1976 Supplement.
- [WS] EM 1110-2-2101, "Working Stresses for Structural Design," 1 Nov 1963.
- [HSR] EM 1110-2-2103, "Reinforcement . . . for Hydraulic Structures," 21 May 1971.
- [ChS] OCE specifications (guidance from DAEN-CWE-DS personnel).

1. PROGRAM PURPOSE AND ORGANIZATION

1.1 \ , PURPOSE OF PROGRAM TWDA

1.1.1 Program TWDA is to be a computer-aided structural design system for analysis and/or design of inverted-T cantilever walls founded on earth or rock. Multiple load cases will allow the wall to act as a floodwall or a retaining wall.

1.1.2 This program criteria specifications document is intended for use by structural engineers. The computer program that these criteria are for does not attempt to establish any soils design criteria; such data must be entered by the user after consultation with soils design engineers. There are no default values for soils criteria parameters, except as provided in the engineering manuals for structural design. 1.2 ORGANIZATION

1.2.1 <u>Structure</u> - The program will be a series of design or analysis modules,* each performing one specific step in the design or analysis process. These modules will be callable, in any logical sequence, from an executive command phase.** While in the executive phase, the user may call various procedures for data entry, data review, saving the current design status, restoring from an old status save, etc. This is illustrated in Figure 1-1.

1.2.2 <u>Data Entry</u> - The data entry procedure will be similar to that for program TGDA, + except that the data phase may be incorporated into the command phase instead of being separate as in TGDA. Features will include:

a. Data are entered by naming the group and listing the values in that group, all on one line.

b. Default values may be requested by entering the letter "D" instead of a numerical value.

^{*} A module is a subprogram that is controlled as one unit and that performs one complete aspect of the purpose of the entire program.

^{**} The executive phase of this program is the central core of the user's flow of control. The user may enter data or start a module while in the executive phase.

TGDA (three-girder tainter gate design/analysis) is a computer program (713-F3-R0-022) developed for LMVD's CASD Committee in 1976.



Figure 1-1. BASIC PROGRAM FLOWCHART

c. Values to be calculated will be identified to the program by typing the letter "C" instead of entering a value.

d. A value that is to be left unchanged from its previous state will be identified to the program by typing the letter "S."

e. The program will look for illogical and inconsistent data and will identify such items to the user for correction or use anyway.

f. The current status of items of input data or of all data values can be reviewed.

g. Multiple-level prompting is provided as in program TGDA, except that the minimum level will be less wordy than in program TGDA, for the more experienced users.

Thus, the program will accept several sets of input data, where the following sets contain only the changes to the data comprising the preceding sets. Repetitive data will remain unchanged.

1.2.3 Data Review - Data review will be available in two ways:

a. Input Data Review will be done as in the data input phase in program TGDA with the LOOK command.

b. <u>Default Value Review</u> will be done in a separate module. Unless reviewed with this option, default values will be set automatically by the user's selection of:

(1) Floodwall or retaining wall criteria.

(2) Hydraulic or nonhydraulic structure criteria.

Making the review of default values optional is expected to enable the experienced user to simplify and expedite his preliminary designs. In any case, the values will be printed out in the report file. The combination of a nonhydraulic floodwall, being illogical, will be rejected. Default values will always be taken from OCE publications; nonstandard values set by the user will be so labeled in the report file and verified interactively.

1.2.4 Restart Capability

a. In addition to the user-controlled SAVE files, the program will use an automatic UPDATE file that is reset after the completion of a command or a calculation module.

b. The RESTart command will restart the program from an old update or saved file.

1.2.5 Volume Of Printout - Printout will be of two types:

a. The printout to the user's time-sharing terminal will be restricted to the minimum needed for the user to make his decisions.

b. A full report of calculations made will be written to a report file that can be listed at a time-sharing terminal and/or sent to the high speed printer in the user's District office ADP Center. 1.2.6 <u>Calculation Modules</u> - A list of the major calculation modules includes:

a. SA - <u>Stability</u> analysis <u>Active</u> pressures for overturning and sliding, calculated along a vertical plane at end of heel.

- (1) Coulomb's equations plus surcharge pressure equations assuming elastic soil.
- (2) Incremental wedge methods (see paragraph 4.3.1b).
- (3) As imputted.

b. FA - Foundation stability <u>A</u>nalysis of completely defined wall (overturning, sliding, and bearing); uses module SA as needed.

c. FD - Foundation stability Design; uses modules SA and FA as needed.

d. SP - <u>Stem Pressures</u> for structural analysis. Same basis as module SA, except that the pressures are calculated at the stem face instead of at the end of the heel. This is for structural analysis of the stem. Heel, toe and key slabs will use pressures based on the stability analysis from modules FA or FD, as described in paragraph 9.1.1.

- e. WA Working stress structural Analysis.
- f. WD Working stress structural Design.
- g. UA Ultimate strength structural Analysis.
- h. UD Ultimate strength structural Design.

2. DATA

2.1 GENERAL

Data will be of two types, basic data and load case data. Basic data will be used as common to all load cases unless overridden by data for a particular load case. Load case data will consist of values applicable to only the one load case. Basic data will also include unchanging data such as wall dimensions.

2.2 BASIC DATA

- 2.2.1 Criteria Selection
 - (1) Floodwall or retaining wall?
 - (2) Hydraulic or nonhydraulic structure?

2.2.2 <u>Wall</u> (basic data for design, described in paragraph 3.3; may be set by groups for different wall types). Major items are listed below:

(1) Top of stem elevation and minimum thickness.

- (2) Toe-side batter of stem.
- (3) Heel-side top panel height and batter of stem.

(4) Heel-side bottom panel batter of stem.

(5) Minimum base slab thickness.

- (6) Bottom of toe elevation or range of values.*
- (7) Toe width or stem ratio.
- (8) Base width, range of values.*
- (9) Base slope, range of values.*
- (10) Key depth, maximum value.*

(11) Key batter, toe-side face.

(12) Key location indicator (0 if at heel, 1 if at stem).

2.2.3 <u>Soils</u> data as illustrated in Figure 3-1 and described in Section 4.

2.2.4 Loads common to all load cases (except ones for which value(s) are reset in load case data), as described in paragraph 2.3.

2.3 LOAD CASE DATA (for each individual load case)

2.3.1 <u>Possible Factors for Describing Any ONE Load Case</u> (in addition to or in place of basic data):

* Varies between limits set by user.

- a. Water
 - Water elevations over heel and over toe, unit weight of water (default = 62.5). See paragraph 3.2.1e for illustration of elevation.
 - Seepage pressure according to descriptions in paragraph 3.3.3f.
- b. Earth
 - Earth geometry over heel, at stem, and over toe if different from basic data. Also soils properties data if different from soils basic data input.
 - (2) Earth pressures on wall (a) calculated from the earth elevations and K-value Coulomb theory, (b) calculated from the earth elevations and increment al wedge theory, or (c) as inputted separately. See paragraph 3.3.3h for more detail.
- c. Horizontal Loads
 - (1) Trapezoidal (linearly varying distributed) loads, horizontal on stem (W_1 and W_3 through W_4 in Figure 2-1).
 - (2) Concentrated horizontal forces and their elevations (PH₁ and PH₂ in Figure 2-1).
- d. Surcharges over heel and over toe, values and locations
 - (1) Distributed, over all or any part of cross section $(W_{\rm H}$ and $W_{\rm T}$ in Figure 2-1).
 - (2) Up to five vertical concentrated line loads parallel to wall, (P through P in Figure 2-1), plus force P centered on the top of the stem and P anywhere on the base.
- e. Wind direction and magnitude, psf
- f. Earthquake effect toward heel and toward toe (really two

subcases)

- g. Design criteria
 - (1) Load factors for R/C Ultimate Strength Design or overstress factor for Working Stress Design.
 - (2) Allowable bearing capacity, range of values over ranges of allowable toe base elevations and base widths (see paragraph 7.2.2).
 - (3) Minimum creep ratio (see paragraph 3.3.3c for guidance).
 - (4) Minimum factor of safety against shear friction sliding (see paragraph 6.1a).



Note that forces P through P are applied at finished grade elevations while P_{vs} and P_{vb} are applied directly to the concrete.

Figure 2-1. ILLUSTRATION OF APPLIED LOADS

2-3

- (5) Minimum safety factor for cohesion and $(\tan \phi)$ data values used in sliding determination by allowable strength equilibrium methods (see paragraphs 6.1.2 and 6.1.3).
- (6) Limiting value of overturning stability resultant ratio (paragraph 5.4.2).
- (7) Reinforced concrete design parameters (see Exhibits E and F for items).
- (8) Specification of "hydraulic" or "nonhydraulic" structure.
- (9) Heel earth cover crack control (see paragraph 3.3.3k).
- h. Loading Classification
 - (1) Long-term operation.
 - (2) Short-term operation.
 - (3) Normal operation plus earthquake.

2.3.2 <u>Typical Application of Load Cases</u> - Any load case may have any or all of the effects described in paragraph 2.3.1.

3. STABILITY

3.1 INTRODUCTION TO STABILITY CRITERIA

3.1.1 The criteria for stability are the most uncertain and among the most important data used in design of an inverted-T wall. A complete stability analysis includes the effects of overturning,* sliding,** base pressures,† and settlement. Settlement computations are beyond the scope of this program.

3.1.2 This computer program will lead the engineer/user through either design or analysis of an inverted-T wall with multiple load cases and a multilayered soil system with assorted surcharges. It will show the engineer a set of recommendations for design or analysis parameters and then use the user's decisions to complete the detailed analyses. 3.1.3 This computer program will design or analyze for shallow-seated stability limited to forces on the wall acting at or above the interface between the bottom of the wall base and the soil. The term "wall base" is used here to include any key. Deep-seated stability which includes the earth beneath the lowest point of the base or key is beyond the scope of the program.

3.2 CONFIGURATIONS

3.2.1 Design Load Configuration References

a. Paragraphs 2.3.1c and 2.3.1d describe surcharge types.

b. Figure 3-2 shows details of wall dimensioning.

c. Figure 3-4 shows location of edges of structural excavation.

d. Paragraph 3.3.3d describes the effects of sheet pile cutoff walls.

e. Figure 3-1 illustrates the soils environment data.

3.2.2 <u>Base Contact Configurations</u> - This computer program works with a unit slice of wall, measured along the vertical plane through the basic working point shown in Figure 3-2. This is taken as being the toe side of the top of the stem.

^{*} Discussed in Section 5.

^{**} Discussed in Section 6.

⁺ Discussed in Section 7.



Figure 3-1. FINISHED GRADE SOILS CONFIGURATION

a. <u>General Case--Trapezoidal Base</u>. In the general case of a curved wall segment, the base contact soil-structure interface is curved. The base plan used for calculations in this program uses straight-line approximations for the curved outer edges of the unit slice:



(See paragraphs 7.1.2b and 7.1.3b for formulas for base pressures.)

b. <u>Usual Case--Straight-Line Wall Segment</u>. The usual case of a straight-line wall utilizes the familiar expressions for pressures under a rectangular base contact area. See paragraphs 7.1.2a and 7.1.3a for formulas for these base pressures.

3.3 GENERAL STABILITY CAPABILITIES

3.3.1 <u>Wall Dimension Variations</u> - The heel side is taken as being the side with the greater driving force on the wall. Variable values are set to the default value, unless defined by the user or controlled by the various design routines. See Figure 3-1 for soils system description. 3.3.2 <u>Quantities For Cost Comparison</u> - Cost comparisons are used for identifying optimum designs and for design memorandum quantity takeoffs. Items considered include:

a. <u>Structural Excavation</u> below existing grade is shown in Figure 3-4. It will be calculated separately for each existing soil layer so that different unit prices can be used in each layer.

- b. Structural Backfill, to either
 - (1) Existing grade or
 - (2) Finished grade if below existing grade. (Concrete volume not included.)



14 A.

Figure 3-2. WALL CROSS SECTION SHOWING MAJOR VARIABLES



١

Figure 3-3. WALL CROSS SECTION WITH ILLUSTRATION OF ALL DEFAULT VALUES FOR HYDRAULIC STRUCTURES

Default values are taken from EM 1110-2-2501, unless otherwise noted.

F



* slope to toe }
slope to hrel
default = level (shown -)
excavation side slope, default = 1:1

% excavation extra width, default = 2.0 each side

Figure 3-4. STRUCTURAL EXCAVATION

c. <u>Volume of Concrete</u> (stem, base, and key calculated and priced separately).

d. Embankment Above Existing Grade.

3.3.3 General Criteria

a. Incrementing Dimensions of Wall - Use a 3-inch increment for base width, toe base embedment, and key length. Use a 1/2-inch increment on concrete thicknesses. Use a 0.1-foot-horizontal to 1.0-foot-vertical increment on base slope. See also paragraph 5.3.1.

b. <u>Limitations on Input</u> - The computer program will allow complete flexibility for input. The specific values shown in this discussion are suggested values; therefore, the designer will be able to override them with his own values (the default limits are for design control as in paragraph 5.3.1):

> (1) Minimum earth cover over top of heel: The default minimum value is (3 feet + exposed stem height/10), but not less than 5.0 feet, from paragraph S-12 of EM 1110-2-2501.

F

F

F

- (2) Elevation of bottom of base at end of tow: Controlled within limits established by the user. The usual highest elevation is to produce no cover over the toe. The usual minimum elevation is to produce an earth cover over the toe equal to one half the wall height. Note that as the toe elevation changes, so does the heel; and this may affect the heel cover discussed in paragraph 3.3.3b(1).
- (3) Base slope: Controlled within limits established by the user. Usual limits are level and a 1:3 slope.
- (4) Key length: Controlled within limits established by the user. Usual limits are no key and 0.8 times the stem height.
- (5) The stem ratio is set by the user. The default value is 0.33, from paragraph 1-09c of EM 1110-2-2501, except 0.25 when water is within 1.05 feet of the top of the stem.

c. <u>Hydraulic Gradient</u> (1/creep ratio) at the toe, based on earth under the base. Minimum permissible creep ratios for boil control may be taken from page 5-5 of the draft manual for floodwalls (Loading No. 1 values are also in EM 1110-2-2501):

Type of Foundation Soil	Floodwall Loading No. 1 for Water at Top of Stem		Floodwall Loading No. 2 for Water 3 Feet Below Top of Stem	
Granular (sand or gravel)	4.0		4.0	г٦
Uniform sands and silts	3.0	r 1	3.0	FD
Well-graded sandy silts	2.0	[F]	2.0	
Lean, sandy and silty clays	1.8		2.0	

d. <u>Sheet Pile Cutoff Walls</u> - Neglect bearing value and sliding resistance of cutoff walls. For boil control purposes only, comparing the creep ratio developed against the allowable minimum value for paragraph 3.3.3c, the program will use two items of input data:

- Effective length of sheet pile below the bottom of key, and
- (2) A control parameter to select the location of the creep path portion between the bottom of the effective length of sheet pile and the end of the toe:
 - (a) One path along the toe-side face of the sheet pile, key, and the bottom of the base-soil interface.
 - (b) Or the other path being a single, straight line from the bottom of the effective length of sheet pile to the end of the toe.

The two creep paths are to provide minimum and maximum creep ratios.

R p.6

e. <u>Drains and Filters</u> - These will be considered offective in shortening the seepage path, which may serve to <u>increase</u> the seepage uplift. In accordance with Engineering Manual recommendations, weep holes will <u>not</u> be considered; but the user may include them if he wants to, by coding his water elevations and/or pressures accordingly.

f. <u>Hydrostatic Pressures</u> - The user has three ways to control the calculation of line-of-creep '.vdrostatic pressures to be used for design. For analysis, the user may input his own set of pressures for any or all load cases, in which case these input pressures will be substituted for the pressures calculated for that load case(s). Note that these options do not apply to boil control as discussed in paragraph 3.3.3d:

- (1) The first control has two options over how multiple load cases are handled:
 - (a) Option 1: Each load case uses its own pressures according to the controls described in paragraphs 3.3.3f(2) and 3.3.3k. This is the default option.
 - (b) Option 2: All load cases use the pressures determined for the first load case listed in input data list.
- (2) The second control has four options that may be used for design or analysis:

- (a) Option 1: The line of creep calculations are as described in EM 1110-2-2501 and as illustrated and discussed in detail in Exhibit H for sliding and Exhibit K for overturning. This is the default option for this control. Its action combines with the heel earth crack control (paragraph 3.3.3k) to determine how the pressures are determined.
- (b) Option 2 (perched water table): Any load case(s) will use the water elevation over the toe for weight and horizontal pressure above the toe only. Uplift will be hydrostatic, based on the water elevation over the heel. This would be selected by the user for a channel with an impervious floor:



(c) Option 3: Pressures will be those caused by the weight of water over the heel and toe. Uplift will be a linear variation between the heel and toe hydrostatic pressures. The user might select this option for a wall on rock:

D



(d) Option 4: Water weight and horizontal pressures above the base will be hydrostatic pressures calculated from the input water elevations. Uplift pressures will be input data for analysis only; will be used as zero for design:



(3) The third control is the heel earth cover crack set of options described in paragraph 3.3.3k. It is not applicable to options (b), (c), and (d) under paragraph 3.3.3f(2).

g. <u>Stability Design for Economy</u> - Working within user-defined ranges of values for toe embedment, base slope, and key length, the program will determine the combination producing the least cost (calculated as described in paragraph 3.3.2). The least-cost combination will be offered to the user for acceptance or modification before the final analysis results are written to the report file and/or written to the user's time-shari. prminal.

h. <u>Type of Analysis</u> - Both Coulomb and incremental trial wedge analysis values of active earth pressures on a vertical plane along the end of the heel will be presented to the user. The user will then select the method to be used (or input his own values) for stability analysis. Default is the Coulomb method. The pressures determined from the final stability analysis will be the ones used for structural design. The methods of calculation are presented later in this document:

(1) Coulomb: See paragraph 4.3.

(2) Incremental Wedge: See paragraph II of Exhibit A.

i. The program will assume, unless otherwise directed by the user, that the wall is a hydraulic structure within the meaning of EM 1110-1-2101.

j. <u>Maximum Movement</u> of floodwalls will be estimated according to the procedure described in paragraph 5-5 on page 5-7 of the draft manual for floodwalls. The values of C_1 and C_2 will be input data to be selected by the user. This estimate is to be optional. Default action is to omit it.

k. <u>Cracks in Earth Cover Over Heel</u> - The presence or absence of vertical cracks in the earth cover over the heel, as discussed on page S-9 of EM 1110-2-2501, will be controlled by the user, with the following options that will be effective for both overturning and sliding stability analyses:

- (1) Option 1: A crack, regardless of the depth of cover, the default action in accordance with paragraph S-15a on page S-18 of EM 1110-2-2501. This will preclude the application of active earth pressure at the heel, with the line of creep starting at the bottom of the crack.
- (2) Option 2: No crack, regardless of the depth of cover. This will cause the line of creep to start at the ground surface. This is the option used in the calculations in Exhibits, H, I, and J and is the default option for retaining walls.

F

F

[ws]

FD



4. APPLIED PRESSURES

4.1 GENERAL BASIS OF EARTH PRESSURES

This section includes the determination of active and passive earth pressures which are obtained from limiting conditions of equilibrium (failure criteria). Methods for approximating at rest earth pressures are also included.

4.2 EARTH PRESSURE APPLICATION LOCATION

Earth pressures will be calculated along vertical planes at the ends of the base for stability determination and base slab design. Earth pressures for stem design will be calculated at the face of the stem.

4.3 METHOD OF COMPUTATION OF EARTH PRESSURES

4.3.1 Active Earth Pressure can be obtained by the Coulomb equation or by incremental trial wedge analysis, as discussed in paragraph 3.3.3h. Exhibit A describes the methods to be used and shows a comparison of the results from the two methods for nine load cases representative of a wide range of practical problems. The user can select either method or may input his own pressures. The default method is by Coulomb's equations.

a. <u>Coulomb's equation</u> for the fully active case in a one-layer soil is shown below. Exhibit A shows the extension to a multiple-layer system:



(Angles are shown +)

where

$$K_{a} = \frac{\sin^{2} (\alpha + \phi)}{\sin^{2} \alpha \sin (\alpha - \delta) \left[1 + \sqrt{\frac{\sin (\phi + \delta) \sin (\phi - \beta)}{\sin (\alpha - \delta) \sin (\alpha + \beta)}}\right]^{2}}$$

= derived from equation 3 or page 2 of EM 1110-2-2502, with β substituted for i and (90 - α) substituted for β .

When $\beta > \phi$, the slope is unstable and so equivalent values of K_a are unrealistic. Data that include a backfill slope angle β larger than ϕ will be rejected and the user will have to input his own value for K_a or use the incremental trial wedge option.

b. <u>The Incremental Trial Wedge Method</u> shown in Exhibit A must be applied with care to a multiple-layer soil system. When the trial wedge method is used in a multiple-layer soil system, more accurate (and more costly) results are obtained from using a failure surface with different slopes in each different type of soil. The weighted average single-plane slope used in Exhibit A costs significantly less money to execute but yields forces as much as 15 percent lower than the more accurate values. The 15 percent value is for an extreme case where the upper half of the backfill is sand ($\phi = 45^\circ$, $\gamma = 120$ pcf) and the lower half is clay ($\phi = 0^\circ$, c = 300 psf, $\gamma = 120$ pcf). For more usual sand ; values near 35°, the error is about 10 percent. The program will have the dual capacity of single-plane or multiple-plane analysis. Single-plane analysis will be used for design; both procedures will be available for analysis. The multiple-plane analysis procedure used in the above comparison is shown in Addendum F to Exhibit A.

c. <u>Arching Active</u> pressures are accounted for in accordance with paragraph 3g of EM 1110-2-2502. The program will provide for a "Correction Factor for Moment Arm" (CFMA) that will be used as an activepressure moment multiplying factor:

 $M = active force \times moment arm \times CFMA$

This will act to increase moment but not horizontal force in both stability and stress analyses.





where

$$K_{p} = \frac{\sin^{2} (\alpha - \phi)}{\sin^{2} \alpha \sin (\alpha + \delta) \left[1 - \sqrt{\frac{\sin (\phi + \delta) \sin (\phi + \beta)}{\sin (\alpha + \delta) \sin (\alpha + \beta)}}\right]^{2}}$$

= derived from equation 4 on page 2 of EM 1110-2-2502, with β substituted for i and (90 - α) substituted for β .

R

R

4.3.3 At Rest Pressures can be approximated by several methods. The at rest state is not a limiting condition and is not subject to a general analytical formulation. Established sources for at rest pressures include, but are not limited to, the finite element method, Jaky's equation, and as described in paragraph 3d on page 3 of EM 1110-2-2502. These methods are described briefly in Exhibit D. The program will not calculate at rest pressure coefficients as such, but the user may input his own horizontal earth pressure coefficients for the soil layers over the heel, to be used instead of Coulomb's active earth pressure coefficients. It will be assumed, unless overridden, that the distribution of at rest pressures is the same as for active pressures.

4.3.4 Force Along Soil-Stem Interface is accounted for by the angle δ in the sketches in paragraphs 4.3.1a and 4.3.2. The default value for δ , to be used if an actual value has not been determine by the user, is zero. The value of δ may be set to any other value for any load case.

4.4 RESISTING FORCE DISCUSSION FOR OVERTURNING AND SLIDING

4.4.1 <u>Precaution</u> - Passive pressure should never be used unless construction and maintenance of the backfill over the toe assure its effectiveness. The program will leave this decision to the user, providing for separate soil layers beyond and over the toe. See soil layers 6 and 7 in Figure 3-1.

4.4.2 Distribution for Overturning Calculations (see Exhibit K for a more detailed discussion):

a. Passive pressure distribution options for overturning analysis of walls with keys, on soil or rock foundations, are shown in part 1 of Figure 4-1. Option "a" in the figure is the default preferred distribution for floodwalls (EM 1110-2-2501, Plate 5). Option "c" in the figure is the default preferred distribution for retaining walls. The pressure values have no arbitrary upper limit since separate calculations for sliding account for passive pressure limiting values. If option "e" (strut) is selected, then all of the horizontal equilibrium force will be taken by the strut.

b. Walls without keys (see part 2 of Figure 4-2), founded on either soil or rock, will be assumed to use a combined horizontal equilibrant force to resist the Σ H forces, broken into two parts:

> A concentrated force along the base of the wall, assumed to be mobilized first but limited to a maximum value of

N tan ϕ + cL

F

where N is the total net force normal to the base.

(2) A force due to passive pressure distribution as described for walls with keys in paragraph 4.4.2a. This force will be assumed to be mobilized after the concentrated force limit is reached and will have no arbitrary upper limit. See part 2 of Figure 4-1.

It is recognized that a wall without a key, that has an inclined base as shown in part 2 of Figure 4-1, falls into a "gray area" of design with respect to overturning, since the wall could behave in a manner similar to a wall with a key. If the user should have a wall without a key but with a sloping base and want the wall to be analyzed with the

procedure described in paragraph 4.4.2a (walls with a key), an input value of 0.01 foot for the key length will cause the desired action but with a wall that essentially has no key.

4.4.3 Limiting Value for Sliding - The limiting value of passive pressure for the sliding analysis may be selected by the user from the following options (methods a and b supply default values as appropriate):

$$P_{p} = \frac{1}{2} \frac{H^{2}}{2} \tan^{2} \left(45 + \frac{4}{2} \right) + 2H \tan \left(45 + \frac{4}{2} \right) c'$$

where

$$\begin{array}{l} \text{''} = \tan^{-1} (\tan \text{;/FS}) \\ \text{c'} = \text{c/(FS + 2c')} \end{array} \qquad \begin{bmatrix} \text{F, from equation's } Aa \& 4b \\ \text{on p. 27, with paragraphs} \\ 1-19e \text{ on p. 29} \end{bmatrix}$$

where ; and c are test values.

b. For Retaining Walls (EM 1110-2-2502, paragraph 3c):

$$P_{p} < \frac{H^{2}}{2} \tan^{2} \left(45 + \frac{H^{2}}{2}\right) + 2H \tan \left(45 + \frac{H^{2}}{2}\right)c$$

[R]

Remember the warning about the uncertainty of cohesion given in the referenced EM paragraph.

c. A User-Supplied Value - In the expressions in options a and b above, the factor, "tan²(45 + t'/2)" will be replaced by Coulomb's K_p as shown in paragraph 4.3.2 (adjusted for multilayer soil system). 4.4.4 Reduction of Excessive Required Passive Pressure from sliding stability is not possible through simple increase in base width, so the program will, in the design mode, increase base embedment, base slope, and/or key length instead. See paragraph 5.3. 4.4.5 An Alternate Resisting Strut Force will be provided as an option to act instead of passive pressure to provide horizontal restraint. This is to provide for structural features, such as channel bottom slabs, that would provide positive restraint. This is illustrated with NPPD = 5 in Figure 4-1.


NPPD data value	diagram in parts 1 and 2	Description
1	a	As in Plate 5 of EM 1110-2-2501 (item b) (default distribution for floodwalls).
2	Ь	Alternate distribution for diagram a, when toe cover (Soil layers 6 & 7)" is assumed ineffective.*
3	с	Default distribution for retaining walls.
4	d	Alternate distribution for diagram c, when toe cover (Soil layers 6 & 7)" is assumed ineffective.*
5	e	Strut reaction, instead of passive pressure.

NOTE: Option e in diagrams (strut reaction) will cause all horizontal resistance to be taken by strut in both sliding and overturning calculations. *Do not use passive pressure above base of toe unless construction and maintenance of backfill will assure its effectiveness.

See Figure 3-1

¢

Figure 4-1. PASSIVE PRESSURE DISTRIBUTION OPTIONS

4.5 HYDROSTATIC PRESSURES

Hydrostatic pressures will be applied as explained in paragraphs 3.3.3f and 3.3.3k. In the computation of line-of-creep pressures, no creep losses are included along the portion of the base where the structure is not exerting a compressive pressure on the subgrade, except on the toe-side face of the key when this face is in passive contact with the subgrade. D

4.6 SURCHARGE PRESSURES

4.6.1 Vertical Pressures on the base will be computed from Boussinesq's equations:

a. The expression for vertical pressures under a point load (unit slice of a line load) is integrated from equation 35.1 on page 203 of Soil Mechanics in Engineering Practice, by Terzaghi and Peck, John Wiley & Sons, 1948:

$$p_{y} = \frac{30}{2\pi z^{2}} \begin{bmatrix} 1 \\ -\frac{1}{r^{2}} \\ 1 + \frac{r^{2}}{r^{2}} \end{bmatrix}^{5/2}$$

where

 $p_v = vertical pressure, psf$

 ϕ = point load, 1b

r = horizontal component of distance from load to point

z = vertical component of distance from load to point

b. The method to be used for computing vertical pressures under a distributed load is shown in Figure 17-7 and in the text between the figure and the end of paragraph 17.10 on pages 259-262 of Soil Engineering by M. C. Spangler, International Textbook Company, 1951: "...Equation 17-1 and Fig. 17-4 may be used to determine the unit pressures at points in the soil which are not under the center of the area over which a uniform load is applied. The influence coefficient for the unit pressure under point 0 due to the uniform load on the area BCFF may be obtained from the coefficients for various rectangles, as follows: Eectangle BCFF = ACIC-ARPO-DEFO+DERC. Multiplying the resulting coefficient by the uniform load on the area BCFF gives the

unit pressure under point 0." Equation 17-8 from page 259 is shown below:

$$\frac{\sigma_{z}}{p} = \frac{1}{4\pi} \left[\frac{2ABH}{H^{2}(A^{2} + B^{2} + H^{2}) + A^{2}B^{2}} \cdot \frac{A^{2} + B^{2} + 2H^{2}}{A^{2} + B^{2} + H^{2}} + \frac{A^{2} + B^{2} + 2H^{2}}{A^{2} + B^{2} + H^{2}} + \frac{A^{2} + B^{2} + 2H^{2}}{A^{2} + B^{2} + H^{2}} + \frac{A^{2} + B^{2} + H^{2}}{H^{2}(A^{2} + B^{2} + H^{2}) + A^{2}B^{2}} \right]$$

where

- A,B = width and length of loaded area
 - H = vertical distance from corner of rectangle $\mbox{ A}\times\mbox{ B}$, down to the point where pressure is being calculated
 - p = uniformly distributed unit load
 - = unit pressure at depth H

Figure 17-7 from page 261 is shown below:



Fig. 17-7. Point in Soil Eccentric to Loaded Area

4.6.2 Horizontal Pressures will be computed using modified forms of the theory of elasticity equations described in Addendum B to Exhibit A. These pressures vill be added to earth pressures computed according to coulomi's equations. The incremental wedge method includes all surcharges in the trial wedge calculations. A study comparing the elasticity equations with the alternate concept of using Boussinesg's equations for vertical pressure and multiplying by the factor K_a is shown in Exhibit G. 4.7 EFFECT OF SEEPAGE PRESSURES ON EFFECTIVE EARTH UNIT WEIGHT Seepage pressures tend to increase the effective unit weight of soil when the seepage flow is downward through the soil and to decrease the effective unit weight of soil when the seepage flow is upward through it. The program will calculate the effective unit weight as follows:

$$\gamma_{eff} = \gamma_b + \frac{\Delta H}{L} \gamma_w$$

for the wall side with the higher water elevation, or

$$\gamma_{eff} = \gamma_b - \frac{\Delta H}{L} \gamma_w$$

for the wall side with the lower water elevation, where

 γ_b = buoyant unit weight of soil γ_{eff} = effective unit weight of soil γ_w = unit weight of water

 $\Delta H/L$ = hydraulic gradient (reciprocal of creep ratio)

The user will have the option of using either effective unit weight or buoyant soil unit weight in the computation of active and passive pressures. Default action will be to use buoyant soil unit weight for both active and passive soil zones. 5. OVERTURNING STABILITY -

5.1 GENERAL

5.1.1 <u>Terminology</u> is as described in paragraph 1-5 on page 8 of EM 1110-2-2501.

5.1.2 Stability Requirements are grouped into sliding, overturning, and bearing. Sliding stability is considered in Section 6, overturning in Section 5, and bearing in Section 7 of this document.

F

R

5.1.3 <u>Horizontal Pressures</u> are calculated as described in Section 4. Actual and allowable bearing pressures are determined as described in Section 7.

5.2 ANALYSIS PROCEDURE

5.2.1 The Moment Center is taken as the toe end of the bottom of the base.

5.2.2 Force Summations in the horizontal and vertical directions will be made simultaneously so that resisting forces can be included when acting along a sloping base. See paragraph 4.4.2 and Exhibit K for a detailed discussion.

5.2.3 <u>Moment Summation</u> establishes the location of the resultant force, which then permits calculation of the earth bearing pressures as described in paragraph 7.1. Moments due to active earth will be multiplied by a "Correction Factor for Moment Arm," to shift the effective loading upward for the arching active case as discussed in paragraph 3g of EM 1110-2-2502. This will act to increase moment but not horizontal force.

5.2.4 <u>Sample Calculations</u> are shown in Exhibit K.

5.3 DESIGN PROCEDURE

The design criteria and options are described in general terms in paragraph 3.3.3.

5.3.1 The Basic Program Procedure for both sliding and overturning stability can be diagrammed thus, in the form of nested FORTRAN DO loops:



User controls toe length parameter manually or sets stem ratio.

Toe elevation parameter is incremented by 3-inch units, between limits set by the user.

Base slope parameter is incremented by 0.1-foot-vertical or 1foot-horizontal units, between limits set by the user.

Key length parameter is incremented by 3-inch units, between limits set by the user.

Program determines minimum base width (to nearest 3-inch unit) to satisfy resultant ratio, bearing pressure, sliding, creep ratio, and heel cover requirements.

If passive pressure from sliding stability check is too high, go to next key length; skip cost comparison.

Calculate construction cost; save design if cheaper than last saved design.

After the nested loops in the program have completed searching the ranges of possible parameter values within the limits set by the user, the design that yields the lowest estimated construction cost is presented to the user for his approval or revision.

5.2.2 By manipulating the limits, or by calling for just an analysis of any one combination of toe elevation, base slope, key length, and base width, he can satisfy himself that the full-range design is the best one. The analysis mode reports the estimated cost along with the pressure values and resultant ratio.

5.3.3 The user can set a desired value for any parameter by setting both the upper and lower limits for that parameter to the (one) desired value.

5.3.4 Section 7 describes how the program determines the actual and allowable bearing pressures under the base.

5.4 RESULTANT RATIO

5.4.1 <u>Definition</u> - The resultant ratio is defined as being the ratio of the horizontal distance from the end of the toe to the point of intersection of the resultant and the bottom of the base <u>to</u> the horizontal base width.

R

5.4.2 Limiting Values

a. 0.33 minimum for:

F

- (1) Floodwalls with water 3 feet below the top of the stem (EM 1110-2-2501 Loading No. 2).
- (2) Retaining walls loaded with active earth pressure and EM 1110-1-2101 Group I loads (EM 1110-2-2502).
- b. 0.25 minimum, recommended for:
 - (1) Floodwalls with water to the top of the stem (EM 1110-2-2501 Loading No. 1).
 - (2) Retaining walls with at rest earth pressure on the stem or maintenance condition loading. This is interpreted to mean EM 1110-1-2101 Group II loads.

c. For an earthquake loading condition, the resultant may fall anywhere within the base provided the allowable foundation pressure for the earthquake loading condition is not exceeded. [D]

5.4.3 <u>Relationship to Percent Effective Area</u> - The relationship is simplified for this computer program by the consideration of only unit slices of wall.

a. The percent effective area is defined as simply the ratio of base area in compression to the total base area, times 100.

b. The relationship between percent effective area and resultant ratio is simple for the case of a rectangular base:

Percent	
Effective Area	Resultant Ratio
100	0.5
100	0.33
75	0.25
50	0.1667

There is a constant ratio of 300:1 between the values calculated from the two concepts for rectangular bases with resultant ratio values of 0.33 or less. For example, 75 : 300 = 0.25.

c. The relationship for a trapezoidal base is more complex and is not as meaningful as it is for a rectangular base, so the derivation is not shown here.

5.5 UPLIFT

The action to be taken in the computations for overturning stability is modified if the resultant force falls outside the kern.

These changes in action affect uplift and bearing pressure computation, as described in Exhibit K.

A CONTRACTOR

6. SLIDING

6.1 AVAILABLE METHODS OF ANALYSIS

This program will use any of the following four methods for determining safety against sliding, as selected by the user: 6.1.1 Shear Friction Method - This method complies with guidance furnished in ETL 1110-2-184. It is the computer program default method for retaining walls. The method uses the passive pressure formula as given in paragraph 4.4.3b and the formulas for the horizontal components of sliding resistance which is mobilized along the assumed failure surface beneath the base of the wall, as given in paragraphs 6.2.1.1 and 6.2.1.2. Deficiencies associated with this method are discussed in paragraph 6.2.1.4. This method is illustrated in Exhibit H. Allowable Strength Equilibrium Method (I) - This method complies 6.1.2 with the provisions in EM 1110-2-2501, using different factors of safety on tan ϕ and c applied to the driving and resisting forces. This method is the computer program default method for floodwalls. The passive resisting force is as presented in paragraph 4.4.3a. This method is illustrated in Exhibit I.

ED

F

6.1.3 <u>Allowable Strength Equilibrium Method (II)</u> - This method is the same as discussed in paragraph 6.1b above except the same factor of safety is used on tan ϕ and c in both the driving and resisting forces. This method is illustrated in Exhibit J.

6.1.4 <u>Modified Shear Friction Sliding Method</u> - This method is presently incomplete and will be completed upon receipt of guidance from OCE. This sliding option, which will be based on a shear friction approach, will incorporate solutions to the deficiencies cited in paragraph 6.2.1.4 relating to the formulations presented in ETL 1110-2-184. In the interim, this option will be provided for in the structure of the computer program but not actually programmed.

6.2 DISCUSSION OF ANALYSIS METHODS

The first three methods listed in paragraph 6.1 are discussed in more detail in paragraphs 6.2.1 through 6.2.3.

6.2.1 Shear Friction Method (ETL 1110-2-184) - The nomenclature for Equations 6-1 and 6-2 is given below:

- R = horizontal sliding resistance which can be mobilized along the critical path beneath the base of the wall
- P_p = passive resistance of the earth or rock wedge adjacent to the wall
- ΣH = net applied horizontal driving force
- ΣV = summation of vertical applied forces above the assumed sliding plane which is below or at the base of the wall
- A = area of the potential failure path which develops the unit shearing strength. (Any portion of the assumed failure plane at the base-foundation interface which is not in compression should be exluded from "A." However, if the assumed failure plane is not at the base-foundation interface but through the soil, no reduction in "A" should be made.)
- ω = angle between the inclined failure path and a horizontal datum plane
- c = cohesive strength = unit shearing strength at zero normal loading along the potential failure path beneath the base of the wall = test ultimate
- ϕ = angle of internal friction of the foundation material (test ultimate value, degrees) or, where applicable, the angle of sliding friction of the wall on the subgrade
- FS = factor of safety

The shear friction method is discussed in the following paragraphs. 6.2.1.1 <u>Resistance Force for Uphill Sliding</u> – The expression for "R" for uphill sliding resistance on a homogeneous (soil or rock) foundation material is taken from Equation No. 1 in ETL 1110-2-184. The angle ω is used with a positive sign in the equation

$$R = \Sigma V \left[\tan (\phi + \omega) \right] + \frac{cA}{\cos \omega (1 - \tan \phi \tan \omega)}$$
(6-1)

See Exhibit B for derivation of Equation 6-1.

6.2.1.2 Resisting Force for Downhill Sliding – The expression for "R" for downhill sliding resistance on a homogeneous (soil or rock) foundation material is taken from Equation No. 2 in ETL 1110-2-184. The angle ω is used with a positive sign in the equation

$$R = \gamma V \left[\tan (\phi - \omega) \right] + \frac{cA}{\cos \omega (1 + \tan \phi \tan \omega)}$$
(6-2)

6.2.1.3 <u>Safety Factor</u> - Sliding stability can be evaluated by following the shear friction formula

$$S_{s-f} = \frac{R + P}{2H}$$

which relates the total available resistance $(R + P_p)$ to the net applied force (SH) which tends to induce sliding. The full test values of φ and c are used in these calculations so that the forces are evaluated at their limiting values. This is the default method for retaining walls and is illustrated in Exhibit H. Passive pressure of earth is calculated using the formula shown in paragraph 4.4.3b or a user-supplied value.

6.2.1.4 <u>Apparent Deficiencies in ETL 1110-2-184 Shear Friction Formulas</u> -The apparent deficiencies in the shear friction safety factor as given in ETL 1110-2-184 are going to be discussed in a WES Miscellaneous Paper which is in the process of being written. These deficiencies are summarized as follows:

a. Any normal component of the passive pressure to the plane of assumed sliding is not considered in the frictional resistance (N tan ϕ). This deficiency exists only when considering inclined failure planes.

b. The safety factor in ETL 1110-2-184 is computed for an inclined plane using horizontal force components. In reality, the safety factor should be considered in a direction consistent with the inclination of the failure plane. The vectors which drive and resist the movement of the structure are along the failure plane; therefore, the safety factor should be the ratio of the resisting to driving forces in the direction of the inclined plane.

c. The frictional, cohesive, and passive resistances are assumed to develop at the same rate. In reality, these resistances do not develop at the same rate. It will therefore never be possible to have a total resistance equal to the sum of their maximums. 6.2.1.5 <u>Passive Resistance of the Wedge Adjacent to the Wall</u> - Passive resistance of soil is calculated as described in paragraph 4.4.3b.

A theoretical value for the passive resistance offered by a homogeneous rock wedge is

$$P_{p} = N \tan (\phi + w) + \frac{cA}{(\cos \omega) (1 - \tan \phi \tan \omega)}$$
(6-3)

where

- w = effective (submerged) weight of the rock wedge above the inclined plane of resistance (plus any surcharge loads)
- ϕ = angle of internal friction or, if applicable, the angle of sliding friction
- ω = angle between the inclined failure plane and the horizontal datum plane (for earth ω = 45 $\phi/2$ usually)
- c = unit cohesion along the failure plane
- A = area of the inclined failure plane along the base of the wedge

a. This simple formulation is valid for only a homogeneous foundation. For layered foundations, passive resistance will be calculated for each layer and superimposed. However, if fissures, seams, cracks, etc., are present, the user can have the program determine the force required to resist sliding and manually evaluate the rock subgrade for its ability to withstand this force. See paragraph 4.4.5.

b. Equation 6-3 above for passive resistance of a rock wedge can be directly transformed into _he usual Coulomb equation for passive earth pressure by introducing assumptions and limitations which are consistent with the Coulomb theory (see Exhibit L for more details).

c. When the presence of a strut is indicated by the input data (item in NPPD being set to 5, see Figure 4-1), the resisting force is calculated as being equal to the net driving force multiplied by the input minimum safety factor (FSMIN) against sliding.

6.2.1.6 Values for the Minimum Shear Friction Safety Factor - Paragragh 7b(3) of EM 1110-2-2502 suggests a value of 1.5 for walls on rock. The minimum shear friction safety factor for a seismic load case should be two thirds of the safety factor for the normal load case, but not less than 1.15.

6.2.2 Allowable Strength Equilibrium Method (I) - Paragraphs 1-16

R

through 1-19 of EM 1110-2-2501 describe a method for evaluating safety against sliding for floodwalls, based on applying a user-selected value of factor of safety on the tan ϕ and c values in the equation for resisting and driving forces. Paragraph S-15d provides an additional description and allows the use of the assumed failure surfaces shown in paragraph 6.3.1 of this document. The procedure includes the following steps, as illustrated in Exhibit I.

a. Assume a trial value of factor of safety FS.

b. Calculate the allowable values for cohesion c' and friction ϕ' from Equations 4a and 4b in paragraph 1-17 of EM 1110-2-2501. They are formulated as follows:

$$c' = \frac{c}{FS + 2q}$$

where

and

q = c' (see paragraph 1-19e of EM 1110-2-2501)

$$c' = \frac{c}{FS + 2c'}$$
$$\phi' = \tan^{-1} \left(\frac{\tan \phi}{FS} \right)$$

The equation for c' can be rearranged to read

$$2(c')^2 + FS(c') - c = 0$$

With FS positive, this equation yields one positive real value of c' and one negative value. The positive one is used:

$$c' = \frac{\sqrt{FS^2 + 8c} - FS}{4}$$

where c and c' are in tsf.

c. Calculate K_A and K_p from ϕ' .

d. Sum all driving forces ΣD_{ω} parallel to the assumed failure plane below the base of the wall. ΣD_{ω} includes the net hydrostatic

forces, if any, acting on the wall. The driving force contributed by the soil is calculated as a function of c' and ϕ' , but the effect is neglected if its value is negative since the soil cannot pull on the structure.

e. Sum the resisting forces ΣR_{ω} parallel to the assumed failure plane below the base of the wall. The resisting forces consist of passive earth forces and friction and/or cohesion forces developed along the assumed failure plane below the base of the wall. ΣR_{ω} is formulated as a function of ϕ ' and c'.

f. Two equations (ΣD_{ω} and ΣR_{ω}) then exist, each of which can be solved for any assumed safety factor FS. The expression $\Sigma D_{\omega} - \Sigma R_{\omega} = 0$ is solved by an iterative procedure with assumed safety factors. The actual safety factor for the structure is the one computed where $\Sigma D_{\omega} - \Sigma R_{\omega} = 0$.

6.2.3 <u>Allowable Strength Equilibrium Method (II)</u> - This method and its procedure are identical with the Allowable Strength Equilibrium Method (I), except that the factor of safety for cohesion FS_c is taken as being equal to the factor of safety for friction FS_{ϕ} . In this method,

$$c' = \frac{c}{FS}$$

and

$$\tan \phi' = \frac{\tan \phi}{FS}$$

6.3 DISCUSSIONS COMMON TO ALL SLIDING METHODS

6.3.1 <u>Assumed Failure Surfaces</u> - The program will try a series of failure surfaces, each one made up of two planes, varying between the limiting cases shown below unless the user exercises his option of specifying a single value for the angle ω :





This position is indicated in program by KFLAG=0 in the input data.

The situation shown above is still being studied by OCE and LMVD.

(The angle ω is defined as the angle from the horizontal, up to the assumed sliding failure plane.)

6.3.2 Effect Due to Some of the Base Not Being in Compression With the Foundation

6.3.2.1 <u>Portion of Base in Compression With Foundation (Percent Effective Base)</u> - The total base-foundation interface of the T wall may not be in compression with the foundation. If any part of the surface under consideration is along the base-foundation interface and is not in contact with the foundation, this portion should be neglected when obtaining the effective base area to resist sliding. However if the assumed failure surface is not along the base-foundation interface but through the soil, no reduction in the area to resist sliding is made.



B.S.C. Splitt

When the resultant falls outside the kern, a portion of the base of a I wall will not be in compression, thus creating a crack which can result in an increase in uplift pressures. This condition will attect the sliding stability analysis when the assumed sliding plane acts along the soil-structure interface below the base of the wall. (For this condition the program will have to recycle back through the line-ofcreep calculations until the creep path assumptions match the final part of the base that is in contact with the foundation.) For example, consider a wall without a key and with a horizontal base. When the resultant falls outside the kern and the assumed sliding plane is along the interface between the base of the structure and the soil foundation, uplift pressures will be computed assuming no creep loss for the portion of the foundation not in compression. For the condition where the resultant falls outside the kern but the assumed sliding plane is through the soil, for example a wall with a key positioned at the extreme end end the heel, no increase in uplift pressure will be considered because the soil does not lift and form a crack as is the case at the soil-structure interface. Another reason the uplift forces are not affected for this condition is because they are forces inside the soil-structure free body

and therefore do not affect the overall sliding stability. 5.3.2.3 coefficients of Friction ; and Cohesion c to be Used in the Analysis - like ; and c values should be consistent with the material being sheared. A plane of tailure through the soil should use the ; and c of the soil. For any of the failure planes along the soil-structure interface, use the ; and c for sliding friction at the interface.

1.-4

7. BEARING PRESSURES

7.1 CALCULATION OF BEARING PRESSURES

Bearing pressures are calculated during overturning analysis from the vertical resultant (see paragraph 5.2.2 for summation of vertical forces) and its location (see paragraph 5.2.3 for summation of moments). The formulas for earth bearing pressure are different for a base in compression over its entire area and for a base with only part of its area in compression.

7.1.1 The Procedure is to first calculate the minimum pressure, assuming that the entire base is in compression. If the minimum pressure calculates to a positive value, then this was the appropriate assumption. If, however, the minimum pressure calculates to a negative value, then the assumption was not appropriate and the pressures must be recalculated with the assumption that only part of the base is in compression. 7.1.2 Entire Base in Compression - The formation and calculations for obtaining base pressures are straightforward for rectangular or trape-zoidal bases; use $f = (P_y/A) + (Mc/I)$ (no biaxial moment). The properties of the base (A, I, c) are for the total base.

a. For a unit slice of a rectangular base, the formula becomes the familiar



b. For a trapezoidal base, the formula becomes as shown in paragraph C-l of Exhibit C.

7.1.3 Part of Base in Compression - The calculation of bearing pressure when only part of the base is in compression is a two-step process: First, find the location of the zero-pressure edge of the part in compression; then, calculate the maximum pressure:



a. For a unit slice of a rectangular base, the procedure is

 d_1 = known location of resultant d_2 = width of base in compression

First, locate the zero-pressure point as being at $d_2 = 3d_1$, for the triangular distribution. Then, calculate the maximum pressure

$$f = \frac{2P_{y}}{d_{2}} = \frac{2}{3} \frac{P_{y}}{d_{1}}$$

b. For a trapezoidal base, the procedure is summarized below. The general figure is also given below (resultant outside kern, toward wide end):



the procedure is as follows:

(1) An incremental search across the base is done to find the zero-pressure point and the corresponding effective base width. The corresponding width must satisfy the expression

$$f = \frac{P}{A} + \frac{M(b - c)}{I}$$

(2) The maximum stress is

$$f = \frac{6R}{b(w''' + 2w')}$$

where $w^{""}$ is computed using the effective base width. The details of the procedure are given in paragraph C-2 of Exhibit C.

7.2 MAXIMUM ALLOWABLE BEARING PRESSURES

7.2.1 <u>Selection of Maximum Allowable Pressure</u> - The determination of criteria for selecting allowable bearing pressures is beyond the scope of this document and of the computer program that it describes. The procedure in the program is based on linear double interpolation between limiting values read in as data.

7.2.2 <u>Data Structure</u> - The wall may bear on one or more of the existing soil layers (soils layers 3, 4, or 5 in Figure 3-1). There will be four allowable bearing pressure data values for each of these soil layers, as shown in the tabulation below:

Soil	At Top o	of Layer	At Bottor	n of Layer
Layer	For Narrowest	For Widest	For Narrowest	For Widest
Number	Base Expected	Base Expected	Base Expected	Base Expected
3	psf	psf	psf	psf
4	psf	psf	psf	psf
5	psf	psf	psf	psf

The "narrowest" and "widest" base widths consistent with these allowable bearing pressure values will also be input data, separately from any data parameter used to control the stability analysis action.

7.2.3 <u>Method</u> - The program will interpolate linearly between the top and bottom elevation of each soil layer to the elevation of the point being checked and between the "narrowest base" and the "widest base" values to the actual base width being checked. This check will be made at the highest and lowest base elevations in each soil layer. 8. EARTHQUAKE ANALYSIS

8.1 INTRODUCTION

The seismic coefficient method of analysis is used in the following earthquake force computations:

ChS

F

Е

E

a. The earthquake--induced inertial forces of the wall, plus that portion of the adjacent earth and/or water which is assumed to act as an added mass with the wall, is computed as the product of mass times acceleration (F = Ma).

b. The dynamic horizontal earth pressure magnitude and resulting force are approximated by the Mononobe-Okabe method.*

c. The hydrodynamic force is obtained by Westergaard's theory. 8.1.1 The seismic coefficient represents the ratio of an assumed acceleration of the wall to the acceleration of gravity. The earthquake forces are superimposed onto the static forces for the loading condition being analyzed for an earthquake (usually the normal operating condition) in order to obtain the total loads for the earthquake loading case. A static solution is then performed for the stability analysis and structural design.

8.1.2 Where a T wall is used as part of a dam, and where failure of the wall could result in loss of life or extensive property damage, then the final design of the T wall should be in accordance with ER 1110-2-1806.

8.2 SELECTION OF SEISMIC COEFFICIENT

8.2.1 Values for the seismic coefficient should be selected carefully. The most important consideration is the proximity of the structure to known earthquake epicenters and/or faults capable of generating an earthquake. Minimum seismic coefficient values for various sections of the United s can be found in the seismic zone maps of Appendix B of ER 1110-2-1 The seismic coefficient, when multiplied by the acceleration of gravity, gives the bedrock acceleration (also used as

* Seed, H. B. and Whitman, R. V. 1970. "Dosign of Earth Retaining Structures for Dynamic Loads," Proceedings, ASCE Specialty Conference on Lateral Stresses in the Ground and Design of Earth Retaining Structures, pp 103-147.

structure acceleration) to be used in the analysis.

8.2.2 These maps can be used as a guide, in the absence of more accurate data, for determining the peak acceleration that should be used in the calculation of lateral pressures generated by earthquakes. 8.2.3 Structures that are in Seismic Zones 3 or 4, and which will endanger life or cause substantial property damage if they fail, and are on soil in which a liquefaction potential exists, should be analyzed by dynamic or pseudodynamic procedures (ER 1110-2-1806). Liquefaction potential for a material is defined as being when the relative density of the in-place material is less than 70 percent. Embankments constructed by hydraulic fill methods are usually found to be susceptible to liquefaction.

E

E

6.1. Ar 1 ++ 84 . .

8.3 LOCAL SOIL CONDITIONS

Soil foundations may modify the intensity of structure motions. A practical method for including the effects of soil foundations on structure motions is to vary the structure acceleration by a soilstructure-resonant factor of 1 to 1.5. This factor depends on the degree of similarity between the natural period of vibration of the retaining structure and the natural period of vibration of the soil foundation. A review of anticipated and actual damage patterns as a function of soil conditions suggests that the forces and damaging effects included in different types of structures are maximized when there is a similarity in the natural period of vibration is not properly established for a soil foundation, the seismic coefficient should be multiplied by 1.5.* 8.4 INERTIA FORCE OF WALL

The inertia force of the wall mass is computed by multiplying the selected seismic coefficient by the weight of the wall. It is applied at the center of gravity. This force is obtained by multiplying the mass by acceleration as follows:

 $F = ma = ma (g/g) = (a/g) W = \alpha'W$

* Uniform Building Code, 1976 Edition, p. 135.

Also, that portion of the backfill above the heel or toe of the wall which is not included as part of the Coulomb wedge is included as an inertial force acting at its centroid.

8.5 LATERAL COHESIONLESS EARTH PRESSURES DUE TO EARTHQUAKES

The dynamic earth pressure magnitude is approximated by the Mononobe-Okabe method. This assumes that the soil developes a Coulomb wedge behind the wall and that the ground acceleration is uniform within the wedge. The effect of the earthquake motions can be represented by forces $k_{\rm b}W$ and $k_{\rm c}W$ where W is the weight of the sliding wedge and $k_{h}g$ and $k_{ij}g$ are the horizontal and vertical components of the earthquake acceleration at the base of the wall. (See Seed and Whitman.) 8.5.1 Active Earth Pressure Conditions - The active pressure force P_{ae} due to an earthquake is computed, in effect, by the Coulomb theory except that the additional inertia forces k_h^W and k_w^W , as shown in Figure 8-1 are included in the computations. Determining the critical sliding surface and the active pressure corresponding to thi urface leads to the equations as given in Figure 8-1. These are th juations used to compute the active earth pressure in this prog im.

a. The influence of the vertical acceleration coefficient k_v on the earth presence coefficient K_{ae} depends on the corresponding component of the horizontal acceleration coefficient k_h . For most earthquakes, the horizontal components are considerably greater than the vertical components. To study the influence of k_v , the vertical acceleration coefficient is taken as $2/3k_h$. For a value of k_h of 0.1, a value of k_v of $2/3k_h$ equal to 0.067 can cause increases or decreases in K_{ae} of about 2 percent. For a value of k_h of 0.3, a value of k_v of 0.20 can cause increases or decreases in K_{ae} of less than 5 percent. From this it seems reasonable to conclude that the influence of k_v can be neglected for practical purposes. (See Seed and Whitman.) Neglecting k_v , the expression for P_{ae} becomes

$$P_{ae} = \frac{\gamma H^2}{2} (K_{ae})$$

b. For applying the increased earth pressure due to an

earthquake acceleration, $\Delta P_{ae}^{}$ is computed by using the following expression:

$$\Delta P_{ae} = \frac{\gamma H^2}{2} (\Delta K_{ae})$$

where

$$\Delta K_{ae} = K_{ae} - K_{a}$$

and

$$P_{ae} = P_a + \Delta P_{ae}$$

where P_a is the total active static force. P_a is applied at 1/3H (see Figure 8-1) above the base and ΔP_{ae} is applied at 2/3H above the base. The distribution of ΔP_{ae} is assumed to vary linearly from a maximum at the top of the wall to zero at the base. This is important in the design of sections along the structure.

$$P_{ae} = \frac{1}{2}H^2 (1 - k_v)K_{ae}$$

where

$$K_{ne} = \frac{\sin^2 (\alpha + z - \alpha)}{\cos \alpha \sin^2 (\alpha - z - \alpha)} \left[1 + \sqrt{\frac{\sin ((z + x) \sin ((z - x - \alpha)))}{\sin (\alpha + \beta)}} \right]^2$$

$$\approx = \tan^{-1} \left(\frac{k_h}{1 - k_v} \right)$$

$$k_h = \text{horizontal ground acceleration}$$

$$k_v = \text{vertical ground acceleration}$$

$$\gamma = \text{unit weight of soil}$$

 $H = 10^{\circ}$ M of vertical plane AB

 ϕ = angle of friction of soil

δ = β

 $\beta =$ slope of ground surface behind wall

 $\alpha = 90$ degrees

c. The critical sliding surface was considered as in Coulomb theory and the active pressure corresponding to this surface leads to the expression for K_{ae} . It is possible to simplify the mathematical calculation for K_{ae} such that it can be obtained directly from Coulomb's active pressure coefficient, as described by Seed and Whitman.

d. For the more general case, the variation of ΔK_{ae} with a_h , shown in Figure 8-2, depends greatly on the slope angle of the backfill β .

e. The wedge's critical sliding surface along an irregular backfill will be approximated by a single line or a continuous series of straight lines, as the user decides. The increase in horizontal earth pressure due to the earthquake will be formulated and computed by multiplying the mass times the acceleration for the various triangular or trapezoidally shaped earth masses. The force will act through the centroid of the mass and in the most critical direction for the condition being considered.

8.5.2 Passive Earth Pressure Conditions - The increase or decrease in



Figure 8-1. ACTIVE WEDGE DURING EARTHQUAKE





Figure 8-2. VARIATION OF DYNAMIC EARTH PRESSURE COEFFICIENT WITH THE HORIZONTAL GROUND ACCELERATION ${\rm A}_{\rm h}$

a fully passive earth pressure due to an earthquake is computed as follows:

$$\mathcal{L}P_{pe} = \frac{\gamma H^2}{2} (\mathcal{L}K_{pe})$$

where

 $\Delta K_{pe} = K_{pe} - K_{p}$

and

 $K_p = static passive pressure coefficient for the fully passive case (see 4.1.4b)$

For the case where passive earth pressure is used as a stabilizing force, a reduction in the passive earth pressure due to an earthquake acceleration is assumed at the same instant the fill pressure behind the structure is increased. If a reduction in K_p has been used for computing an effective K_p for the static case, this same reduction in ΔK_{pe} is used. P_{pe} is applied at 2/30 above the base. The pressure distribution of P_{pe} is the same as assumed for the active earth pressure condition.

8.5.3 At Rest Earth Pressure Conditions - The increase in an at rest earth pressure due to an earthquake is approximated by the Mononobe-Okabe method. The change in the active earth pressure coefficient ΛK_{ae} is first computed as described in paragraph 8.5.1, and then multiplied by the ratio K_r/K_a to obtain the change in the at rest earth pressure coeff int ΛK_{re} . The change in at rest earth pressure is then computed as follows:

$$P_{\mathbf{re}} = \frac{1}{2} \frac{H^2}{2} (K_{\mathbf{re}})$$

Only the horizontal component of the earthquake acceleration is considered. P_{re} is applied at two thirds the height of the fill above the base. The pressure distribution of P_{re} is the same as assumed for the active earth pressure condition. 8.6 LATERAL COHESIVE EARTH PRESSURES DUE TO EARTHQUAKES

The computation of the dynamic earth pressure for cohesive soils is beyond the scope of this computer program. A nonlinear finite element analysis to account for inelastic strains in the soil could possibly be used for critical cases.

8.7 WATER PRESSURE DUE TO EARTHQUAKES

8.7.1 Method - If the backfill over the toe or the heel is saturated to some level, the dynamic soil force should be determined by using a combination of the Mononobe-Okabe and Westergaard theories. When water exists on both sides of the structure for the loading condition being analyzed, simultaneous increases and decreases in water pressure on opposite sides of the structure are included in the analysis. To include the dynamic effect of water in a saturated backfill using Westergaard theory, the computation is divided into two parts. For that part of the soil below the saturation line, the increase of the force from the earth pressure is computed by the Mononobe-Okabe method using the buoyant soil weight. The increase of the force due to the water in the backfill is computed by the Westergaard theory, and 100 percent of this is used and applied with the other forces as shown in diagram (5) of Figure 8-3. If water is the only medium which acts on the wall, the Westergaard theory is applicable to determine the pressure and its distribution on the wall. 8.7.2 Westergaard Theory - By the Westergaard theory, the dynamic water pressure down to depth y below the surface for a total water depth h is expressed by Equation 3 on page 5 of EM 1110-2-2200 as

$$P_{e_2} = \frac{2}{3} C_e \alpha y \sqrt{hy}$$

D

The additional moment at depth y due to P_{e_1} is given by

$$M_{e} = \frac{4}{15} C_{e} \alpha y^{2} \sqrt{hy}$$

with

$$C_e = \frac{51}{\sqrt{1 - 0.72 \left(\frac{h}{1000 t_e}\right)^2}}$$

where g is acceleration of gravity (32.2 ft/sec^2) .



Figure 8-3. DYNAMIC PRESSURE VARIATIONS

9. STRUCTURAL DESIGN CRITERIA

9.1 GENERAL

9.1.1 Critical Sections and Alternate Pressure Diagrams will be selected as described in paragraphs 1-12, Chapter 1, of EM 1110-2-2501 F as amended by Paragraph S-21 of the March 1961 Supplement to EM 1110-2-2501, unless the user elects to use the overturning stability pressures exactly as calculated. The default procedure will be as described in EM 1110-2-2501. All wall components will be designed for flexure, shear, and deflection. The heel slab sill also be designed for these effects plus axial tension caused by passive pressure on the key (if the key is under the heel). The stem will be designed for flexure, shear, and deflection, plus the axial compression caused by the weight of concrete. Critical sections for shear will be at the face of the stem in the heel, at the bottom of the slab in the key, at a distance d from the face of stem in the toe, as at a distance d above the top of the slab in the stem. The key and/or too should be checked as deep beam brackets if they are short enough compared to their thicknesses. 9.1.2 The Method of Analysis will be chosen by the user from the two options of Ultimate Strength Design (USD) or Working Stress Design (WSD), both in accordance with American Concrete Institute Standard No. 318-77 (ACI 318-77) with notation from the 1971 edition (ACI 318-71 with 1976 Supplement). This standard will be modified by the requirements of EM 1110-1-2101, "Working Stresses for Structural Design," for WSD. The label "hydraulic" or "nonhydraulic" will be set as described in paragraph 1.2.3.2 and used to select alternate sets of default values for analysis parameters. These values may be changed by input data. 9.1.2.1 Working Stress Design procedures to be used will be in accordance with the "Alternate Design Method" described in paragraph 8.10 of ACI 318-77. The equations and default parameters to be used are shown in Exhibit E.

9.1.2.2 Ultimate Strength Design equations and default parameters are shown in Exhibit F for nonhydraulic structures. These equation and parameters for hydraulic structures will be issued when completed and approved by OCE.

9.1.3 Pressures for Structural Design will be as described in paragraph 1.2.0d. Stems will be designed for pressures selected as described in paragraph 9.3. Heel, for, and key slabs will be designed (or pressures derived from the overturning stability analysis described in paragraph 5.2. The program is to be organized to permit use of load factors in ACI Ultimate Strength Design (USD) procedures, but implementation of this will be postponed until after completion of current research in USD of hydraulic structures. Stability analysis and working stress design will be for the summation of the actual loads.

9.2 REINFORCING STELL DETAILS

9.2.1 The Primary Reference is Chapter 7 of ACI Standard 318-71 with the 1976 Supplement. This is supplemented and amended for hydraulic structures by EM 1110-2-2103, "Details of Reinforcement--Hydraulic Structures," whenever the EM is more restrictive.

9.2.2 Default Values for Concrete Clear Cover over reinforcing steel will be according to the following table:

	Structure Type		
	Ordinary	Hydraulic [HSR]	
Stem, heel-side face	2 inches	3 inches	
Base, top face	2 inches	3 inches	
Base, bottom face	3 inches	4 inches	

9.2.3 Compressive Reinforcement will not be used, for two reasons:

a. The bar cover requirement, especially for hydraulic structures, forces any compressive reinforcement so close to the neutral axis that it is very inefficient.

5. Consideration of the implications of ACI 318-71 paragraph 9.12.5 and the corresponding Commentary leads to the belief that lateral reinforcement would be required to keep compressive reinforcement from popping out of the surface of the slab.

9.2.4 Placement of <u>Reinforcing Steel</u> - The user will have specified in the input data the maximum bar size and the minimum bar spacing. The program will check whether one such layer will be sufficient for the required area of steel per foot. If not, the program will place as much



[HSR]

steel as possible in the first (outer) layer and the remaining steel in the second layer (and a third layer if needed). The program will then check to see if the resulting actual effective depth d is sufficient. If it is not sufficient, the program will increase the concrete thickness as needed. The user will always have the option of looping back through a new stability analysis or design. The program will not attempt to place and cut off bars, but will show the required steel area and bond requirements per foot at suitable intervals across the base and up the stem.

9.3 STEM LOADINGS

9.3.1 <u>General</u> - Earth and other pressures for stem structural analyses will be calcualted as described in Section 4, applied directly to the face of stem as described in paragraph 4.2. Thus, surcharges located over the base will be considered to be vertical loads for stability analysis but will be considered to cause horizontal pressures applicable to stem stress analysis.

9.3.2 <u>Alternate Loading</u> - It will be possible for the user to substitute his own stem design pressures for the ones calculated by the program. Such a substitution will not change the pressures used for design of the base slabs.

EXHIBIT A: COMPARISON OF COULOMB AND INCREMENTAE WEDGE EARTH PRESSURE VALUES

This exhibit is a report by Dr. Michael W. O'Neill, Assistant Professor of Civil Engineering at the University of Houston. The report presents the calculation procedures he will use in a subroutine to calculate active earth pressures that will be a part of modules SA and SP of TWDA.

Procedures Followed

1. Coulomb Active Earth Pressure Diagram and Total Active Force:

A. Following the procedure described in Addendum A to this Exhibit, compute the active Coulomb earth pressure immediately above and immediately below each change of stratum, except for the surface stratum line (for which calculations are not made above the line) and a horizontal line drawn through the base of the wall (for which calculations are not made below the line).

B. Obtain the horizontal component of the earth pressure by multiplying each earth pressure computed in Step A by the cosine of the angle of wall friction at the depth at which the calculations are being made (Equation 2 of Addendum A). Add hydrostatic pressure if appropriate.

C. If surcharge loads exist, obtain the added lateral pressures due only to the surcharge at the depths described in Step A and at intermediate depths where required by a high stress gradient using the equations presented in Addendum B. Add these pressures to those obtained in Step B and connect the points to give a continuous pressure distribution.

D. Integrate the horizontal pressure diagram constructed in Step C from the top to the base of the wall to obtain the total active Coulomb horizontal wall force ${\rm P}_{_{\rm AHC}}$.

II. Trial Wedge Active Earth Pressure Diagram and Total Active Force:

A. Establish several nodes along the wall. Each node will be a point for which the active force acting above the point will be computed.

Five equally spaced nodes were used in this comparative study.

B. For each ith node, lay off several straight lines (rays) at 5- to 10-degree intervals extending to the ground surface. See Addendum C. Compute the active force on the wall for the wedge defined by each ray, the wall, and the ground surface, including the weights of all surcharge loads between the wall and the point of intersection of the ray with ground surface, using the method shown in Addendum C.

C. Select the active force for Node i as the maximum value obtained from all rays through Node i. Denote the horizontal component ${\rm P}_{\rm AHTi}$.

D. Determine the increment of active force acting between Nodes i and i-l as $P_{AHTi} - P_{AHTi-1}$. Divide the result by the vertical distance between Nodes i and i-l. The resulting pressure is the average pressure between Nodes i and i-l, which is assumed to be constant between the nodes.*

E. Obtain the total active norizontal trial wedge wall force P_{AHT}^{-} by integrating the pressure diagram obtained in the previous steps from the top of the wall to the base.

Discussion of Comparisons

For Cases 1, 3, and 6, the Coulomb and trial wedge procedures should give identical results. The differences obtained in the pressure diagrams and total horizontal active forces are due to minor computational errors in the graphical procedure for obtaining the trial wedge values and to the means of displaying the results (i.e., step functions for trial wedge as opposed to continuous functions for Coulomb). By increasing the number of nodes, the trial wedge solutions will converge to the Coulomb solutions.

For Cases 2 and 7, the Coulomb procedure yields higher values of active pressure primarily due to the method of handling surcharge. In the trial wedge approach, the surcharge weight contributes to and is

* Wu, T. H., Soil Mechanics, 1st ed., Allvn and Bacon, 1966, p 278.

included in a plastic failure model (the wedge). In the Coulomb analysis, the pressures due to the backfill are computed assuming plastic failure conditions, but the surcharge effects utilize elastic theory to compute the added pressures due to surcharge. Combination of the plastic and elastic methods in the latter case produces an inconsistency which yields an overestimate of the lateral stresses.

For Cases 4, 5 and 8 (and for Case 9 as well), use of the Coulomb active pressure coefficients in a $\phi = 0$ soil (clay) is equivalent to using Rankine-Rosae coefficients. That is, the Coulomb equations as commonly employed offer no advantage over the Rankine equations for use with purely cohesive soil and, in fact, lead to an inaccurate estimate of the active stresses within the zone of purely cohesive soil. When the backfill surface is horizontal, Coulomb active pressures in the clay are too high (Cases 4 and 5), and when the backfill surface slopes upwards, the Coulomb active pressures are too low.* Further, the standard Coulomb equation for active pressure coefficients does not contain a provision for specifying wall adhesion as a function of soil cohesion. Only an angle of soil-wall friction can be specified. Thus, not only are the Coulomb equations inaccurate for pure clays ($\phi = 0$), they are also inconvenient to use since wall adhesion must be converted to an equivalent angle of wall friction, a process for which few criteria exist.

For Case 9, the Coulomb approach is quite approximate since the active earth pressure equations do not permit modeling of a broken back-fill surface. The degree of correlation with the trial wedge approach is dependent on the engineer's choice of the equivalent backfill slope, a factor obtainable only through subjective means.

* When the ; , the Coulomb and Rankine active pressure coefficients are complex numbers. The value used in computations is the real part of that number, which is always approximately equal to unity regardless of the slope of the surface.

Addendum A

Procedure for obtaining Coulomb active earth pressure at a given depth (Symbols explained on following sheet)

1. Compute:

$$K_{a} = \frac{\sin^{2}(d+\phi)}{\sin^{2}d\sin(d-\delta)\left[1 + \sqrt{\frac{\sin(d+\delta)\sin(d-\beta)}{\sin(d+\beta)}}\right]^{2}} \dots \dots (1)$$

For clay backfill, wall adhesion is considered by using a value of δ of 10° and $\phi = 0$. For this special case, when $\alpha = 90^{\circ}$:

$$K_a = Sin (90 - 6) \left[1 + \sqrt{\frac{5in (6) sin (-8)}{5in (90 - 8) sin (1 + 8)}} \right]^2$$

and when a horizontal backfill exists $\beta = 0$, so that

$$Ka = \frac{1}{\sin(90-5)}$$

NOTE: The above coefficient approximates the Rankine value.

NOTE: The properties ϕ and δ are those exactly at the depth under consideration.

2. Compute the active pressure (horizontal component):

$$P_{h} = \left[\overline{P}_{s} K_{a} - 2c V \overline{K}_{a} \right] \cos \delta + \delta_{w} \overline{Z}_{w} \dots (2)$$

NOTE: Again, c and δ are the properties of the soil and interface exactly at depth under consideration.

¹Bowles, J. E., Foundation Analysis and Design; McGraw-Hill, 1968, p. 270. ²Ibid, p. 284


NOMENCIATURE FOR COULOMB EQUATIONS

Addendum B

Lateral earth pressure on walls using theory of elasticity⁽¹⁾

To the lateral pressures calculated from the use of the Coulomb equation the following lateral pressures are added: C_{1} , A_{2} , A_{3} , A

CaseA: Line Lood



¹Bowles, J. E., <u>Foundation Analysis and Design</u>, McGraw-Hill, 1968, pp. 297-303.

Addendum C



W = weight of Block A + weight of Block B tweight of Block C + Q + Q (X4) Cohesion (C): C = C, l, + C, l, + C, l.

*Buoyant below water table.Weights of Blocks (including effects of buoyancy) denoted by W_A , W_B , W_C .

Compute P_{AHTi4} (horizontal component of total active wall force, Node i, Ray 4) $P_{AHTi4} = \overline{P}_{AT} \cos 5 + \gamma_{W} (\overline{Z}_{1W} + \overline{Z}_{2} + \overline{Z}_{3})^{2} (0.5)$

where γ_{ω} = unit weight of water.

¹Bowles, J. E., Foundation Analysis and Design, McGraw-Hill, 1968, pp. 291-295 (Basic Vector Diagram and Mechanical Procedures)

²Terzaghi, K., <u>Theoretical Soil Mechanics</u>, Wiley, 1943, pp. 95-99 (Stratified and Cohesive Backfill Considerations).



Pressure comparisons













.



الألفية والم









A-14

Addendua E

Calculation worksheets for Addendum D:

Cases 1, 2, 3

Cases 4, 5

Case 6

Case 7

Case 8

Case 9











. **1**2444

TRIAL	WEDGE
WEIGHT	1 FOR

CASE 11213			
		<u>C] +3</u>	<u> </u>
PT B Ray 1: $W = 2 \times 5 \times 10 \times 5 \times 550$		5.70	510
Prover W= J×5×110 Y.5= 825		825	125
Roy 3: W= 475 × 110 ×. 5- 1100		100	1100
Paul W = Ex5 x110 x.5 (1000)=	2375	1275	2375
Kins: W= 6x5 x110x, F 41 mg	2650	1610	2650
Rub' W= 2.5×5×110×5		963	163
(bel : 13)			
	<u>cı</u>	(2	<u><u> </u></u>
$B_{1} \subset \cdot B_{2} := 10 (6)(10)(5)$	3300	4500	28:2
-67.4/2)(5)(.1)			
(† 1000)			
$P_{2} \rightarrow V(z + p(z)/\mu_{0})(z)$	3850	1950	3304
(+1))			
Pa 7. 11- 10(0 (11) (5)	4400	5400	3776
(- 1,2,4 (u.t.) (r)) ²			
(1000) V			
$\frac{1}{100}$	2750	3750	2310
$\frac{1}{4} \frac{1}{4} \frac{1}$			

CHSES 1, 2, 3 (contid)			
PT.D.	<u></u>	<u><u> </u></u>	
$R_{av} = 15(7)(10)(15)$	5775	1775	4307
$\int -\frac{1}{2} \left[-\frac{1}{2} \left[\frac{1}{2} \left[\frac{1}{2} \left[\frac{1}{2} \right] \right]^{2} \right]$			
11000			
$R_{11} 2 V W = 15(9)(10)(15)$	7425	8425	5 552
@[[]]/(/////////////////////////////////			
41000			
R_{m+3} , $W = 15(11)(110)(15)$	9075	10,075	6797
PT.F Ray 1: $W = 20(10)(110)(.5)$	11,000	12,000	7420
0[(2)/(25)/(2)/4.5)]	,		
Ги- тЮ			
[11000]			
Ray 2: W= 20(12) (110) (15)	13,200	14,200	
$[-62.4(15)(9)(.5]]^{D}$			
(tlum)®			
		••••	
(ay 3: W= 20(14) (110) (15)	15,400	1/2,000	10,486
[-62.4 (15) (i) (.5)]			
[+1000]			
	17,600	18,600	11,948
[-62.4 (15)(12)(1)]			
[1000] ²			

CALCULATIONS OF HORIZONTAL PRESSURES

CENTER OF \$ TEP	,	DPAT		A PANT (= APAN (A) 20)		(=) Pur (5')				
	Cı	دع	C3*	ĊI	CZ.	c3*	CI	cz	c3*	Care
2.5'	420	600	420	395	564	395	79	113	79	79
7.5'	1180	1500	980	lioq	1410	921	222	282	184	340
12.5'	2100	2150	1350	1973	2020	1269	397	404	254	72B
17.5'	2900	2800	1750	2725	2631	1644	545	526	329	11019

.

* DOES NOT CANTAIN HYDROSTATIC WATER EFFECT.

```
AX INCLUDING " "
```

WATER PRESSURES FOR CASE 3:

7.5': $62.4(5)^{2}(0.5) = 780^{4} \implies 156psf$ (step value for 5'-10') 12.5': $67.4(10)(0.5) = 780 = 2340^{4} \implies 468psf$ 17.5': $67.4(15)^{2}(0.5) = -780 = 3900^{4} \implies 780psf$

TOTAL TRIAL WEDGE FOR CES FROM & of S oF PRESSURES (MAY ALSO BE ODDAINED DIRECTLY FROM LART (PTE) WORKE FORCE) HORIE. COMPANIENTS :

হা	PAHT CZ	<u>C3</u>
79+5		
222+5		
397 45		
545 8 5		
6215 16	6625 16	11,25016.

$$\frac{COULONB PRESURF}{DISTRIBUTIONS}$$

$$\frac{CASE 1}{DISTRIBUTIONS}$$

$$\frac{CASE 1}{ad 20': K_{a} = 0.297}$$

$$p_{n} = 110(20)(0.297) = 653 \text{ psf}$$

$$p_{n} = 653 \cos 20^{\circ} = 6/4 \text{ psf} (20')$$

$$\frac{P_{AHC} = -5/40}{P_{AHC} = -5/40} = \frac{6}{10}$$

$$\frac{CASE 2}{COULONE} = \frac{100}{COULONE} = \frac{100}{COULONE} = \frac{100}{COULONE}$$

$$\frac{CASE 2}{COULONE} = \frac{100}{COULONE} = \frac{100}{CO$$

COUWHB PRESSURE DUTRIBUTIONS

at su

с (1.

$$\frac{CASE 3}{at 5!} : K_{a} = 0.297$$

$$p_{h} = 110 (5) (0.297) \cos 20^{\circ} = 154 psf$$

$$at 20': \overline{p}_{a} = [110 (5) + 47.6 (15)] (0.247) \cos 20^{\circ} = 353 psf$$

$$u = 67.4 (15) = 936$$

$$p_{AHC} = \frac{154 (5) (.5) + (154 + 1289) (15) (.5) = \underline{11,208 lb}$$







Ŧ

<u>ئ</u> _ د

17 M

- 1 - 2

Why We Wa CASE 4 (novelies for PTB) PIC USE WEIGHTING Ray 1 W. = 250 2250(0) + 82560) $W_2 = 825$ = 7.33" W=3775 S = Jana = 150 Roy 2 W. = 2975 $b = 7.33^{\circ}$ $\delta = 15^{\circ}$ $W_{2} = 963$ W= 3928 $R_{34} 4$ $W_1 = 3825$ φ= 7,33° ' δ=15° W2 = 1228 W=5062 Ray 3 W, = 3400 φ= 7.33° δ= 15° W_ = 1100 W = 4500 Ray 5 Wi = 4250 1375 $\phi = 7.33^{\circ}$ $\delta = 15^{\circ}$ W2=1375 W= 5625

PT. D. Cose" $\frac{R_{ey}}{W_{1}} = 7.5 \text{ MO } \text{ X120} \text{ S} \text{ F} 7.5 \text{ X5 } \text{ X110} = 8625}{= 963}$ $V_{12} = 3.5 \times 5 \times 10 \times 0.5$ 9.63 W = ,9588 18625(0) + 963 (33) 5585 = 3.0° $\delta = 2(10) + 1(20) \neq 3 = 13.3^{\circ}$ Roy 2 W, = 8.7 XIOX120X015 + 8.7 X 5×110 = 10,005 = 1155 W2 = 4.2 XT XILOXOIT 11,160 $p = \frac{1155}{11160} (20) = 3.0^{\circ}$ 5=13.30 Ray 3 W = 10 x10 x170x0.F + 10 xF y110 = 11, 100 = 1375 VIL = 5XF XIIOX.F W=12875 q = 3.2° 0 = 17.7° Ray 4 W= 113 x > x 12 0 YO. (+ 11. J X (X 110 = 12, 945 WL = 5.7 × 5 XHO × 9.5 = 154" 14,563 A-60 \$=3.20 8=13.3°

Point I Care 4 _____!', <u>____</u>___ V V = 5x5+05x115 + 5x10x120 + 5x5x110= 10,188 W2 = 10 X10 x 120 + 10x5 x110 = 11,500 $W_1 = \frac{1375}{1375}$ W= 23,063 137: (35+ 6,182/23) 10,6 ¢ = $\mathcal{E} = 1(20) + 2(10) + 1(13) = 13.1^{\circ}$ + 1-2 (200) = 354016 2 1. 1 1. x + 115 x ... + 5.5 X12 X120 - 5.5 X 5 X10 = 11, 206 W_ 11 x : 5 x : 5 x : 20 + = 11 x 5 x : 10 = 12,650 $V_{1} = 5.5 \times 102.5$ = 1513 W = 25,370 11,205 (20) + 1513 (20) = 10.6° 1º 35 370 8 = 131° 0, 7. 1990 He ____ - ... _ _

But E - Conel Paris Encore HAN PATIS W. = 4.5 x5 × 115 x0,5 + 45(10)(120) + 4.5(5)(10) = 9149 $W_2 = 9 \times 10 (120) (0.5) + 9(5) (110) = 10350$ = 1238 $W_2 = 4.5(5)(10)(5)$ W = 20,757 $\phi = \frac{9[69(20) + 1238(30)}{60,757} = 10.6^{\circ}$ S = 13.1° $C_{1} = 3370$ Ray 4 (ITERER THAN 1+2) W= 4 x5 x 115 x 0.5 + 4 (10/(120) + 4 (1) (110) = 8150 9200_ $W_{1} = 8 \times 10 \times 120 \times 5 + 8(5)(110)$ 2 1100 W, = 4×5 × 110 ×.5 18,450 $\phi = 10.6^{\circ}$ $\delta = 13.1^{\circ}$ $C_s = 3205$ ____ _____





$$\frac{COULONB PRESSURES}{CASE 4}$$

$$Q' = 0$$

$$\frac{5'}{(100)}; \hat{\delta} = 20^{\circ} \qquad K_{e} = 0.247$$

$$\delta = 110 \text{ pcf}$$

$$p_{H} = 5(110)(0.297) = 163 \text{ psf}_{A} \text{ cos } 20^{\circ} = 154 \text{ psf}$$

$$\frac{5'}{(200)}; \hat{\delta} = 0^{\circ}$$

$$C = 200 \text{ psf}$$

$$(\delta' = 120 \text{ pcf})$$

$$K_{c} = 1.015$$

$$p_{H} = 357 \text{ psf } \cos 10^{\circ} = 352 \text{ psf}$$

$$\frac{15'}{(clay)}; K_{e} = 1.015$$

$$p_{H} = [5(110) + 10(120)](1.015) - 2(200)(1.015)^{Y_{e}}$$

$$= 1373 \text{ psf}$$

$$p_{H} = 1373 \cos 10^{\circ} = 1352 \text{ psf}$$

$$\frac{15'}{(clay)}; K_{a} = 0.44$$

$$p_{H} = (1750)(0.44) - 200(0.44)^{Y_{h}} = 637 \text{ psf}$$

$$\frac{15'}{P_{H}} = [1750 + 5(115)](.44) - 200(.44)^{Y_{h}} = 890 \text{ psf}$$

CASE 5 WORKSHEET
NOTE ! ONLY THE WEIGHTS ARE DIFFERENT FROM CASE Y. .'. USE CASE & DIAGRAMS WY MODIFIED WEIGHTS. (RED LINES
PT3 A+B - same an for Case 4
<u>ere</u>
$K_{ay} 1! W_{i} = 2550 - 62.4(5)(3)(.7) = 2082$
W = 2907
Ray 2: W, = 2975-62.4(5)(35)(15) = 2429
$W_2 = 963$
W = 3392
Ray 3: W, = 3400-62.4(5)(4)(5)= 2776
$W_2 = 1100$
W = 3876
Ray 4: $W_1 = 3825 - 42.4(4.5)(5) = 3123$
$W_2 = 1222$
W = 751 $R_{01} = 1$ $W = 4250 - (2.4 125)(5) = 3470$
$W_{2} = 137\Gamma$
w = 4845

2.50

.

1

A-32

.

$$\frac{P\Gamma D}{R_{av}} = 6225 - 7.5 \times 10 \times .5 \times 62.4 = 6285$$

$$W_{z} = \frac{963}{7264}$$

$$W = 7264$$

$$\psi = 763(30) = 3.97^{\circ} \approx 4^{\circ}$$

$$\delta = 13.3^{\circ}$$

$$R_{av} z$$

$$W_{1} = -8.7 \times 10 \times 62.4 \times .5 + 10,005 = 7291$$

$$W_{2} = \frac{1155}{9446}$$

$$R_{av} 3$$

$$W_{1} = 11,500 - 10 \times 10 \times 62.4 \times .5 = 8580$$

$$W_{2} = \frac{1375}{9755}$$

$$R_{av} 4$$

 $W_1 = 12,99\Gamma - 11.3 \times 10 \times 62.4 \times 5 = 9469$ $W_2 = \frac{1568}{11,037}$

$$\frac{PTE}{Ray 1}: W_1 = 10,188 - 25(.5)(.2.4) - 50(.2.4) = 6288$$
$$W_2 = -100 \times 5 \times 62.4 + 11,500 = 8380$$
$$W_3 = 1375$$

- 4

al is shirt

W= 16043

W = 14439

$$\varphi = \frac{1775(39) + 6268(20)}{16043} = 10,4°$$
 for all pre

$$\begin{array}{rcl} Pay & 2 &: W_{1} = & 11,206 - 5.5 \times 5 \times 5 \times 5 \times 10 \times 10 \times 12.4 = & 6916^{*} \\ W_{L} = & 12,650 - & 11 \times 10 \times 5 \times 62.4 & = & 9218 \\ W_{2} = & & = & \frac{1513}{17647} \\ W_{2} = & & & 1513 \\ W_{3} = & & & 169 - 4 \times 5 \times 5 \times 62.4 - & 4.5 \times 10 \times 62.4 = & 5659 \\ W_{2} = & & & 10350 - & 9 \times 10 \times 5 \times 62.4 & = & 7542 \\ W_{3} = & & & & = & 7542 \\ W_{3} = & & & & = & 7542 \\ \end{array}$$

Case 5 Horiz. Forces + Strosses
2.5': 422*
$$T_y = 84psf$$

7.5': -450 (cos 20°) + 1600 (cot 1?') = -422 + 1545 = 1123
+ 62.4 × 5² × .5 = 780
Ray = 1903* $T_x = 381psf$
12.5': -1600 cos 15° + 4100 cos 13° = -1545 + 3485 = 2450
+ 62.4 × 10³ ×.5 - 780 = 2740*
PAHT = 4790 $T_x = 958psf$
17.5': = -4100 cos B + 6300 cos 13 = -3995 + 6139 = 2144
+ 62.4 × 15² ×.5 - 2240 - 780
PAHT = 6044
 $T_x = 1209psf$

Backlander - Dist

$$\frac{5'}{(sound)}; \quad p_{N} = 5(110) (K_{a}) = 163 \text{ psf}$$

$$\frac{p_{n}}{p_{n}} = p_{N} \cos z_{0}^{\circ} = \frac{154 \text{ psf}}{1.015}$$

$$\frac{5'}{(clary)}; \quad p_{N} = 5(110) (K_{a})^{-2} - 2C (K_{a})^{-1.015} = 155 \text{ psf}$$

$$\frac{p_{N}}{p_{N}} = 15.5 \cos 10^{\circ} = \frac{153 \text{ psf}}{153 \text{ psf}}$$

$$\frac{|5'|}{(c_{LAY})} = \overline{p}_{N} = [5(110) + 10(57.6)](1.017) - 2(200)(1.017)^{V_{L}}$$

$$= 740 \text{ psf}$$

$$\overline{p}_{N} = 729 \text{ psf} (\overline{p}_{N} \cos 10^{\circ}) \qquad \frac{-62.4}{57.6}$$

$$U = 10(62.4) = 624 \text{ psf}$$

$$\frac{15'}{p_{N}} = \overline{p}_{N} + U = \frac{1353 \text{ psf}}{1353 \text{ psf}}$$

$$\frac{15'}{(SAND-CLAY)} = [5(110) + 10(57.6)](0.44) - 200(0.44)^{V_{L}} = 343 \text{ psf}$$

$$\overline{p}_{N} = 343 \cos 13^{\circ} = 353 \text{ psf}$$

$$u = 624 \text{ psf}$$

$$\frac{1}{20'} = \bar{p}_{h} + u = 977 \text{ psf}$$

$$\frac{20'}{[54007-CLAY]} = \bar{p}_{h} = [S(110) + 10(C7.6) + 5(C2.6)](0.114) - 200(0.44)^{V_{2}} = 476 \text{ psf}$$

$$\bar{p}_{n} = 478 \cos 15' = 466 \text{ psf}$$

$$u = 15(62.4) = 936 \text{ psf}$$

$$\frac{1}{2}h = \bar{p}_{h} + u = 1402 \text{ psf}$$



1 1

> FOR THIS CASE, IT IS ONLY NECESSARY TO COMPUTE TOTAL TRIAL WEDGE FORCE FOR PURPOSES OF COMPARISON. PRESSURE DISTRIBUTIONS SHOWN AS STOP FUNCTIONS FOR CONSISTENCY IN STHER CASES.

> > A-37



CASE 6

Ì






Weights Case 7 PT B W = 2 x 5 X 10 X 5 = 550 Ray 1: "d 2} W= 3 XF X110 X. 5 = 825 W= 4 V5 x110 Y. F = 1100 21 W= 5xt x1104.1 = 1375 * 4; " Ξ ; W= 6 XI X 110 X. 5+200 = 1850 PIC: W= 10 x 5 x110 x,F = 2750 Ray " 2 : W = 10 x 6 x110x, 5+200 = 3500 " 3 : W = 10 ×7 × 110× 5 1900 = 4250 W = 10x4 x110x, 5 +600 = 5000 : Ч: " 5: W= 10x9 ×110x.5+800 = 5750 W: 15× 11×10 x.5 : 1700 = 10275 King I W= 15x12 110 X.5 + 1400 = 11 300 2 W= 15× 13 ×10 × 1 + 1600 = 12325 • W1 15x 14 X1:3 x.1 +1800 - 13 350 15x 15 X 110 X. 5 + 7000 - 14375 Ξ. <u>ए</u>] = <u>P1(:____</u> Empl: W= 20 × 10 × 10 ×.5 + 1602 = 17200 W 20 X 15 X110 X.5 + 2020 18 500 2: 11- 27 116 X117 X.5 1 2000 : 19600 15900 ----· u W

CASE 7 - Trial Wedge horizontal forces and stresses 2.5': PAH = 430 cos 20° = 404# Ph = 404/5 = 81 psf 7.5': PAH = 1850 cos 20 -404 = 1334# Pn = 1334/5 267 psf 12.5'; PAH = 4300 cos 20 - 404-1374 = 2303 p. = 2303/5 461psf 17.51: PAH = 7450 20120 - 404-1314-2303 2960 -Pn = 2960/5 592 psf PAHT = 7001 # COULONB PRESCURES KA = 0.297

20':
$$p_h = 20(110)(0.297) \cos 20^\circ = 6/4 \text{ psf}$$
 (linear)
addied pressure due to surchange - next page
 $\underline{P_{A:i..}} = 6/4 (10) + 2.5(19.6 + 94.1 + 45.7 + 38.3 + 29.0 + 21.5 + 16.23.7 + 16.23.7 + 16.23.7 + 16.7$



PT,	B (roc)	sīn B	sir ² x	co: K	(fe + sing rin w - sing wird)	A Jh (pof)
1	0.157	0,156	0.985	6.0149	0,308	19.6
2	0.401	0.391	0.872	0,128	0.692	44.1
3	0.506	0.485	0.719	0.281	0.718	45.7
4	0.52.4	0,500	0.572	0.422	0.602	38.3
5	0.506	0.475	0.442	0.552	0.456	29.0
6	0.480	D.467-	0.345	0.655	0.337	21.5
.,	0.454	0.436	0.273.	0.127	0.255	16.23
2	D 410	0.279	0.220	0.740	0.187	11.9









CASE 8 CON'T





......

HORIZ FORECTS I FURICISES (TRIAL WEDGE).
0.5:
$$F_{H} = 507^{*}$$
 $p_{h} = 101 \text{ psf}$
5-10: $F_{H} = 500 \text{ cosl} 5 - 507 = 0^{*}$ $p_{h} = 0 \text{ psf}$
10-15: $F_{H} = 4150 \text{ cosl} 13.9^{\circ} - 483 = 12$
 $4039 - 463 = 3556^{\circ}$ $p_{h} = 711 \text{ psf}$
15-20: $F_{H} = 10,900 \text{ cosl} 13.3 - 4039$
 $= 6569^{\circ}$ $p_{h} = 1314 \text{ psf}$
 $P_{AHT} = 507 + 0 + 3556 + 6569 = 10,632^{\circ}$

Care 8 - Weights
PT D:
Ray 1: (6×3-1×3) 0.5×10= 5×3×0.5×110= E25#
2:5x4(1)x110 = 1100
3 : 5×5 (.5) ×110 = 1375
4: 5×6 (.T) ×110 = 1650
$5 : 5 \times 2 \times 10 \times .7 = 50^{4} \phi = 20^{\circ}$
5=20°
PTC. S= 15.
Ray 1: W= 5×2.2 × 0.5×120
+5x2.2 x 110
$+ 0.8 \times 2.2 \times 1/0 \times 5 = 1870$
$W_{8} = 5.8 \times 2.8 \times 0.5 \times 10 = 873$
$W + 1 = 2763^{+}$
φ=9.7°
Rey 2: W = 5x 2, 5x 0.5 x120
+5 x2.5 X 110
$+0.5 \times 2.5 \times 10 = 2279$
$W_{1} = 5.85 \times 3.3 \times 5 \times 10 = 10.62$
$W = 3341^{"} p = 9.1$
Roy 3: VJ = 5X 2.8x 0.5 X120
1 +5 ×2.8 × 110
$+0.9 \times 2.8 \times 5 \times 10 = 2519$
$W_{2} = 5.9 \times 4.1 \times 5 \times 10 = 1330$
\overline{W} = 3949 ϕ =10

¢

Core & weights (could) ·17-C: Kay 4: W, = 5x 3.2 x 0. F x120 +5× 3,2× 110 + 1,1 × 3.2 × ,5 ×110 = 2914 W. = 6.1 × 4.9 ×.5 ×110 = 1644 4558 Ø=11.0 W Ξ Roy 5: W, = 5 x 3.5 x9 5 x120 15 X 3.5 X 110 3206 +1.2 × 3. 5 × 0.5 × 110 = W3 = 6.2 x 5.5 X0.5 X110 = 1876 5082 p:11.13 W) Ξ p=? (reparately for co. ch.) (sue above) 5=15° C = 5.4×500 = 2700 * C = 5.5 × 500 = 2750 <u>(== 5,7 x 500 = 2850</u> Cy = 5.9 ×500 : 2950 C= = 6.1 ×500 = 3050

CASE 8 WETS CONT. PTD: 5= 13.30 Ray 1: W= 10x55×120 ×05 + 5 X 5,5 X 110 +1.8 x 5.5 x 110 x 0 5 = .6 870 $W_{1} = 6.9 \times 4.5 \times 110 \times 0.5 = 1708$ $W^{2} = \frac{1708}{8578} 3^{2} = 6.0^{2}$ C=5700 011X 5.3 X 2 + 7846 + 21 × 62 × 110 × 25 = W2 = 7.1 x 5.6 X110 x 2.5 = 2187 $W = \frac{2187}{1007} = 6.5^{\circ}$ 10,033 loy 3: W, = 10 x J.2 x 100 x0.5 + 5 x7,2 X110 + 2.4 ×7.2 ×110 × 95 = 9230 W2 = 7.3 × 6.8 × 110 × 0.5 = _2730_ Puy 4: W, - 10 x2 x120 x25 + 5×8 x 110 t 2.7 x 8 × 110 × 0.5 = 10358 W, = 7.6 × 8×110×0.5 = 33+4 $W = \frac{13,732}{\phi}$ 5= 6375

Case & Wets (Cont'd) Pi 12 (cot'd) Rays: W= 8.6 X10 K.5 X120 +8,6 x5 - X 110 +8,6×2,9×110×,5 = 11262 W2 = 7.9 × 9.3×110 ×.5 = 4041 C= 6400 15303 W = \$=7.90 Noy 6: W = 9.2X10X.5.120 + 9,245 × 110 + 9.2 Y3 X 110 X,5 = 12098 C= 6450 W2 = 8.1 x, 5 x 10,7 x 110 = 4767 $W = \phi = 8.5^{\circ}$ 16865 PT E 15= 13.3° Poyl W = 4 × 5 × 115 × 5 +4 x10 x 120 4-4X5 × 110 + 4 X1.3 X 110X.5 = 8436 $C = \frac{1205}{645^{\circ}} + \frac{7.8 \times 10 \times 120 \times 5}{7.8 \times 6.3 \times 110} + \frac{7.8 \times 2.6 \times 10 \times 5}{110 \times 5} = \frac{11201}{110}$ 10 W3 = 8.9 X 9.3 X110 X.5 = 4552 24187

Ray 2 W, = 4,2 Y5 X115 X.5 + 4,2 ×10 ×120 + 4.2 X5 X 110 Ø= 12.8° + 4.2× 1.7×110×.5 = 8858 C=6550 W2= 8,3× 10×120×15+8,3×6.4×110 W = 9.1 y10.5 x10x5 S 12 056 5255 W-4.5 x 5 X115 X.5 Ray 3 14,5 X10 X120 +4.5 X5 X110 +4:5×1.4 ×110×,5 = 7515 Q= 13,0° W, = 8, B XIDX120X.5 C=6700 + 8.8 × 6.5 × 110 18.8 x 2.8 x 110 x ,5 = 12927 W3 = 9.4 ×11,7 ×110×.5 = 6049 = 28491 W $W_{1} = 4.8 \times 5 \times 115 \times 5$ Ray 4 +4.6 X10 X120 +4,8 X5 X110 $1.4.8 \times 1.5 \times 10 \times 5 = 10,176$ 13.30 W, = 9.3 × 10 × 120×.5 19.3 × 6.6 × 110 C=6850 + 9,3 × 3 × 110 ×,5 = 13,866 W= 9.7 × 13 × ×110×5 6936 30,978 N=

Case 8 - Coulmb Stresses 5' sand: KA = 0.40 (b/0=0.6) p_h = 110(5)(0,40) = 225.5psf.cos200 = 211.9 21 cley: use Raubin mice Contout buch down for \$>\$ KA = cosp - cosp + (cor2) -1 3=18.435° Coz 18.4= ,9487 KA = 0.9487 . 9487 - VO.T = 0.9487 -0.12 .9487 -0.12 $= 0.9487 \frac{(1487)^2 - 0.01}{(9.487)^2 + 0.01} + -1$ Re = 0.948 $\frac{p_{n_{s'}}}{(521.4 - 97.6 - -452 psf)} = 0$ $p_{h_{19}} = [110(5) + 120(10)](.948) - 1000(.948)^{1/2}$ = 685.4 psf

Case &- Coulorilo Strepser 15' 50 CR KA = 0.81 $\frac{P_{h}}{P_{h}} = \left(\begin{array}{c} 0.81 \\ (1750) \\ (1417.r) \\ - \end{array} \right) - \begin{array}{c} 2.20 \\ (.81) \\ 160 \\ - \end{array} \right) \frac{1}{2} \begin{array}{c} \cos 13.3 \\ \cos 13.3 \\ - \end{array}$ $= \begin{array}{c} 1204 \\ 1204 \\ - \end{array} \right) sf$ 20' sa Cl K4 = 0.81 $\frac{p_{h}}{2} = \left[0.81 \left(2325 \right) - 180 \right] \cos 12.7^{\circ}$ = 1657 psf PAHO 1204 + 1657 2 × 5 Ξ 7152 -412+685.4 X10 1165 + 211.9 x r 530 -__ 8848



f k







WEIGHTS - CATES S= 13.30 $\frac{PTD - (1) W_{1} = (6.3 (10) (.5) (120)}{+ (6.3 (\frac{517.1}{2}) (110)} = 7973$ q= 10,114 10 = 615 $W_2 = 5.7(7.1)(110)(.5) = 2226$ C = 595010,199 (2) $W_1 = 7.3(10)(.5)(120)$ $+7.3(\frac{5+7.6}{7})(110) = 9439$ $\sqrt{5}=13.5^{\circ}$ - 2 (0,9) (110) (.5) = 2871 C= 6240 = 12,310 \mathbb{W} $(3) W_{1} = 8.4(10)(.5)(120)$ $\frac{+8.4^{(3+7.9)}(10)}{(7.9)} = \frac{11,000}{(7.9)} = \frac{5-13.70}{(7.9)(8.9)(110)(.5)} = \frac{7.2}{3447} = \frac{7.2}{-6575}$ = V.J 14,449 Ψ , = $\hat{\mathcal{I}}$, \mathcal{I} (10) (. \mathcal{I}) (120) $+7.5(\frac{1.7+5}{2})(110) = 12,597 \quad 5=13.3^{\circ}$ $W_{2} = (8.2)(11.3)(110)(.5) \qquad \phi = 7.4^{\circ}$ - 6(3)(110)(.5) = 4106 W = 16,703(5) W, = 10(10) (15) (120) $+10(\frac{3.115}{2})(110) = 13,215$ $\delta = 13.3^{\circ}$ $M_{1} = \frac{9.3}{(11.8)(110)(.5)} = \frac{9}{4419} = \frac{7.5^{\circ}}{C} = \frac{7}{125}$

A State of the sta

WEIGHTS CASE 9 (CONT'D)				
$PT = -DW_{1} = 3.7 (5) (115) (.5) + 3.7 (10) (110) + (5+12) (3.7) (110) = 7,783$				
$W_{z} = 7.4(10)(12c)(.5) + \frac{(12+8.7)}{2}(7.4)(110) = 10,504$	$\frac{\phi = 14^{\circ}}{5 = 12.3^{\circ}}$			
$W_3 = 0.4 (8.8) (110) + 9(6) (110)(.5) = 3744$	c = 8950#			
W = 22031				
$(2) W_{1} = 4.2(5)(117)(15)$ + 4.2(10)(150)				
$+\left(\frac{5+6.+}{1}\right)(1,2)(110) = \frac{8835}{1}$				
$W_{2} = \frac{7.3}{(\frac{6.4}{2} + 4.2)} (10) - (.6(.3)(.7)(10) - (.6(.3)(.7)(10)) + (.7)(10)(120)(.5) = 12,09.2$				
$W_3 = 9(7.3)(110)(.5) = 3614$	$\frac{p=11.6}{\delta=12.2^{\circ}}$			
W = 24,54				

1

Weight Core 9 (Cond'A) Pt. E $\bigcirc - W_1 = 5(4.5)(115)(15)$ +10 (4,5) (120) + 5.7r(4.5)(10) = 9540 $W_{2} = 9.1 (10) (12.0)(.5)$ $1 - 9.1 (\frac{1.5 + 3.7}{2})(110)$ - 1.8 (0.6) (110) (.5) = 13509 $p = 115^{\circ}$ $W_{z} = 8.15(9)(110)(.5) = .4034$ W_____ 27083 -(4) - W = 5(5)(115)(.5)+ 5 (10) (120) +5.85(s)(110) = 10,655 $W_2 = 10(10)(120)(.T)$ + 10 ((10) (10) 15,070 -(1)(31)(110)(.7) = $W_3 = 9(9)(110)(15) = 4455$ $W = 30100 \ \phi = 11.5^{\circ}$

TRIAL WEDGE STRESSES! 0-5 $T_{H} = 540 \cos 20^{\circ} \left(\frac{1}{5}\right) = 101 \text{ psf}$ 5-10 5- = 0 psf $10-15 \qquad \overline{V_{11}} = \frac{4180}{4180} \cos 13.3^{\circ} - 4.83 = 717psf$ $15.20 \quad (T_{h} = \frac{7000 \text{ cor-}13.3^{\circ} - 4069}{5}) = 1057 \text{ psf}$ $P_{A+1T} = 507 + 3585 + 5275 = 937016$ COULOMB STREPSER (USE EQUIVALINT BACKGOPE) 5 (SANO) : KA : 3 = 10° B/1 = 0.33 Kr. (Corps Manual) = 0.34 $T_h = 0.34(110)(5)\cos 20^\circ = 176\,psf$ ______ S' 15' LCLAL - SATURE AJ_CATE 8._____ 15' (SANO. WAY) - Ka= 0,50 (Coyn- Manuel) $\overline{U}_{n} = \left\{ \mathcal{D}, \Gamma \left[10(s) + 120(i0) \right] + 200 \right\}, s = 200 \sqrt{15}$ 714pst Case. - ?~ (Smid-Clay) Jn= \$ 0.5 [110 (5)+120(10)+115(-)]-2001.5 [con 13.3 994 psf.

CALCULATION OF PANC: (Come 7) _____ 176 (5) (12) - 685 + 計 (6.1)(1) _____ +(114+994)(5)(.5) = 680016

Addendum F

Errors inherent in trial wedge method

Planar sliding is a basic premise of the trial wedge method. In a homogenous soil system planar sliding is consistent with the failure mode assumed in other methods of analysis, such as the Rankine and Coulomb methods, and is essentially correct. In a stratified system planar sliding can be assumed to occur within each stratum, but the angle of failure will vary depending on the angle of internal friction of the various soils. Hence, the overall failure surface will be piecewise planar and not fully planar in the sense assumed in the trial wedge method. See Fig. AF-1. In such a case the maximum active force obtained from a trial wedge analysis will be slightly less than true maximum active force.

In order to evaluate the approximate upper limit of the error, an extreme case was studied, as depicted in Fig. AF-1. A wall that is completely frictionless is backfilled with strata of clay ($\phi = 0$) and sand (variable ϕ) of equal thickness. The total active force acting on the wall was evaluated from the Rankine earth pressure equation (correct solution) and from the trial wedge method (approximate solution). The apparent maximum error observed in the study was 15 percent.

Since the problem studied represents an extreme "mismatch" between backfill materials, the error for most practical problems

encountered should be less than that reported above. Nonetheless, the user is cautioned that such an inherent error does exist in using the trial wedge method in stratified soils.

A second type of inherent error occurs whenever stiff cohesive soil overlies cohesionless soil, the trial wedge method will yield unconservative results. An error occurs because the entire wedge is assumed to fail as a block in the trial wedge method, so the underlying cohesionless soil cannot fail and produce active forces until the weight of the wedge and overlying surcharge is sufficient to cause shearing failure in the clay. In a physical sense, active forces can be produced in the cohesionless soil before failure of the overlying clay occurs. Information relative to this condition is given in the discussion of Example No. 4, earlier in Exhibit A.



1.

----- Trial wedge failure surface (planar) ----- Possible true failure surface

c (Clay)	¢ (Sand)	8	PAHT	PRANKINE	4 EFPCR
30J p∍f	30 deg.	120 pcf	13120 lb.	14007 16.	ó
30J psf	35 deg.	120 pcf	12337 lb.	13625 lb.	9
300 psf	40 deg.	120 pcf	11715 lb.	13305 lb.	12
30∪ psf	45 deg.	120 pcf	11080 lb.	13030 1Ъ.	15
	Tabawash Romana	Andadaa fuur	Annumentan of D	Lanam Cliddon	

Fig. AF-1. Inherent Errors Arising from Assumption of Planar Sliding

$$\frac{\text{ERROR CONPUTATIONS}{2}$$

$$\frac{\text{CIT} 1}{2}: C = 300 \text{ psf} (clay) \\ \varphi = 30^{\circ} (stud) \\ \chi = 120 \text{ pcf} (bdk) \\ \frac{P_{AHT}}{Y} = \frac{13120 \text{ lb.}}{120 \text{ lb.}} (obtained by computer) \\ \frac{P_{AHT}}{P_{AAUYINE}}: \\ a. Sand Layer: top: $\mathcal{T}_{n} = 0 \\ bottom: \mathcal{T}_{n} = 1200 \text{ tann}^{\circ}(15^{\circ}-15^{\circ}) \\ = 1200 \text{ tann}^{\circ}(15^{\circ}-15^{\circ}) \\ \frac{10 \text{ ytoo}}{2} = \frac{200016}{2} \\ b. Clay Layer: top: \mathcal{T}_{n} = \chi h - 2c \\ = 120(10) - 600 = 600 \text{ psf} R_{n} \text{ cay} = \\ bottom: \mathcal{T}_{n} = 120(20) - 600 = 1800 \text{ psf} \frac{16000 \text{ lb}}{10000 \text{ to } 1500 \text{ km}} \\ \frac{110000}{2} \text{ Frank KINI} = \frac{14000 - 1521220}{140000 \text{ th}} \times 100 = 60.2869.5 \text{ say } \frac{670}{2} \\ \frac{1200016}{2} \text{ sao psf} (sout) \\ \chi = 1200 \text{ to the same term of t$$$

* Rankine and Coulomb Prossures are Identicol when There Is No Well Friction, ASIs the Case for This Problem

$$\frac{P_{RNYLINE}}{P_{RAUKINE}}$$
a. Sand Layer: top: $T_{n} = 0$
bottom: $C_{n} = 1200$ ton: $(45^{\circ} - 17.5)$
 $= 325$ psf
b. Clay Layer: no change: $P_{n,cuny} = \frac{12,000}{2}$
 $P_{RAUKINE} = P_{n,smo} + P_{n,cuny} = \frac{13625}{13625}$
 $P_{RAUKINE} = P_{n,smo} + P_{n,cuny} = \frac{13625}{13625}$
 $P_{RAUKINE} = \frac{13625}{13325}$ x loo = 9.453% SAY 9%
 $Y = 120$ psf (clay)
 $P_{RAUKINE}$: a. Sand Layer: top: $T_{h} = 0$
 $P_{RAUKINE}$: a. Sand Layer: no change: $P_{n,cuny} = \frac{12000}{2}$
 $P_{RAUKINE} = P_{A,smo} + P_{n,cuny} = \frac{153051}{2}$
 $P_{RAUKINE} = P_{A,smo} + P_{n,cuny} = \frac{153051}{2}$
 $P_{RAUKINE} = P_{A,smo} + P_{n,cuny} = \frac{153051}{2}$
 $P_{RAUKINE} = 2500 psf (clay)$
 $P_{RAUKINE} = 2500 psf (clay)$
 $P_{RAUKINE} = P_{A,smo} + P_{n,cuny} = \frac{153051}{2}$
 $P_{RAUKINE} = P_{A,smo} + P_{n,cuny} = \frac{153051}{2}$
 $P_{RAUKINE} = 2500 psf (clay)$
 $P_{RAUKINE} = P_{R,somo} + P_{R,cuny} = \frac{13030}{2}$
 $P_{RAUKINE} = P_{R,somo} + P_{R,cuny} = \frac{13030}{2}$

•

EXHIBIT B: DERIVATION OF EQUATIONS FOR SLIDING

1.7.1

Derivation of equations for uphill sliding as given in ETL 1110-2-184. Note that the horizontal component of the applied loads is not included in the derivation.



Vertical Section Through Sliding Mass on Sloping Surface

The satety factor $S = \frac{R}{S-f}$

Where R = resisting force (It is applied as a driving force in the derivation to determine its value when it is just equal to the horizontal components of the resisting forces)

H = driving force

Sum Vertical Forces; $o = V - N \cos \omega + \tau A \sin \omega$ Where $\tau = C + \frac{N}{A} \tan \phi$ then $o = V - N \cos \omega + cA \sin \omega + N \sin \omega \tan \phi$

 $o = V - N(\cos \omega - \sin \omega \tan \beta) + CA \sin \omega$

$$N = \frac{V + sA \sin\omega}{\cos\omega - \sin\omega \tan \phi}$$
(B-1)

Sum Horizontal Forces :

$$o = R - N \sin \omega - \tau A \cos \omega$$

$$o = R - N \sin \omega - CA \cos \omega - N \cos \omega \tan \phi$$

$$= R - N(\sin \omega + \cos \omega \tan \phi) - CA \cos \omega$$

$$N = \frac{R - CA \cos \omega}{\sin \omega + \cos \omega \tan \phi}$$
(B-2)

Equate N in equations B-1 and B-2:

$$\frac{V + cA \sin \omega}{\cos \omega - \sin \omega \tan \phi} = \frac{R - cA \cos \omega}{\sin \omega + \cos \omega \tan \phi}$$

$$R - cA \cos \omega = \left(V + cA \sin \omega\right) \left(\frac{\sin \omega + \cos \omega \tan \phi}{\cos \omega - \sin \omega \tan \phi}\right) \quad (B-3)$$
but $\frac{\sin \omega + \cos \omega \tan \phi}{\cos \omega - \sin \omega \tan \phi} = \frac{\frac{\sin \omega + \cos \omega \tan \phi}{\cos \omega - \sin \omega \tan \phi}}{\frac{\cos \omega}{\cos \omega} - \frac{\sin \omega}{\cos \omega} \tan \phi}$

$$\frac{\tan \omega + \tan \phi}{(1 - \tan \omega \tan \phi)}$$

ty identity

$$\frac{\tan\omega + \tan \phi}{1 - \tan \omega \tan \phi} = \tan (\phi + \omega)$$

Substituting into equation B-3 we have

$$R = cA \cos\omega + V \tan(\phi + \omega) + cA \sin\omega \tan(\phi + \omega)$$

= $V(\tan(\phi + \omega)) + cA(\sin\omega \tan(\phi + \omega)) + cA \cos\omega$
= $V(\tan(\phi + \omega)) + cA(\sin\omega \tan(\phi + \omega)) + \cos\omega$ (B-4)

by identities

$$\tan (\phi + \omega) = \frac{\tan \phi + \tan \omega}{1 - \tan \phi \tan \omega}$$

since $\tan \phi = \frac{\sin \phi}{\cos \phi}$ and similar for ω :

$$\therefore \tan (\phi + \omega) = \frac{\frac{\sin \phi + \sin \omega}{\cos \phi + \cos \omega}}{1 - \frac{\sin \phi \sin \omega}{\cos \phi \cos \omega}}$$

B-3

Get common denominator:

_	cosW	sinò	+	sinω	COSÓ
-		cos¢	cc	sω	
	cosò	cosw	-	sinò	sinw
		cos¢	c	ງຣຟ	

 $\frac{\cos\omega\sin\phi + \sin\omega\cos\phi}{\cos\phi\cos\omega - \sin\phi\sin\omega}$

divide by ccs ø we have :

<u>cosωsin ø</u>	+ sinch cos ø
cos ø	cos ø
cos ø cosw	sin ø sinw
cos ø	cos ø
	$\frac{\cos\omega\sin\phi}{\cos\phi}$

 $= \frac{\sin\omega + \cos\omega \tan \phi}{\cos\omega - \sin\omega \tan \phi}$

$$\operatorname{rultiply by sin} \omega: = \frac{\sin \frac{\omega}{\omega} + \cos \omega \sin \omega \tan \phi}{\cos \omega - \sin \omega \tan \phi};$$

$$\operatorname{add cos} \omega: = \frac{\sin \frac{\omega}{\omega} + \cos \omega \sin \omega \tan \phi}{\cos \omega - \sin \omega \tan \phi} + \cos \omega$$

$$= \frac{\sin \frac{\omega}{\omega} + \cos \omega \sin \omega \tan \phi}{\cos \omega - \sin \omega \tan \phi} + \cos^{2} \omega - \frac{\sin \omega \cos \omega \tan \phi}{\cos \omega - \sin \omega \tan \phi}$$

$$= \frac{\sin^{2} \omega + \cos^{2} \omega}{\cos \omega - \sin \omega \tan \phi}$$

$$= \frac{\sin^{2} \omega + \cos^{2} \omega}{\cos \omega - \sin \omega \tan \phi}$$

$$= \frac{1}{\cos \omega - \sin \omega \tan \phi};$$

Substitute into equation B-4 and multiply the denominator of the last term by cos we have: cos w

$$R = V(\tan(\phi + \omega)) + CA = V(\tan(\phi + \omega)) + CA = V(\tan(\phi + \omega)) + CA = COS(\omega) = COS$$

The formula for downhill sliding is derived in a similar manner.

EXHIBIT C: EQUATIONS FOR EARTH PRESSURE UNDER A TRAPEZOIDAL BASE

CASE 1: Resultant Within Kern CASE 2: Resultant Outside Kern

C-1



$$f = \frac{R}{A_g} + \frac{R(c-6)R^2 \frac{1}{2}}{I_{cg}}$$

C-1.2.1 Stress 2+ reference End of base: (y=c-o)

$$f = \frac{R}{A_g} + \frac{R(c-b')c}{I_{cq}}$$

Note: if b'> c, then the second term goes negative and the My/I stress subtracts from the P/A stress.

$$f = \frac{R}{A_g} + \frac{R(c-b')(c-B)}{I_{eg}}$$
The second term is partially

$$f = \frac{R}{A_g} + \frac{R(c-b')(c-B)}{I_{eg}}$$
The second term changes
Nore: if b'>c, then the second term changes
sign and the two terms will wind up adding numerically.

- 2. If the stress at the narrow end is negative, then the resultant is outside the kern, toward the wide end (Case 2) and it must be calculated accordingly.
- b. If the stress at the wide end 1s negative, then the resultant is outside the kern, toward the narrow end (Case 3) and it must be calculated accordingly.
- (. If the strenges of both ends are positive, then the resultant is inside the kun and the case I expression is valid.



 $A_{g} = no change if w' and w'' are interchangel,$ $I_{cg} = no change if w' and w'' are interchanged,$ $<math display="block">C = \frac{B(Zw'+w'')}{3(w'+w'')} as shown Bbove.$ $f = \frac{R}{A_{g}} + \frac{R((-b')(\pm y))}{I_{cg}}$ $f_{reference and} = \frac{R}{A_{g}} + \frac{R((-b')(c))}{I_{cg}}$

 $E \cap$ the expression for froterence and only the sub-expression + (c-b')(c) is different between the two reference locations: $(c-b') \subset On page C-4$ is checked against (c'-b'')c' on page: C-1: From page C-4, C and b' are seen to be equal to C' and b'', respectively. Therefore the two equations are equivalent.
CASE Z RESULTANT QUITSIDE THE KERN C-2.0 When the resultant is outside the kern, a portion of the base is no longer in compression. The remainer of the base which is in compression and applying a stress to the subsarface is called the "EFFECTIVE BAS: AREA", Shown (111) below:





C-Z.1 DETELMINING THE EFFECTIVE BASE WIDTH

To find the width of effective base area, the performs an incremental search across the actual base width to locate the zero-pressure point. This is done by holding the location of Resultant on the base constrant and varying the base deminsion so that f=0 in the equation:

$$f = \frac{P_A}{T} \pm \frac{Mc}{T}$$

FOR A TRAPEZOIDAL BASE

 $-f = \frac{R}{A_g} + \frac{R(C - b')(C - B)}{T_{L_f}}$

$$A_{g} = \frac{\omega' + \omega''}{2} B$$

$$T_{cg} = \frac{B^{3} (\omega'^{2} + 4\omega'\omega'' + \omega''^{2})}{36 (\omega' + \omega'')}$$

$$C = \frac{B(2\omega''+\omega')}{3(\omega''+\omega')}$$

B = BW - PELTAB

DELTAB = Incremental Distance from end of zero compression W' & W' are dependent on B

C-22 DETERMINING THE MAXIMUM STRESS

The maximum stress occurs opposite to the end of zero compression, which yields the pressure distribution diagram below



Applying the law of simple mechancis, Resultant force is equal to the volume of the pressure distribution diagram. This diagram can be divided in to three geometrical shape; one Prism and Two Pyramids:



Another retroit to find the volume of the diagram would be to integrate the distrubution over the area as show below:



The two method yield the same equation -

;

Where for the maximum stress is!

$$f = \frac{6R}{B(\omega' + 2\omega'')}$$

EXHIBIT D: COMPUTATION OF AT REST EARTH PRESSURE

D-1 PRESSURE CALCULATION

D-1.1 Direct Determination of Expected Pressure - Two of the more common methods available for the direct determination of horizontal at-rest earth pressure are described below:

- a. The finite element method can be used to predict atrest earth pressures. It accounts for multiple layer soil systems and for irregular backfill and surcharges. However, it is beyond the scope of this program.
- b. Boussinesq-type elasticity equations can be used to calculate prossures. Manipulation is required to account for sloping backfill, and is beyond the scope of this program.

D-1.2 Use of Horizontal Earth Pressure Coefficients - There are several common methods available for determining coefficients (K_r) to be multiplied by the earth weight to obtain horizontal pressure due to a level backfill. Several of them are discussed below:

- a. Field or laboratory tests by soils engineers can yield a suitable value for K_r. WES Technical Report S-75-16 discusses various methods and their accuracy.
- b. Approximate coefficients for typical soil types are shown in paragraph 3.d on page 3 of EM 1110-2-2502. These coefficients are valid for only level backfill. The procedure described in the referenced EM paragraph can be used to get a coefficient for a sloping backfill, but the method becomes invalid when the backfill slope angle exceeds the equivalent ϕ angle (ϕ ') calculated from equation 6 on page 3 of the EM.
 - (1) For an average sand (K = 0.5), the maximum slope is 1V:2.82H (19.47122^o) for this method.
 - (2) For an average clay (K = 0.85), the maximum slope is 1V:12.29H (4.6507^o)^r for this method.
- c. Jaky's equation

$K_r = 1 - \sin \phi$

can be used for cohesionless soils with level backfill. See WES TR S-75-16 for a discussion and comparison with measured values. It has been suggested that Jaky's equation can be modified to approximately account for sloping backfills, based on the similarity between the numerators of the Rankine active earth pressure coefficient for level and sloping backfill conditions. This can be demonstrated as shown below:

D-1

$$K_{a} = \frac{1-\sin\phi}{1+\sin\phi} \quad \text{Rankine's equation for level backfill}$$

$$K_{r} = \frac{1-\sin\phi}{1} \quad \text{Jaky's equation for level backfill}$$

$$K_{a} = \cos_{\beta} \quad \left[\frac{\cos\beta - \sqrt{\cos^{2}\beta - \cos^{2}\phi}}{\cos\beta + \sqrt{\cos^{2}\beta - \cos^{2}\phi}} \right] \quad \text{Rankine for sloping fill } \star$$

$$K_{r} = \cos_{\beta} \quad \left[\cos\beta - \sqrt{\cos^{2}\beta - \cos^{2}\phi} \right] \quad \text{Modified Jaky for sloping fill} \star$$

$$\text{Where } \phi = \text{angle of internal friction}$$

 β = angle of backfill slope

D-2 GRAPHIC COMPARISON OF METHODS - Figure D-1 shows a graphic comparison of several methods for calculating K_r, with ϕ varying from 0° to 90° and β varying from 0° to 80°.

* This is the total earth pressure coefficient, not its horizontal component.



D-3

EXHIBIT E: FORMULAS FOR WORKING STRESS DESIGN OF CANTILEVER SLABS WITH AXIAL FORCE (ACCORDING TO APPENDIX B OF ACI 318-77)

E-1 NOTATION is as shown in ACI 318-71.

E-2. DEFAULT VALUES

item	for hydraulic structures	for other structures		
f _c /f'c	0.35	0.45		
f max.	20000 psi	20000 psi for grades 40 or 50 24000 psi for grade 60 or higher		
$n = E_s/E_c$	$\frac{29000000}{57000 f'_{c}} = \frac{508.77}{f'_{c}}$	$\frac{508.77}{f'_c}$		

E-3. FLEXURE

E-3.1 BASIC RELATIONSHIPS (Reference ACI Publication SP-3, "Reinforced Concrete Design Handbook for Working Stress Method") for moment and axial



a. For the concrete stress f to be at its maximum allowable value, c

$$k = 1 \qquad \text{and } j = 1 - \frac{k}{3}$$

$$1 + f_{c}$$

$$\frac{s}{nf_{c}}$$

b. The equivalent external (applied) moment is then Ne, which is found from

Ne =
$$N(\frac{M}{N} + d'') = M + Nd'',$$

where N is plus for axial compression. Manipulating this relationship, and the ones in the previous paragraph, to get N out of the denominator, yields

Ne = Nd" + M = N(d-
$$\frac{h}{2}$$
) + M = N(d - $\frac{d+cover}{2}$) + M = $\frac{N}{2}$ (d-cover) + M

E-3.2.1 Determining minimum slab thickness Summing moments about T yields Ne = Cjd = $f_c bkjd^2 \frac{1}{2}$ $f_c bkjd^2 \frac{1}{2} = \frac{Or}{2}d + M - \frac{N(cover)}{2}$ or

$$(f_bkj)d^2 - Nd - (2M-N(cover)) = 0.0$$
 (E-1)

Solving the quadratic equation E-1 for the smaller real, positive root will yield the minimum allowable effective depth d. Adding the cover to the center of the reinforcing steel yields the minimum slab thickness for moment at that location. Note that the effect of axial tension is to reduce the compressive stress and so yield a smaller value for minimum slab thickness. Such reduction must be applied with caution.

E-3.2.2 Determining reinforcement for balanced design in Sections with minimum effective depth, as in Equation E-1:

Taking moments about the concrete force C, using the relationships in paragraph E-3.1a and E-3.1b, derive an expression for A_s to have the steel stressed to its allowable stress simultaneously with the extreme fiber concrete stress reaching its allowable:

N (e-jd) = Tjd
Ne - Njd = Tjd
$$\frac{N}{2}$$
(d-cover) + M - Njd = Tjd

$$M + N \left[d\left(\frac{1}{2} - j\right) - \frac{cover}{2} \right] = Tjd = A_s f_s jd$$

$$A_s = \frac{M}{f_s jd} + \frac{N}{f_s jd} \left[d\left(\frac{1}{2} - j\right) - \frac{cover}{2} \right]$$
(E-2)

ķ₫

↓^εs

0

(1-k)d

E-3.2.3 Determining reinforcement for Sections with more than the minimum effective depth (under-reinforced) from Equation E-1:

- Given: M, N, d, f_s, h, cover, d" Find = A_s
- a. Basic relationships:
 - $f_s = E_s \epsilon_s$, by definition
 - $\varepsilon_{c} = \varepsilon_{s} \frac{k}{1-k}, \text{ from similar triangles}$ $\varepsilon_{c} = \varepsilon_{c} \frac{k}{1-k}, \text{ from similar triangles}$ $f_{c} = E_{c}\varepsilon_{c} = E_{c}\varepsilon_{s} \frac{k}{1-k} = E_{s}\varepsilon_{s} \frac{E_{c}}{E_{s}} \frac{k}{1-k} = \frac{f_{s}}{n} \frac{k}{1-k}$
- b. Summation of axial forces = 0:

$$C-T = N$$

$$C = \frac{f_c bkd}{2} = \frac{f_s bd}{2n} \frac{k^2}{1-k}, \quad T = A_s f_s$$

c. Summation of moments about t = 0:

$$Cjd - Nd'' - M = 0$$

$$j (1 - \frac{k}{3}) = \frac{3 - k}{3}$$

$$\frac{f_s bd}{2n} \frac{k}{1 - k} (\frac{3 - k}{3}) d - Nd'' - M = 0$$

$$\frac{f_s bd^2}{6n} (\frac{3k^2 - k^3}{1 - k}) - Nd'' - M = 0$$

$$\frac{f_s bd^2}{6n} k^3 - \frac{f_s bd^2}{2n} k^2 - (M + Nd'')k + (M + Nd'') =$$

Solving this cubic equation for the smallest real, positive root will yield the actual value of k. Then, from the force summation $C-T \approx N$, get

$$\frac{f_{s}bd}{2n} \frac{k^{2}}{1-k} - \frac{f_{s}A_{s}}{s} = N$$

$$A_{s} = \frac{bd}{2n} \frac{k^{2}}{1-k} - \frac{N}{f_{s}}$$
(E-3)

E-3.3 ANALYSIS (by rearrangement of Equation E-1)

E-3.3.1 Concrete stress:

$$f_{c}bkjd^{2} - Nd - (2M - N(cover)) = 0$$

$$f_{c}bkjd^{2} = N(d-cover) + 2M$$

$$f_{c} = N \frac{d-cover}{bkjd^{2}} + M \frac{2}{bkjd^{2}}$$
(E-4)

E-3.3.2 Steel Stress:

1

Substituting the basic relationships of T for those of C in equation E-1:

$$A_{s}f_{s}jd = M + N \left(d \left(\frac{1}{2} - j\right) - \frac{cover}{2}\right)$$

$$f_{s} = \frac{M}{A_{s}jd} + \frac{N}{A_{s}jd} \left[d \left(\frac{1}{2} - j\right) - \frac{cover}{2}\right]$$
(E-5)

E-4. SHEAR (See ACI 318-71, paragraph 8.10.3)

Allowable stresses will be 55% of the code allowable for ultimate strength design given in ACI 318-71, Chapter 11. In section 11.4.4 of ACI 318-71, the term N_u will be used as 2.0 times the axial force N from the key.

a. Shear stress is calculated using equation 11-3 from ACI 318-71:

which for working stress design rules, becomes

$$v = \frac{V}{bd}$$

b. The <u>allowable stress</u> is calculated according to equation 11-8 in ACI 318-71 for sections with axial load (tension is used as negative):

$$\mathbf{v}_{c} = \frac{2\left(1 + \frac{N_{u}}{500 \text{ A}_{g}}\right)\sqrt{f'c}}{2}$$

which is ACI 318-71 paragraph 11.4.1 when $N_u = 0$. Equation 11-4 in ACI 318-71 is used for sections without axial load:

 $v_c = 1.9 \sqrt{f'_c} + 2500 \frac{A_s}{b_w d} \frac{V_u d}{M_u}$, but not over 3.5 f'_c , using V for V_u

and M for M since they would have the same load factor. (In equation 11-4, (V_ud) shall not exceed M_u).

Both values of v_{c} are multiplied by 0.55 before comparison with v.

E-5. DEVELOPMENT OF REINFORCEMENT

Development of reinforcement (bond strength) shall be as required in ACI 318-71 Chapter 12, except that computed shear V shall be multiplied by 2.0 and substituted for V_u . In computing M_t , the quantity (d-a/2) may be taken as 0.85d. Where the A_s provided is more than twice that required, the stress may be considered as always less than 0.5f_y for the purpose of satisfying provisions relating to splices. Splices will not be calculated in this program, but the information is presented in this appendix for completeness. This presentation assumes no hooked ends except in keys.

Reinforcement in T-Wall components is assumed to be negative moment reinforcement as considered in ACI 318-71 paragraph 12-3 on page 43 of the code. Stirrups will not be used in this program.

Embedment length into the span shall be as required by ACI 318-71 paragraphs 12.1.1 (hooks shall be used if necessary) and 12.1.4 (reinforcement shall extend beyond the point at which it is no longer required to resist flexure for a distance equal to the effective depth of the member (d) or 12 bar diameters, except at ... the free end of cantilevers.

Flexural reinforcement shall not be terminated in a tension zone unless one of the following conditions is satisfied:

- 1. The shear at the cutoff point does not exceed two-thirds that permitted ...
- 2. For #11 and smaller bars, the continuing bars provide double the area required for flexure at the cutoff point and the shear does not exceed three-fourths that permitted.

This may be applicable to the heel slab when there is a key under the heel: At least one-third of the total reinforcement provided for negative moment at the support shall have an embedment length beyond the point of inflection not less than the effective depth d, 12 times the bar diameter, or onesixteenth the clear span (one-eighth of the cantilever span), whichever is greater.

Development length in inches shall be the product of the basic length for #11 or smaller bars,

$$\frac{0.04A_{b}f_{y}}{\sqrt{f'c}}$$

times one or more of the appropriate factor5listed below:

- 1. Top bars (more than 12" or concrete below) = 1.4 .
- 2. F over 60000 psi = not permitted.
- 3. Bars modified by 1. or 2. above, and spaced laterally at least 6 inches on center = 0.8 (1.4 X 0.8 = 1.12).
- 4. Bars modified by 1. or 2. above and in excess of that required A (reg'd)

 $=\frac{s}{A_{g} \text{ (used)}}$

The development length shall not be less than twelve inches, but not less than 0.0004 $d_h^{}$ f $_v^{}.$

EXHIBIT F: FORMULAS FOR ULTIMATE STRENGTH DESIGN OF NON-HYDRAULIC STRUCTURES WITH CANTILEVER SLABS WITH AXIAL FORCE (ACCORDING TO ACI STANDARD NO. 318-77)

F-1. NOTATION is as shown in Appendix B to ACI 318-71.

F-2. DEFAULT VALUES

	Item	Default Value
a.	E _s	2900000
ь.	Maximum actual f _s from normal long-term loading	not applicable
c.	fy	40000 psi
d.	Capacity reduction factors: flexure shear, bond	0.9 0.85
e.	Steel area limiting reinforcement ratio: $\frac{A_s \text{ allowable}}{A_s \text{ balanced}} = R$	(0.5 if load case includes earthquake)
£.	Maximum ratio $\mathbf{\hat{p}} = \frac{\mathbf{f} \mathbf{y} \mathbf{A} \mathbf{s}}{\mathbf{f}_{c}^{t} \mathbf{b} \mathbf{d}}$	not applicable when $h \ge \frac{10}{10}$ (ACI318-71
g,	Minimum ratio <u>f_vAs</u> ** bd	200.0
h.	Maximum strain ^e	0.003
i.	Stress rectangle magnitude	0.85f ['] _c
j.	β ₁ in ACI 318-71 par. 10.2.7	0.85
k.	Load factors (1) flexure shear on concrete bond on top bars bond on other bars	not applicable "' "'
	 (2) Using D=concrete weight + water weight V = applied forces vertical + earth weight W = wind E = earthquake, horizontal or vertical H = earth horizontal + surcharge horizontal F = seepage horizontal + uplift 	 (1) 1.4(D+F)+1.7(V+H) except: a. If sign of D is opposite to H or F, use 0.9 for D. b. If sign of V is opposite to H or F, omit V.
Not	e that V and H are the orthogonal components the ACI Code load factor group called L.	 (2) With wind or earthquake: a. Add(1.7W+1.87E) and use additional overall factor of 0.75.
*fo AC *AC	or deflection control, based on 21 318-63, par. 1507 21 318-71, par. 10.5.1	 b. If D+W+E only:0.9D+ 1.3W+1.43E. c. Use bigger of a or b above.

rev Sep 78

F-1

F-3. FLEXURE

References:

1. Commentary on ACI Code 318-71, Chapter 10.

- 2. PCA Notes on ACI 318-71, Lecture 4.
- 3. ACI Publication SP-17(73), Commentary on Flexure.

F-3.1

BASIC RELATIONSHIPS for Moment and Axial Force.



With N as shown (for tension use -N): The external equivalent moment is $M_{us} = M_u + N_u \frac{(d-h)}{2}$ (F-1)

From the strain diagram:

$$c_{b} = d \left[\frac{\varepsilon_{c}}{\varepsilon_{c} + \frac{f_{y}}{E_{s}}} \right] = 0.685d \text{ for } f_{y} = 40,000, \ \varepsilon_{c} = 0.003, \ \varepsilon_{s} = 29000000 \quad (F-2)$$

From the stress diagram:

$$j_{u}d = d - \frac{\beta_{1}c_{b}}{2}$$
 (F-3)

F-2

The internal moment is

$$\frac{M_{ue}}{\phi} = (j_u d) \begin{bmatrix} b & a & (0.85f'_c) \end{bmatrix} \text{ for concrete}$$
(F-4)
Where $a = \beta_1 c_u$ for balanced design, and

$$\frac{M}{\phi} = j_u d \begin{bmatrix} A \\ S \\ balanced \end{bmatrix} for steel (excluding N) (F-5)$$

F-3.2 DESIGN

F-3.2.1 <u>Concrete thickness for strength</u>, - The expression for minimum effective depth "d" is derived by substituting equations F-1 through F-3 into equation F-4 and solving for d, as shown in Addendum A to Exhibit F.

 $d = f(\phi, b, \beta_1, f'_c, \epsilon_c, E_s, M_u, N_u, d'')$ (F-6)

To account for the R factor defined in paragraph F-2e, an adjustment must be made in the application of Equation F-6. This adjustment is described in the following discussion. The effective depth "d" and the reinforcing steel area " A_s " must be sufficient to resist the applied moment and axial force. However, the ACI code limits the allowable steel area to a fraction "R" of the steel area for a balanced design. These two conditions can be satisfied either by determining the effective depth for an increased moment equal to M_{us}/R or by going into equation F-6 with an "effective width" equal to b times R. F-3.2.2 Steel reinforcement

a. The area of steel required for balanced design, with the R factor added to get the maximum reinforcement allowed is

$$A_{s_{max}} = \underbrace{\frac{M_{us}R}{d - \frac{\beta_1 c_b}{2}}}_{f_y \phi} - \frac{N_u}{\phi f_y}, \qquad (F-7)$$

which will be strong enough because R was included when equation F-6 was used.

b. Since the actual value of d will, in general, be greater than the theoretical value, the full A is excessive. The actual need is based on (1) calculating a from the max concrete strength based on actual values of M_{us}, N_u, and d, and then (2) using this actual value of a instead of $\beta_1 c_b$ in equation F-7:

(1) Get a:

 $M_{u} + N_{u} (d-\underline{h}) = M_{us} = C(j_{u}d)\phi ;$ substituting C = (0.85 f'_c) a b and $(j_{u}d) = d - \frac{a}{2}$

to get $\begin{bmatrix}
\frac{1}{2} \cdot 85f'_{c} \cdot b\phi}{2} = 0,$ using the absolute value of M_u and N_u as + for compression. Then solve

for a as the smaller real, positive root of the quadratic equation.

(2) Get
$$A_s$$
:

$$A_s = \frac{M_u + N_u (d - \frac{h}{2})}{\phi f_y [d - \frac{a}{2}]} - \frac{N_u}{\phi f_y}$$
(F-8)

This must not be greater than the value from equation F-7 or the value of the expression in paragraph F-2f. It must not be less than the value of the expression in paragraph F-2g.

F-3.3 ANALYSIS

wí

l

F-3.3.1 The concrete resisting moment in flexure can be calculated using equation F-4 with equations F-2 and F-3, using $\beta_1 = c_1$ for a:

$$M_{us (available)} = (d - \frac{\beta_1 c_b}{2}) \begin{bmatrix} 0.85 f'_c b \beta_1 c_b \end{bmatrix}^{1/2}, \quad (F-9)$$

th c_b calculated using equation $F-2$

F=3,3,2 The steel resisting moment, which must be less than the concrete resisting moment by at least the ratio R from paragraph F-2e, can be calculated by substituting equation F-1 into equation F-8, using the relationship

$$M_{us (available)} = (j_u d) (\phi f_y A_s + N_u)$$
(F-10)

F-4 SHEAR

a. The <u>actual</u> shear stress can be calculated with equation 11-3 from ACI 318-71:

$$v_u = \frac{V_u}{\phi b_v d}$$

b. The <u>allowable</u> shear stress is calculated from equation 11-8 in ACI 318-71 for sections with axial load (tension is entered with a negative sign):

$$v_c = 2 (1 + \frac{N_u}{500A_g}) \sqrt{f_c'}$$

which is ACI 318-71 paragraph 11.4.1 when $N_{tr}=0$. Equation 11-4 in ACI 318-71 is used for sections without axial load:

$$v_c = 1.9 \sqrt{f_c'} + 2500 \frac{A_s}{b_w d} \frac{V_u d}{M_u}$$
, but not over $3.5 \sqrt{f_c'}$.

(In this equation, $(V_u d)$ shall not exceed M_u .)

F-4

F-5 DEVELOPMENT OF REINFORCEMENT

Development of reinforcement is the same as for Working Stress Design (Exhibit E, par. E-5), except that the computed shear is $V_{\rm u}$ and is not multiplied by 2.0.

Addendum A: Derivation of Equation for d

Given
$$M_{us} = (j_u d) \left[(b \beta_l c_b) (0.85 f'_c) \right] \phi$$
 (F-4)

with
$$C_b = \frac{d\varepsilon_c}{\varepsilon_c + f_y} = \frac{d\varepsilon_c}{\varepsilon_c E_s + f_y} = \frac{d\varepsilon_c E_s}{\varepsilon_c E_s + f_y}$$
 (F-2)

and $j_u d = \frac{d}{2} - \frac{\beta_1 c_b}{2}$, derive equation F-6:

1. Substitute equation F-3 into equation F-4 and simplify:

$$M_{us} = {\begin{pmatrix} d \\ - \frac{\beta_{1}c_{b}}{2} \end{pmatrix} b\beta_{1}c_{b} & 0.85f'_{c}\phi} \\ = \frac{d\phi b\beta_{1}c_{b} & 0.85f'_{c}}{1} - \frac{\beta_{1}^{2}c_{b}^{2}b & 0.85f'_{c}\phi}{2} \\ 2M_{us} = \phi 2db\beta_{1}c_{b} & 0.85f'_{c} - \beta_{1}^{2}c_{b}^{2} & b & 0.85f'_{c}\phi \end{cases}$$

2. Substitute equation F-2 into the equation above:

$$2M_{us} = \phi^{2}db\beta_{1} \quad 0.85 \quad f'c \quad \left[\frac{d\epsilon_{c}E_{s}}{\epsilon_{c}E_{s}+f_{y}}\right] - \phi b\beta_{1}^{2} \quad 0.85f'c \quad \left[\frac{d\epsilon_{c}E_{s}}{\epsilon_{c}E_{s}+f_{y}}\right]^{2}$$

$$2M_{us} = \frac{2\phi b\beta_{1} \quad 0.85f'c \quad \epsilon_{c}E_{s}}{\epsilon_{c}E_{s}+f_{y}} \quad - \frac{\phi b\beta_{1}^{2} \quad 0.85f'c \quad \epsilon_{c}^{2}E_{s}^{2}}{(\epsilon_{c}E_{s}+f_{y})^{2}} \quad d^{2}$$

$$2M_{us} = \frac{\phi b\beta_{1} \quad 0.85f'c \quad \epsilon_{c}E_{s}}{(\epsilon_{c}E_{s}+f_{y})^{2}} \quad - \frac{\phi b\beta_{1}^{2} \quad 0.85f'c \quad \epsilon_{c}^{2}E_{s}^{2}}{(\epsilon_{c}E_{s}+f_{y})^{2}} \quad d^{2}$$

$$d^{2} = \frac{2M_{us} \quad (\epsilon_{c}E_{s}+f_{y})^{2}}{\phi b\beta_{1} \quad 0.85f'c \quad \epsilon_{c}E_{s} \quad \left[2(\epsilon_{c}E_{s}+f_{y}) - \beta_{1}\epsilon_{c}E_{s}\right]}{(\epsilon_{c}E_{s}+f_{y})^{2}} \quad d^{2}$$
3. Substitute equation F-1 for M_{us} with $h = d+d''$:

$$d^{2} = 2\left[w + w \quad \left(d - d+d''\right)\right] = w + 2$$

$$d^{2} = \frac{2\left[M_{u} + N_{u}\left(d - \frac{d+d''}{2}\right)\right]\left(\varepsilon_{c} + f_{y}\right)^{2}}{\phi b\beta_{1} 0.85f'_{c} \varepsilon_{c} \varepsilon_{s} \left[2\left(\varepsilon_{c} + f_{y}\right) - \beta_{1} \varepsilon_{c} \varepsilon_{s}\right]}$$

F-6

$$1 = \underbrace{\begin{bmatrix} M_{u} + \frac{1}{2}N_{u}d - \frac{1}{2}N_{u}d'' & (\varepsilon_{c} E_{s} + f_{y})^{2} & 2 \\ \phi b \beta_{1} 0.85f'_{c} \varepsilon_{c} E_{s} \begin{bmatrix} 2(\varepsilon_{c} E_{s} + f_{y}) - \beta_{1} \varepsilon_{c} E_{s} \end{bmatrix} d^{2} \\ 1 = \underbrace{\begin{bmatrix} M_{u} - \frac{1}{2}N_{u}d'' & (\varepsilon_{c} E_{s} + f_{y})^{2} & 2 \\ \phi b \beta_{1} & 0.85f'_{c} & \varepsilon_{c} E_{s} \begin{bmatrix} 2(\varepsilon_{c} E_{s} + f_{y}) - \beta_{1} \varepsilon_{c} E_{s} \end{bmatrix} d^{2} \\ + \frac{N_{u} (\varepsilon_{c} E_{s} + f_{y})^{2}}{\phi b \beta_{1} (0.85)f'_{c} & \varepsilon_{c} E_{s} \begin{bmatrix} 2(\varepsilon_{c} E_{s} + f_{y}) - \beta_{1} \varepsilon_{c} E_{s} \end{bmatrix} d} \\ 4. \text{ Substituting DDEN} = \frac{\phi b \beta_{1} (0.85)f'_{c} \varepsilon_{c} \varepsilon_{s} \begin{bmatrix} 2(\varepsilon_{c} E_{s} + f_{y}) - \beta_{1} \varepsilon_{c} E_{s} \end{bmatrix} d}{(\varepsilon_{c} E_{s} + f_{y})^{2}} \\ 0 = \underbrace{\begin{bmatrix} M_{u} - \frac{1}{2}N_{u}d'' \\ 0 \end{bmatrix} \frac{d}{2} + \frac{N_{u}}{DDEN} - \frac{1 * DDEN}{d} - 1 \\ 0 = \underbrace{\begin{bmatrix} M_{u} - \frac{1}{2}N_{u}d'' \\ 1 \end{bmatrix} \frac{2}{2} + \frac{N_{u}}{u} - \frac{1}{2} \\ 0 = \underbrace{\begin{bmatrix} M_{u} - \frac{1}{2}N_{u}d'' \\ 1 \end{bmatrix} \frac{2}{d} + \frac{N_{u}}{u} - \frac{1}{d} \end{bmatrix}$$

$$DDEN d^{2} - N_{u}d - \left[M_{u} - \frac{1}{2}N_{u}d'\right] = 0$$

6. Solve quadratic for roots of d.

7. Keep the smallest real, positive root as the value of d.

F-7

EXHIBIT G: COMPARISON OF BOUSSINESQ AND EOWLES SOLUTIONS FOR PRESSURES ON RETAINING WALLS DUE TO SURCHARGE LOADS

G-1 SOURCE - This comparison was made by Dr. W. P. Dawkins on 20 June 1977.

G-2 USE OF BOUSSINESQ THEORY-VERTICAL P TIMES K - The comparisons are based on the premise that horizontal pressures doe to surcharges could be obtained by multiplying the vertical pressure predicted by the Boussinesq equation by the coefficient of active earth pressure. The Boussinesq solutions are given by Eqns 67, page 87, for a line load, and Eqns (c), page 363, for a point load, in <u>Theory of Elasticity</u> by Timoshenko and Goodier.

G-3 USE OF BOWLES' EQUATIONS - The quations for the Bowles solution are given in Fig. 11-20(a), page 356, for a point load, and in Fig. 11-21 page 357, for a line load, in Foundation Analysis and Design, second edition, by Bowles.

G-4 PRESSURES - Pressure distributions for the two solutions are shown in Figs. G-1 and G-2. The Boussinesq solutions are plotted for $K_a = 1.0$. Pressures for the point load cases are those which occur at points on the wall nearest to the point load.

G-5 RESULTANT FORCES and their points of action on the wall are shown in Fig. G-3 for each of the pressure distributions. For the point loads, pressures are assumed to be constant over a unit length of the wall. G-6 COMPARISON - For both line and point load cases, the Bowles equations give larger resultant forces, with the point of action of the resultant being higher on the wall than the Boussinesq solutions for loads at a distance greater than two tenths of the wall height.







Fig. G-1b Line Load



Fig. G-2a Point Load



Fig. G-2b - Point Load



 $R = \int_{H} g_{h} dH$ Line logd: $\overline{R} = g_{h} Ig$ Point logd: $\overline{R} = g_{h} H IP$ $\overline{Y} = Y_{R} / H$

	Line Load				Point Load				
	Boussiness +		Bowles		Boussinesgt		Bowles		
m	R	$\overline{\nabla}$	Ŕ	$\overline{\mathbf{Y}}$	ন	$\overline{\mathbf{Y}}$	R	 	
0.1	1,156	0.568			-	-			
0.2	0.731	0.471	¥	*	1.128	0.570	*	4 44	
0,3	0.502	0.409	-		0.616	0.482	-	ŕ	
0.4	0.356	0.366	0.547	0.608	0.372	0.421	0.788	o.587	
0.6	0.189	0.310	0.468	0.519	0.157	0.344	0.457	0.476	
0.8	0.105	0.276	0.388	0.464	0.073	0.299	0.277	0.407	

* For ka= 1.0

* Bowles solution constant for m < 0.4

Fig. G-3 Resultants







.....

--

WES FORM NO. REVICET 19F8 1563

H-2

COMPUTED BY. DATE FILE NO. SUBJECT EXHLBIT H SAMPLE RETAINING WALL SHEAR FRICTION SLIDING ANALYSE CHECKED BY SHEET NO. 3 DATE CALCULATE HYDROSTATIC UPLIFT: Net Head (AH) = 15.5 - 3 - 3 = 9.5 FT Creep path (Lc) = $9.2 \pm 1.5 \pm \sqrt{14.8^2 \pm 5.7^2}$ +1.5 + 3.0 = 9.2 + 1.5 + 15.86 + 1.5 + 3.0= 31.06 FT - = 0.306 = hydraulic gradient 9.5 ΔH 31.06 0.3674 KSF 'u, - P. Uz 0.65% KSF LINE of creep method for hydrostatic pressures assumes uniform loss of potential head along the creep path (from ground surface down to and **R**Í 144 20283 KSF along failure plane and back t ground surface) USING option (2) failure plane and back to of paragraph 3.3.3 K 1.5 14.8 Hydrostatic pressures: PO base of Hia = 0.0625 (10.5) = 0.65625 Ksf P1 = 0.65625 + 0.0625 (9.2) - 0.306 (9.2) (0.0625) = 1.0553 kst P2 = 1.0553 - 1.5 (0.306) (0.0625) = 1.027 KSt P3 = 1.027 - (0.0625)(5.7) - (0.306)(15.86)(0.0625)=0.3694 K PO ground over toe = 0.3614 - 4.5(0.0625) - 0.306(4.5)(0.0625) 20

WES FORM NO. 1253

H**-**3

SAMPLE TETATIONG	UBJECT EXHIBLT H	COMPUTED BY	DATE	FILE NO.
UPLIFT FORCES: $U_1 = 0.3614 (16.3) = 5.9886 k/ft.$ $U_2 = (1.021 - 0.3614)(14.8)(\frac{1}{2})((0521.06) = 4.355 k/ft)$ $U_3 = (1.021 - 0.3614)(1.5) = 0.9894 k/ft$ $U_4 = (1.0553 - 1.021)(1.5)(\frac{1}{2}) = 0.0212 k/ft$ $U_4 = (1.0553 - 1.021)(1.5)(\frac{1}{2}) = 0.0212 k/ft$	SAMPLE RETAINING WALL HEAR FRICTION SLIDING ANALYS	S CHECKED BY	DATE	SHEET NO. 4
UPLIFT FORCES: $U_1 = 0.3614 (16.3) = 5.9886 k/ft.$ $U_2 = (1.021 - 0.3614)(14.8)(\frac{1}{2})((0521.06) = 4.355 k/ft)$ $U_3 = (1.021 - 0.3614)(1.5) = 0.9894 k/ft$ $U_4 = (1.0553 - 1.021)(1.5)(\frac{1}{2}) = 0.0212 k/ft$		·····		
$U_{1} = 0.3694 (16.3) = 5.9886 k/ft.$ $U_{2} = (1.029 - 0.3694)(14.8)(\frac{1}{2})(10521.06) = 4.355 k/ft.$ $U_{3} = (1.029 - 0.3694)(1.5) = 0.9894 k/ft.$ $U_{4} = (1.0553 - 1.029)(1.5)(\frac{1}{2}) = 0.0212 k/ft.$	UPLIFT FORCES :			
$U_{1} = 0.3694 (16.3) = 5.9886 k/ft.$ $U_{2} = (1.029 - 0.3694)(14.8)(\frac{1}{2})((0.521.06) = 4.555 k/ft.$ $U_{3} = (1.021 - 0.3694)(1.5) = 0.9894 k/ft.$ $U_{4} = (1.0553 - 1.029)(1.5)(\frac{1}{2}) = 0.0212 k/ft.$				ſ
$ \begin{array}{l} U_{2} = (1.027 - 0.3674)(14.8)(\frac{1}{2})(10521.06) = 4.555 \frac{1}{7}t_{4} \\ U_{3} = (1.027 - 0.3674)(1.5) = 0.9894 \frac{1}{7}t_{4} \\ U_{4} = (1.0553 - 1.027)(1.5)(\frac{1}{2}) = 0.0212 \frac{1}{7}t_{7}t_{4} \end{array} $	$V_1 = 0.3674 (16.3) = 5.98$	86 K/ft.		
$V_{2} = (1.021 - 0.3614)(14.8)(2)(10521.06) = 4.555 744 V_{3} = (1.021 - 0.3614)(1.5) = 0.9894 k/ft V_{4} = (1.0553 - 1.021)(1.5)(1/2) = 0.0212 k/ft V_{4} = (1.0553 - 1.021)(1.5)(1/2) = 0.0212 k/ft $				
$U_{3} = (1.027 - 0.3674)(1.5) = 0.9894 \text{ k/ft}$ $U_{4} = (1.0553 - 1.027)(1.5)(1/2) = 0.0212 \text{ k/ft}$ (1.5542)	$V_2 = (1.021 - 0.3814)(14.8)$)(5)((0521.00)= 4,555.7	**> ZU =
Uq = (10553-1027)(15)(1/2) = 0.0212 K/ft	$V_3 = (1.027 - 0.3674)(1.5) =$	- 0.9894 K/ft		11.55424
	$l_{a} = (10553 - 1027)(15)(1)$	1) = 0,0212 K/.	f+	
				J

REV OCT 1968 1253

SUBJECT E	XHIBIT H		DATE	FIL	E NO.			
SAMPLE Shear Fr	RETAINING WALL		DATE	SHF	ET NO.5			
SYMBOL				FORCE				
311602	FACTORS			+	4			
C ₁	$\frac{1}{2}(1.5 + 2.0)(15.5)(0.15)$			4,07				
۲,	1.5 (16.3) (0.15)			3.67				
C 3	1 (0.7)(S.7)(0.15)			0.30				
C4	(1.5)(5.7)(0.15)			1.2 ° 9.32				
WE,	$\frac{1}{2}(2) \left(\begin{array}{c} 8.7 \\ 0.125 \end{array} \right) \left(\begin{array}{c} 0.125 \\ 0.25 \\ 15.5 \end{array} \right) \left(\begin{array}{c} 0.125 \\ 0.125 \end{array} \right)$		_	2.17				
WE 2	(3)(0.125)(5.6) $\frac{1}{2}(3)[(3)(\frac{0.25}{(5.5)})](0.125)$			2.10				
Ww atove E.	$\frac{1}{2} (10.5) \begin{bmatrix} (10.5) \begin{bmatrix} 0.0625 \end{bmatrix} (8.73) \\ \frac{1}{2} (10.5) \begin{bmatrix} (10.5) \begin{bmatrix} 0.25 \\ 15.5 \end{bmatrix} \end{bmatrix} (0.0625)$			5.93 0.06 5.99				
WN	$\frac{1}{2}(14.8)(5.7)(0.125)$			5.29				
		 		5.02				
($+ W_E + W_W + W_N$			24.41				
HIG	$0.656(10.5)(\frac{1}{2})$	3.444						
HIB	$(1.0553 - 0.656)(9.2)(\frac{1}{2})$	1.837						
H, c	0.656 (9.2)	6.035						
Hz	0.3674(4.5)(-2)		0.827					
U _H	1.027 + 0.3674 (5.7) 2		3.974					
HA	$\frac{0.0625(9.2)^2}{2} \left[\frac{1}{2} an^2 (45^2 - \frac{20^2}{2}) \right]$	1.297	o use	bechuse Ve pressi	neglection			
	$-2(0,7)(9,2)$ $\left[\frac{1}{2} \tan \left(45^{\circ} - \frac{20^{\circ}}{2} \right) \right]$	-9.019	exis	4				
Hiat Hi	$H_{16} + H_{1c} - H_2 - U_H + H_A$	6.515						

NES FORM NO. 1253

H-5

OMPOTEC H 1.1.40 EXIBIT H SAMPLE RETAINING WALL SHEAR FRICTION SLIDING ANALYSIS HET FEL BY SHEET NO 6 $H_{p_1} = 2 ch \sqrt{k_p} = 2 ch \sqrt{tan^2 (4s^* + \Phi/2)} = 2 (0.7) (4.5) \sqrt{tan^2 (4s^* + \Phi/2)}$ Hpc = 8.997 KIPS $H_{PØ} = \frac{1}{2} rh^{2} kp = \frac{1}{2} rh^{2} tan^{2} (45^{\circ} + 4/2) = \frac{1}{2} (0.0625) (4.5)^{2} tan^{2} (45^{\circ} + 292)$ Hpp = 1.291 KIPS $\Sigma H_p = \Sigma H_{Pc} + Z H_{Pd} = 8.991 + 1.291 = 10.288 \text{ kips}$ THE SHEAR- Friction safety factor against shiding as given in ETL 1110-2-184, is obtained by dividing the horizontal skiding resistance (ZR) obtained from resisting forces including those along the critical path (plane or combination of planes which others the least resistance to skiding, by the summation of horizontal service loads (ZH) to be applied to the structure. ZV = 24.41 - 11.5542 = 12.8558 K $\Sigma R = \Sigma V \cdot \tan(\phi + w) + \frac{C \cdot A}{\cos w [1 - \tan \phi \cdot \tan w]} + H_{p}$ $= 12.8558 \tan (20^\circ + 21.06^\circ) + \frac{(0.7)(15.86)}{(0.521.06^\circ) + - \tan 20^\circ \tan 21.06^\circ]}$ + 10.288 = 11.199 + 13.836 + 10.288= 35.323 KIPS ZH from sheet H-2 chear - friction safety factor against skiding dong the total surface on sheet H-2: $\frac{\Sigma R}{\Sigma H} = \frac{35.323}{6.515} = 5.42$ Fs =

NELENDMIN ... IDER



WES FORM NO. 1253

H-7

E	XHIBIT H		1 × 4 × 4	· · · · ·	• • • • • •		
SAMPLE Shear Fi	RETAINING WALL RICTION SLIDING ANALYSIS	·	·		8		
STHBOL	MBOL FACTORS		FORCE				
	4			+t	<u>ا</u>		
Ċ,	$\frac{1}{2}(1.5+2.0)(15.5)(0.15)$			4.09			
ζ,	1.5 (16.3) (0.15)			3.61			
Сз	$\frac{1}{2}(0.7)(5.7)(0.15)$			c.30			
C.	(1.5)(5.7)(0.15)			1.28			
	(2)(8.7)(0.125)		+	2.17	• •		
. WE,	$\frac{1}{2}(2)\left[(2)\left(\frac{0.25}{15.5}\right)\right](0.125)$			2.17			
a (u) e	(3)(0,125)(5,6)			2 10			
	$\frac{1}{2}(3)(3)(\frac{1}{5\cdot5})(0.125)$		 	2.11			
t www.	(10.5)(0.0625)(8.73)			5.73			
E	$\frac{1}{2}(10.5)(10.5)(-15.5)(0.0625)$	_		5.79			
W.	(0, 125)(14, 8)(5, 7)			10.55			
	-(0,125)(0,7)(5,7)(0,5)			- 0.25			
C	$+ W_{\rm E} + W_{\rm W} + W_{\rm N}$			29.69			
Hia	0.656(10.5)(2)	3.444					
His	$(1.0783 - 0.656)(9.2)(\frac{1}{2})$	1.943					
Hic	0.656 (9.2)	6.035					
UH	(0.5) (10.2) (0.8073)		4.117				
HA	$\left[\frac{0.0625(9.2)^{2}}{2}\left[\tan^{2}\left(45^{\circ}-\frac{20^{\circ}}{2}\right)\right]\right]$	1.297					
	$\left[-2.0(0.7)(9.2)\left[\tan(45^{\circ}-\frac{20^{\circ}}{2})\right]\right]$	- 9.019					
Hia + His	$+ H_{ic} - U_{H} + H_{a} =$	7.305					
	-						
:							

WES FORM NO. 1253

Н-8
$$\frac{V_{H,ET}}{SAMPLE RETAINING WALL
SAMPLE RETAINING WALL
SHEAR RETAINING WALL
SHEAR$$

H-9

S. B ± - 1	EXHIBIT H	COMPLETE HY	2-A * F	F
SAMPLE	RETAINING WALL			
the f	orces which champe d	ue to checkin	g trial	failure
surfa	ce 2 is When the	resistance for	d drivin	me forces
atter	ted b. changes in	uplift forces a	over new	crap
Path ,	and uplift forces (V	erticals).		·
(1)	-(0.125)(14.8)(5.9)		c	
~~~~	-(0.125)(0.7)(5.7)	(0.5) = -0.2	<u>S</u>	
		10.3	0	
<b>Z</b> ∨ =	ZC + WEI + WEZ	$+ W_{W} + W_{N}$	- 0	
-	= 9.32 + 2.17 + 2.11	+ 5.79 + 10.30	- 15.367	17
=	= 14.32 KIPS			
Hen	$= \frac{1}{2} (10.2)^2 (0.0625) ($	fan2 (45° + 2	0/2))	
4	= 3.25 (2.04) = 6.0	63 KIPS		
H	$= 2(0,7)(10,2)/4am^{2}$	$(45^{\circ} + 20/2)$		
<b>2</b> 11				
	= 20.39 KIPS			
Hp:	$= H_{PQ} + H_{PC} = 6.63$	3 + 20.39 =	27,02	KIPS
ZR =	ZV. tand + C.	A + He		
	- 1122 tan20° +	0.7 (14.8) +	29.02	
			21.02	
	5.212 + 10.36	+ 210 Z		
-	= 42.592			
shear	friction safety fac	tor	- 7R	47.592
surfa	ice 2)	fairure = S.	$F = \frac{z_H}{z_H}$	
		<b>=</b> 5	. 8 3	
		_ •		
				I

**.**...

H-10

SUBJECT EXHIBIT H	COMPUTED BY:		DATE	FILE NO.
SAMPLE RETAINING YVALL SHEAR FRICTION SLIDING ANALYSIS	CHECKED BY:		DATE	SHEET NO. //
SUMMAR	<u>Y</u>	<u> </u>		
Failure Surface #1	-			-S = 5.42
Failure Surface #2	-	¥		-S= 5.83
So, for these two limiting	failure p	larres	FS=	5.42

SUBJECT EXHIBIT H	COMPUTED BY	DATE.	FILE NO.
SAMPLE RETAINING WALL			
SHEAR FRICTION SLIDING ANALYSIS	CHECKED BY	DATE	SHEET NO.

The total base-foundation interface of the T-wall may not be in compression with the foundation. If any part of the surface under consideration is along the base-foundation interface and is not in contact with the foundation, this portion should be neglected when obtaining the effective base area to resist sliding. However, if the Assumed failure surface is not along the base-foundation interface but through the soil, no reduction in the Area to resist sliding is made.

A portion of the base of A T-wall will not be in compression when the resultant falls outside the kern thus creating A crack which can result in an increase in uplift pressures. This condition will affect the sliding stability Analysis when the Assumed sliding plane Acts Along the soil - structure interface below the base of the wall (for this condition the program will have to recycle back through the line of creep calculations until the creep path Assumptions match the final part of the base that is in contact

NES FORM NO. 1253

H-12

SUBJECT EXHIBIT H	COMPUTED BY:	DATE	FILE NO.
SAMPLE RETAINING WALL SHEAR FRICTION SLIDING ANALYSIS	CHECKED BY:	DATE	SHEET NO.
		· · · · · · · · · · · · · · · · · · ·	
with the foundation). Fo	r example, cons	ider A wa	11
without a key and with	A horizontal base	, when the	2
resultant falls outside the	kern and the a	issumed sli	iding
plane is along the interf	ace between the	base of	the
structure and the soil f	oundation, uplift	pressure	S
will be computed Assur	ming no creep lo.	ss for the	2
portion of the foundat	ion not in con	npression	' <b>.</b>
For the condition when	e the resultant	falls out	side
the kern but the Assumed	sliding plane is	through	the
soil, for example A wall	with A key posi-	tioned at	the
extreme end of the he	el (see figure sho	own below	ر ( ر
NO INCREASE IN UPLIFF P	ressure will be	considere	d
because the soil does w	of lift And form	A crack	As
is the case at the soil	-structure inte	rface.	
<u>-</u>			
	Assumed Failure	plane	
		-	
WES FORN NO. REV OCT 1968 1253			

SUBJECT EXHIBIT H	COMPUTED BY:	DATE	FILE NO.
SAMPLE RETAINING WALL			
SHEAR FRICTION SLIDING ANALYSIS	CHECKED BY:	DATE	SHEET NO.

Another reason the uplift forces are not affected for this condition is because they are forces inside the soil-structure free body and therefore, do not affect the overall sliding stability.

The  $\phi$  and C values should be consistent with the material being sheared. A plane of failure through the soil should use the  $\phi$  and C of the soil. For any of the failure planes along the soil-structure interface, use the  $\phi$  and C for sliding friction at the interface.



I-1



1-2



WES FORM NO. 1252 REV OCT 1968 1252

I-3

. . . . EXHIBIT I . . . . . SAMPLE FLOODWALL SLIDING ANALYSIS NET APPLIED FORCE TENDING TO INDUCE SLIDING ∑ D = (H1-H2-VH) cosw + HA. (See sheet H-5 for H1, H2 ¢ UH Values) =(11.316-0.827 - 3.974) (0521.06 + 2.4683 kn - 19.1910 C / Ka = 6.08 + 2.4683 KA - 17.1710 CJ KA REACTION FORCES TENDING TO RESIST SLIDING  $\Sigma R_{W} = H_{R_{W}} + H_{P_{W}} = [\Sigma V. (OSW + \Sigma H. Sinw]. tam p' + c'.A + H_{P_{W}}$ =[12.8558.cos21.06 + {11.316-0.829-3.994} sim 21.06 + 12 (0.125 - 0.0625) KA · (9.2) - 2. C (9.2) KA SIN 21.06 -{± (0,125 - 0,0625).kp. (4,5)2 + 2. (-(4,5)/kp]. Sin 21.06"]. tan \$\$\$\$ + 2. (-(4,5)/kp]. Sin 21.06"]. 15.86 c + 0.5905 kp + 8.349. c JE, = [14.338+ 0.9505 KA - 6.612 c/KA - 0.2294 Kp - 3.23411 c/Kp]. tan \$\$\$ 15.86C + 0.5905 kp + 8.399. C/Kp EQUILIBRIUM OF APPLIED AND REACTION FORCES CONSTDERING TRIAL FAILURE SURFACE 1 WITH  $[\Sigma D_w = f_1 (FS, FS + 2C')]$  AND  $\Sigma R_w = f_2 (FS, FS + 2C')]$ , The equiLIBRIUM Telationship [ZDw = ZRw] becomes  $[f_1(FS, FS+2c') = f_2(FS, FS+2c')] \text{ which } can be \\ transposed to read[f_1(FS, FS+2c') - f_2(FS, FS+2c') = 0]$ The computation of FS involves solving the expression  $\Sigma D_{\rm W} - \Sigma R_{\rm W} = 0$  $f_1(FS, FS + 2c') - f_2(FS, FS + 2c') = 0$ by an iterative procedure, either graphically (by hand) by an iterative procedure, either graphically (by hand) or analytically (on the computer). The procedure includes (1) assuming a trial value of FS, (2) calculating allowable values of  $\phi'$  and c', (3) calculating ka and kp from  $\phi'$ , (4) substituting c', ka, and kp into Z Dw and Z Rw, and (5) either plotting Z Dw = fi (FS, FS+2c') and Z Rw = fi (FS, FS+2c') to get the the point of intersection or solving for FS to make ZDW - ZRW = 0 and a search in the second s

1-4

COMPUTED BY DATE FILE NO. SUBJECT EXHIBIT Ι SAMPLE FLOODWALL SLIDING ANALYSIS CHECKED BY DATE SHEET NO. 5 An example of the above procedure is as follows :  $\phi' = \tan^{-1}\left(\frac{\tan \phi}{FS}\right)$  where  $\phi = 20^{\circ}$  $C' = \frac{C}{FS + 2q}$  where q = C' $C = 0.35 \pm 5f$ The positive value of c' is  $C' = \frac{\sqrt{FS^2 + 8C} - FS}{4}$  where C and C' is in tsfFor FS = 1  $\phi' = \tan^{-1}\left(\frac{\tan 20^{\circ}}{1}\right)$ = 20[°]  $C' = \frac{\sqrt{(1)^2 + 8(0.35)} - 1}{4} = 0.23734 + 51$ or 0.49468 Ksf  $\tan \phi' = \tan 20^\circ = 0.36397$  $k_{A} = \tan^{2}(4s^{\circ} - \frac{\phi'_{2}}{2}) = \tan^{2}(4s^{\circ} - 2o^{\circ}/2) = 0.49029$  $k_p = \tan^2(45' + \Phi'/2) = \tan^2(45' + 20'/2) = 2.03961$  $\sqrt{k_{P}} = \sqrt{2.03961} = 1.42815$ HA. = 2.4683 KA - 17.1710 C JKA = 1.21018 - 5.70720 = -4.50 (meglect if megative)  $ZD_{W} = 6.08 + 2.4683 k_{A} - 17.1710 C'\sqrt{k_{A}} = 6.08 + H_{A_{W}}$ = 6.08 - 4.50 = 6.08

WES FORM NO. 1253

I-5

COMPUTED B SUBJECT DATE FILE NO. EXHIBIT I SAMPLE FLOODWALL SLICING ANALYSIS HECKED BY DATE SHEET NO. 6  $H_{R_{\phi}} = [14.338 + 0.9505 k_{A} - 6.612 c' \sqrt{k_{A}} - 0.2294 k_{P}$  $-3.23411 c' / k_{f} + 4am \psi' = 3.62$  $Hr_{c} = 15.86c' = 7.53$ Hpg = 0.5905 Kp = 1.20  $H_{P_c} = 8.399 c'\sqrt{k_P} = 5.69$  $ZRW = HR\phi + HRc + HP\phi + HPc = 18.04$  $Z P_{W} - Z D_{W} = 18.04 - 6.08 = 11.96$ Continue assuming safety factors and calculating ERW-ZDW until their difference is negligitle. The system is then in equilibrium with the driving forces equal to the resisting forces and the safety factor which produces this condition is the safety factor against sliding.

sue.	ECT		E X	(H	IBI	Т	I				COM	PUTED BY	04	TE		FILENG
S	. A P	1PL DIJ	E NG		FLC A N	ALY	WA SL	ււ Տ			CHE	CKED BY:	D	16		SHEET NO T
	GF	LAI	PHI	CAL	=	SOL	UT	101	۷				1			
ZR 20.	11.96	7.48	4.80	1. 72	0.04	0.03	0.02	0.01	0.0	<b>8</b> 0 ° ° 0 –		mg FS+2c') f with				- 4
∑ R w	18.04	13.56	10,88	1.80	6.12	6.11	6 · 10	6.09	6.08	6,00	ຸ ພິ	50. di (FS) = 4.0			ę ing	
HPc	5.69	3.19	3.06	2.06	1,54	1.54	1.54	1.53	1.53	·5	Ϋ́́́́́́́́́́́́́́́́́́́́́́́́́́́́́́́́́́́́	ksist tot			Cros:	
HPR	1.20	0.95	0.85	0.75	11.0	١٢٠٥	11.0	11-0	11.0	16.0	um e	tion to			imete S = 4.0	-
HRC	1,53	5.93	4.82	3.45	2.66	2.66	2.65	2.65	2.64	2.60	5 S S (	10° 0			APProx APProx	AFEIT
HEF	3.62	2.69	2.15	1.54	1.2.1	1.20	1.20	1.20	1.20	8 I÷	alue	tive = m				of s
Z Dw	6.08	6.08	6.08	6.08	6.08	6.08	6.08	6.08	80.9	6.08	V K V	force there there are a second to the second	/	/		ACTOR
HA.	- 4.50	- 3.5 2	- 2.63	- 1.37	- 0.58	-0.57	-0.56	- 0.56	- 0, 55	-0.51	regat	2 40 - 22 - 52			orce	E -   V -   V
c	0.475	0.374	0.304	0, 21 8	0.168	0.168	0.167	0.167	0.166	0.164	т фт	R - fr			lied F	2 IR
<u> </u>	1.43	1, 2 T	1.20	ŀ-13	0  - 	1,09	1.09	1.09	1.09	1,09	*	R R.			A APP	-
k e	2.04	l, 6 2	44	۲2.1	1,20	1.20	1.20	1.20	1.20	1.19		$\sim$	/		υ ⁻ Ν.	
k۸	0.440	0.618	0.696	0.185	0.834	0.834	0.835	0.835	0,835	o. 838				ļ	A W	-
tano'	0.364	0.243	0.182	0.121	160.0	160.0	0.041	0,090	060.0	0.089			1141	<b>1</b> г :		
¢	0.02	13.64	10.31	56.9	5,20	5.19	5.19	5,16	5.15	5,07	1.4	EAR FT)	<u>و</u> s / רו או	   7	רב (	- <del>°</del> -
FS	0.1	ŝ	5.0	3,0	4 0	4.01	4.02	4.03	4.04	410						
WES P		40. 168	1253													

1-7



1-8





I-9

COMPLITED BY DATE FILE NO SUBJEUT EXHIBIT I SAMPLE FLOODWALL SLIDING ANALYSIS SHEET NO. 10 CHECKED BY DATE NET APPLIED FORCE TENDING TO INDUCE SLIDING ZDn = Hi - Un + HA ( See Sheet H-5 for Hi and Un values) = 11.422 - 4.117 + 2.645 kA - 18.4 C/KA = 7.305 + 2.645 KA - 18.4 C / KA REACTION FORCES TENDING TO RESIST SLIDING ZRH = HRW + HPW = V. tamp' + C. A + 3.2513 Kp + 20.4 C/Kp = 14.3223  $\tan \phi' + 14.80c' + 3.2513 kp + 20.4 c' / ke$ EQUILIBRIUM OF APPLIED AND REACTION FORCES WITH [ $zD_{H} = f_{1}(FS, FS + 2c')$  AND  $ZR_{H} = f_{2}(FS, FS + 2c')$ ], the equiLIBRIUM relationship [ ZDH = ZRH ] becomes  $[f_1(FS,FS+2c') = f_1(FS,FS+2c')]$  which can be transposed to read [filFs, FS+2c') - fr(Fs, FS+2c')=0] the computation of FS involves solving the expression  $z D_{H} - z R_{H} = 0$  $f_1(FS, FS + 2c') - f_2(FS, FS + 2c') = 0$ by an iterative procedure, either graph: (all, (by hand)) or analytically ( on the computer). The procedure includes (1) assuming a trial value of FS, (2) calculating allowable values of  $\phi'$  and c', (3) calculating ka and kp from  $\phi'$ , (4) substituting c', ka, and kp into  $\mathbb{Z}$  DH and  $\mathbb{Z}$  RH, and (S) either plotting  $\mathbb{Z}$  DH = ficFS, FS+2c') and  $\mathbb{Z}$  RH = f2 (FS, FS+2c') to get the point of intersection or solving for FS to make Z DH - ZRH = 0

WES FORM NO. 1253 REV OCT 1968

SUØ.	ECT	:	E	XT	71	BI	τ	I					co	MPUT	TED BY:		DATE		F	ILE NO	).
	SA	HP		5	1	FL		) D ( . Y S	15	LL			сн	ECKE	ED 84:		DATE	:	S	HEETI	10.1
	<u>G R</u>	A	5 H J	Ū	L.	S	٥L	ŬĪ	10	N		_							1		
ZRH-ZBH	2S. 39	16.65	11.90	6.80	4.13	2,50	1,04	0.06	500 <b>3</b>	0.02	0.0	- 0.01	+0.0-		st zc') with						
z₽H	32.10	23.96	19.21	11-+1	1.44	9.81	11.8	8.00	7.98	7.97	1.96	1.95	9.93	0	liding FS, F 6.96					P	
He	13.83	9.69	7.43	S. et	3.75	2-99	2.48	2.15	2.14	2-14	2.13	2.13	21.2	Hab =	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1						
HPe	6.63	5.26	4.67	4.14	3.90	3.96	3.67	3.61	3.61	3.61	3-61	3.61	3.61	ž	resizer					1.6.4 2.5 6.9	
HRC	7.03	5.53	4.50	3.22	2.49	20.2	1.69	- <del>4</del> 8	1.48	-+-	1-41	<b>641</b>	1.46	a 55 u	to alutio six					et to	VIELY
Hee	5, 21	3.48	2.61	1-14	1.30	1.04	0.81	96.96	0, 7S	0. 1S	56.0	0.15	914	(re )	Aind M = 0						5
₹D _H	1.31	1.31	1.31	1.31	1.31	1.31	የፊገ	9.94	1.95	7.95	96.6	1.46	7.47	k vel	ce te trativ c´) =	0 0 1		ł			
HA NO	- 4.82	-3.11	-2.82	-1.47	- 0.62	- 0.04	0.36	0.63	0.64	0.65	0.65	0.65	0.66	rgativ	よ 「 「 こ よ こ よ こ 、 む い む い し い し い し い し い し い	ă N I					AL EA
c,	0.475	0.374	0.304	0.218	0.168	0.136	0.114	0.100	0.100	0.019	\$60.0	0.099	9.099	ير مر	eactic mput	μη Μ. Μ		Force			
16	1.43	1.29	1.20	51.1	01-1	80.1	1.06	So.I	1.05	50.1	1.05	1.05	1.05	*	ຊີບັງ 1 ສັ	É		Applia			
k,	2.04	1.62	1.44	1, 29	1.20	1.16	5 i 3	11-1	Ę	F.	=-	1.1	R-		N	/	/	= Het			
k A	0.440	0.618	0.696	0.185	0.834	0.865	6 8 8 6	0.900	0.100	0.900	106.0	101.0	0.901		/	/		₩ \			Ē
tam ¢	0.364	0.243	0.182	0.121	10.0	0.013	0.061	0,053	0.053	0.052	0.052	0.052	0.052				L	\ 			
φ′	20.0	13.64	10.31	6.92	5.20	4-16	3.41	3.02	10.6	3.00	2.99	2.99	86-2		AR FT)	א דו אנ	1291	<u>ه</u> ۲ ۲	3280	) I	•
FS	0	<u>ب</u>	0 %	0.	04	5.0	6.0	6.9	56.9	6.94	6.46	6.99	100								

I-11

SUBJECT: EXHIBIT I SAMPLF FLOOD WALL SLIDI: G ANALYSIS	COMPUTED BY:	DATE:	FILE NO.
--------------------------------------------------------------	--------------	-------	----------

The total base-foundation interface of the T-wall may not be in compression with the foundation. If any part of the surface under consideration is along the base-foundation interface and is not in contact with the foundation, this portion should be neglected when obtaining the effective base area to resist sliding. However, if the assumed failure surface is not along the base-foundation interface but through the soil, no reduction in the area to resist sliding is made.

A pontion of the base of A T-wall will not be in compression when the resultant falls outside the kern thus creating A crack which an result in AN increase in uplift pressures. This condition will affect the sliding stability Analysis when the Assumed sliding plane Acts along the soil - structure interface below the base of the wall (for this condition the program will have to recycle back through the line of creep calculations until the creep path Assumptions match the final part of the base that is in contact

SUBJECT EVUIDITT	COMPUTED BY:	DATE:	FILE NO.
SLIDING ANALYSIS	CHECKED BY:	DATE:	SHEET NO.
		1 <u></u>	
	- ·		
with the foundation). I	or example, cons	sider A wa	11
without a key and with	A horizontal base	, when the	٤
resultant falls outside th	e kern and the 1	Assumed sl	purp
plane is along the inter	face between the	wase of	the
structure and the soil	foundation, uplift	pressure	. <b>S</b>
will be computed Assi	aming no creep lo	ss for the	2
portion of the founda	tion not in con	npression	1.
For the condition when	re the resultant	- falls out	side
the kern but the Assumed	sliding plane is	s throug h	the
soil, for example A wa	ll with A key posi	tioned at	the
extreme end of the h	eel (see figure sh	own below	و(ں
No increase in upliff	pressure will be	considere	ed
because the soil does a	ist lift and form	A crack	As
is the case at the soi	1-structure inte	rface.	
	Assumed Failure	plicme	

1-13

SUBJECT	EXHIBIT I SAMPLE FLOODWALL	COMPUTED BY:	DATE	FILE NO.
	SLIDING ANALYSIS	CHECKED BY:	DATE:	SHEET NO.

Another reason the utility forces are not affected for this condition is because they are forces inside the soil-structure free body and therefore, do not affect the overall sliding stability.

The  $\phi$  and C values should be consistent with the material being sheared. A plane of failure through the soil should use the  $\phi$  and C of the soil. For any of the failure planes along the soil-structure interface, use the  $\phi$  and C for sliding friction at the interface.

WES FORM NO. 1253





The second

.

;

1,

J-2



NES FURMIN 1253 HEVIDET 1968

COMPLITED BY DATE FIL- NO EXHIBIT J SAMPLE SLIDING FLOODWALL ANALYSIS CHECKED RV DATE HEE NO. 4 NET APPLIED FORCE TENDING TO INDUCE SLIDING ∑ D = (H1-H2-VH) cosus + HA. (See sheet H-5 for H1, H2 & VH Values) =(11.316-0.827 - 3.974) (05 21.66 + 2.4683 kA - 19.1910 C / KA = 6.08 + 2.4683 KA - 19.1910 CV KA REACTION FORCES TENDING TO RESIST SLIDING  $\Sigma R_{W} = H_{R_{W}} + H_{P_{W}} = [ZV. COSW + \Sigma H. Sinw] + tan \phi' + c'.A + H_{P_{W}}$ =[12.8558.cos21.06 + \$ 11.316 - 0.827 - 3.914} sim 21.06" + 「之(0.125-0.0625)·KA·(9.2)-2·C(9.2)/KA子EIM 21.06-{± (0,125 - 0,0625). kp. (4,5)2 + 2. (* (4,5)/ kp.]. Sin 21.06"]. tan \$\$\$ + 15.86 C + 0.5905 Ky + 8.399. C JE, = [14.338+ 0.9505 KA - 6.612 c/KA - 0.2274 Kp - 3.23411 c/Kp]. tom \$\$\$ + 15,86c' + 0,5905 kp + 8,399, c/kp EQUILIBRIUM OF APPLIED AND REACTION FORCES CONSIDERING TRIAL FAILURE SURFACE 1 WITH [ZDw=fi(FS) AND ZRN=fz(FS)], the equilibrium relationship [ZDw = ZRw] becomes  $[f_1(FS) = f_2(FS)]$ which can be transposed to rend  $[f_1(FS) - f_2(FS) = 0]$ The computation of FS involves solving the expression  $ZD_{W} - ZR_{W} = 0$  $f_1(FS) - f_1(FS) = 0$ by an iterative procedure, either graphically (by hand) or analytically ( on the computer). The procedure includes (1) assuming a trial value of FS, (2) calculating allowable values of  $\phi'$  and c', (3) calculating ka and kp from  $\phi'$ , (4) substituting c', ka, and kp into ZDw and ZRw, and (5) either plotting ZDW = filFS) and ZRW = filFS) to get the point of intersection or solving for FS to make ZDw - ZRw = U

SUBJ	ECT	E	Y	нı	BI	τ	۰J					Tee	MPI	TED BY	<u>/:</u>		DATE		FILE NO	<b>.</b>
51	M		16	F		AL	NW Y		L S			CH	ECI	ED BY			DATE	·	SHEET	vo. 5
7	<u>2 K</u>	AP	HI	<u>C A</u>	<u> </u>	50	LU	11	ON	L					2					
ZP, -ZD	11.49	4.74	5, 91	2, 11	0,21	0.07	0.06	50.0	0.03	20.0	0.01	-0.01	- 0.06		= A =					-4
≊R."	23.57	15, 8 2	11.99	8.19	6.29	6.15	6.14	6.13	6.11	6.10	6.09	6.07	6.02		skidi (FS)	10.0			Bw1 St	
H۹	8.40	4.98	3.52	2.21	1.61	l, s T	1.56	1.56	1.55	1.55	1.55	I. 5 <b>4</b>	1.53	0	4515 +- (23	≊ D⊷ ⊧		ļ	Cro	
HPE	1.20	0,95	0.85	sto	11.0	١L ю	01.0	010	0 L •0	01.0	01.0	01.0	010	A	a y	3			janate = 4.15	VETY
HRC	11-10	1.40	5,55	3.10	2,78	11.2	2.70	2.69	2.69	2.68	2.68	2.67	2.64	Ч Н	Ling of	N II C			Approx ed FS	3 54
HRG	2,86	2.48	2.07	1.53	1.20	61.1	1.1	61.1	L۱۰۱	1.16	1.16	1-16	sı i	WOSS.	+ energy and a series of the s	4				L 1
Z Dw	6.08	6.08	6.08	6.08	6.08	6.08	6.08	6.08	6.08	6.05	6.08	6.08	6.08	د م	furce We S	3			   	FACIO
HA w	- 1, 21	-4.11	-3.30	- 1.61	-0.69	-0.62	-0.61	- 0.60	- 0.60	- 0.59	-0.58	-0.58	- 0.55	valu	tion	14-5			w ce	1 T
`ن	01.0	0.467	0.350	0.233	0.175	161.0	0110	061.0	0.169	0.169	0.169	0.168	0.167	tive	- Reac	Ed Fa			1.64	
∕ ¥⊳	ŀ43	1.27	1.20	113	01:1	1.09	1.09	1.09	1.09	1.09	1.09	1.09	1.09	mega	Z Rw : Comp	لألاءالح	/		d d V	-
kр	2.04	1.62	1.44	1.21	1,20	614	119	61.1	614	614	61.1	61.1	61.1	tr *	$\searrow$				Fon -	
k۸	0.490	0.618	0.696	0.785	0.834	0.838	0.838	0.838	0.839	0.839	0.839	0.840	0.841		/	/			N N −	F
tamp'	0.364	0.243	0.182	0.121	160.0	0.089	0.089	0,088	0.088	0,088	0.088	0.087	0,087	<u></u>		.1.1				
φ	20.0	13.64	10.31	592	5,20	5.07	5,06	5.05	5.04	5.02	5.01	5.00	4.95	(	18 FT	NEN	17/S	ē KIP	Force (	0
FS	0	2	0 3'0	3.0	4 0,4	4	4.11	4.12	413	4-14	4.15	4.16	4.20						<u></u>	

J-5

4 12 12



1

J**-**6



.

SAMPLE EXHIBIT J SAMPLE EXHIBIT J SAMPLE ELDODWALL SLIDING ANALYSIS NET APPLIED FORCE TENDING TO INDUCE SLIDING $ZD_{H} = H_{I} - U_{H} + H_{A}$ (See Sheet H-S for H, and U _H value $= 11.422 - 4.117 + 2.645 k_{A} - 18.4c'/k_{A}$ $= 7.305 + 2.645 k_{A} - 18.4c'/k_{A}$ REACTION FORCES TENDING TO RESIST SLIDING $ZR_{H} = H_{R_{W}} + H_{P_{W}} = V. tam \phi' + c'.A + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}$ $= 14.3223 tam \phi' + 14.80c' + 14$				······
NET APPLIED FORCE TENDING TO INDUCE SLIDING $ZD_{H} = H_{I} - U_{H} + H_{A}$ (See Sheet H-S for H, and U _H video $= 11.422 - 4.117 + 2.645 k_{A} - 18.4c'/k_{A}$ $= 7.305 + 2.645 k_{A} - 18.4c'/k_{A}$ <u>REACTION FORCES TENDING TO RESIST SLIDING</u> $ZR_{H} = H_{R_{U}} + H_{P_{W}} = V. tan \phi' + C'.A + 3.2513 k_{F} + 20.4c'/k_{F}$ $= 14.3223 tan \phi' + 14.80c' + 3.2513 k_{F} + 20.4c'/k_{F}$ <u>EQUILIBRIUM OF APPLIED AND REACTION FORCES</u> <u>CONSIDERING TRIAL FAILURE SURFACE2</u> With [ZD_{H} = f.(FS) AND ZR_{H} = f_{2}(FS)], the equilibrium relactionship [ZD_{H} = ZR_{H}] be comes [f_{1}(FS) - f_{2}(FS)=0] the computation of FS involves solving the expression $ZD_{H} - ZR_{H} = 0$ $f_{1}(FS) - f_{2}(FS) = 0$ by an iterative procedure, either graph: cally (L Rand or analytically (on the computer). The procedure includes (1) assuming a trial value of FS, (2) calculating allowable values of $\phi'$ and c', (3) calculating ka and kp from $\phi'_{1}$ (4) Substituting c', ka and kp into ZD_{H} and ZR_{H}, and (S) either	SAMPLE FLOODWALL SLIDING ANALYSIS	COMPUTED BY	DATE	FILE NO.
$ \Sigma D_{H} = H_{I} - U_{H} + H_{A} ( See Sheet H-5 for H, and U_{H} Value = 11.422 - 4.117 + 2.645 k_{A} - 18.4c'/k_{A} = 7.305 + 2.645 k_{A} - 18.4c'/k_{A}  REACTION FORCES TENDING TO RESIST SLIDING   \Sigma R_{H} = H_{RO} + H_{PW} = V. tan \phi' + C.A + 3.2513 k_{P} + 20.4c'/k_{P} = 14.3223 tan \phi' + 14.80c' + 3.2513 k_{P} + 20.4c'/k_{P}  EQUILIBRIUM OF APPLIED AND REACTION FORCES  CONSIDERING TRIAL FAILURE SURFACE2  With [Z D_{H} = f.(FS) AND ZR_{H} = f_{2}(FS)], the equilibrium relactionship [ ZD_{H} = ZR_{H}] becomes [f_{1}(FS) - f_{2}(FS)=0]  the computation of FS involves solving the expression  Z D_{H} - ZP_{H} = 0  f_{1}(FS) - f_{2}(FS) = 0  by an iterative procedure, either graph: call ( l_{Raudi of analytically ( on the computer). The procedure  includes (1) assuming a trial value of FS, (2) calculating allowable values of \phi' and c', (3) calculating k_{A} and k_{P} from \phi', (4) Substitutingc', k_{A} and k_{P} into ZD_{H} and ZR_{H}, and (S) either$	NET APPLIED FORCE TEN	DING TO IND	UCE SL	IDING
= 11.422 - 4.117 + 2.645 k _A - 18.4c ² /k _A = 7.305 + 2.645 k _A - 18.4c ² /k _A <u>REACTION FORCES TENDING TO RESIST SLIDING</u> Z R _H = H _R _N + H _{P_W} = V. $\tan \phi'$ + C ² A + 3.2513 k _P + 20.4c ² /k _P = 14.3223 $\tan \phi'$ + 14.80c ² + 3.2513 k _P + 20.4c ² /k _P <u>EQUILIBRIUM OF APPLIED AND REACTION FORCES</u> <u>CONSIDERING TRIAL FAILURE SURFACE2</u> With [Z D _H = f ₁ (FS) AND Z R _H = f ₂ (FS)], the equilibrium relationship [ZD _H = Z R _H ] becomes [f ₁ (FS) + f ₂ (FS)] which can be transposed to read [f ₁ (FS) - f ₂ (FS)=0] The computation of FS involves solving the expression Z D _H - Z P _H = 0 f ₁ (FS) - f ₂ (FS)=0 by an iterative procedure, either graph: cdl ₂ (k ₂ Rady or analytically (on the computer). The procedure includes (1) assuming a trial value of FS, (2) celculating allowable values of $\phi'$ and c', (3) calculating k _A and k _P from $\phi'$ , (4) Substituting c', k _A and k _P into Z D _H and Z R _H , and (S) either	$\Sigma D_{H} = H_{I} - U_{H} + H_{A}$ ( Se	e sheet H-5	for H. an	d Un values)
= 7.305 + 2.645 RA - 18.4C/KA <u>REACTION FORCES TENDING TO RESIST SLIDING</u> $ZR_{H} = H_{R_{W}} + H_{P_{W}} = V. tan \phi' + C.A + 3.2513 kp + 20.4 c/Fp = 14.3223. tan \phi' + 14.80c' + 3.2513 kp + 20.4 c/Fp EQUILIBRIUM OF APPLIED AND REACTION FORCES CONSIDERING TRIAL FAILURE SURFACE2 With [ZD_{H} = f.(FS) AND ZR_{H} = f_2(FS)], the equilibrium relationship [ZD_{H} = ZR_{H}] becomes [f.(FS) = f_2(FS)] which can be transposed to read [f.(FS) - f_2(FS)=0] The computation of FS involves solving the expression ZD_{H} - ZP_{H} = 0f.(FS) - f_2(FS) = 0by an iterative procedure, either graph: cally (by fandor analytically (on the computer). The procedureincludes (1) assuming a trial value of FS,(2) calculating allowable values of \phi' and C',(3) calculating KA and kp from \phi', (4) substitutingC', KA and kp into ZD_H and ZR_H, and (S) either$	= 11.422 - 4.117 + 100	2-645 kA - 18.40	:/ KA	
REACTION FORCES TENDING TO RESIST SLIDING $ZR_{H} = H_{R_{W}} + H_{P_{W}} = V. \tan \phi' + C.A + 3.2513 kp + 20.4 c/Fe = 14.3223 tan \phi' + 14.80 c' + 3.2513 kp + 20.4 c/Fe EQUILIBRIUM OF APPLIED AND REACTION FORCES CONSIDERING TRIAL FAILURE SURFACE? With [ZD_{H} = fi(FS) AND ZR_{H} = f_2(FS)], the equilibrium relationship [ZD_{H} = ZR_{H}] becomes [fi(FS) - f_2(FS)] which can be transposed to read [fi(FS) - f_2(FS)=0] The computation of FS involves solving the expression ZD_{H} - ZR_{H} = 0 fi(FS) - f_2(FS) = 0 by an interative procedure, either graph: cally (by fault or analytically (on the computer). The procedure includes (1) assuming a trial value of FS, (2) calculating allowable values of \phi' and c',(3) calculating ka and kp from \phi', (4) substitutingc', ka and kp into ZD_{H} and ZR_{H}, and (S) either$	= 7.305 + 2.645 K	a - 18.4 C J Ka		
$ZR_{H} = H_{RW} + H_{PW} = V. tan \phi' + C.A + 3.2513 k_{P} + 20.4 c/F_{P}$ $= 14.3223 tan \phi' + 14.80c' + 3.2513 k_{P} + 20.4 c/F_{P}$ $\frac{EQUILIBRIUM OF APPLIED AND REACTION FORCES}{CONSIDERING TRIAL FAILURE SURFACE2}$ $With [ZD_{H} = f_{1}(F_{S}) AND ZR_{H} = f_{2}(F_{S})], the equilibrium relationship [ZD_{H} = ZR_{H}] be(omes [f_{1}(F_{S}) - f_{2}(F_{S})])$ $Which can be transposed to read [f_{1}(F_{S}) - f_{2}(F_{S}) = 0]$ $The computation of FS involves solving the expression ZD_{H} - ZR_{H} = 0$ $f_{1}(F_{S}) - f_{2}(F_{S}) = 0$ $b_{2} an iterative procedure, either graph: cally (l_{2} Randor on alglically (on the computer). The procedure includes (1) assuming a trial value of FS, (2) calculating allowable values of \phi' and c', (3) calculating ka and kp from \phi', (4) substituting c', ka and kp into ZD_{H} and ZR_{H}, and (s) either$	REACTION FORCES TENDI	NG TO RESIST	SLIDII	<u>ve</u>
= 14.3223 tan φ' + 14.80 c' + 3.2513 kp + 20.4 c V kp EQUILIBRIUM OF APPLIED AND REACTION FORCES CONSIDERING TRIAL FAILURE SURFACE 2 With [Z Dn = fi(FS) AND ZRH = f2(FS)], the equilibrium relationship [ZDH = ZRH] becomes [fi(FS) = f2(FS)] Which can be transposed to read [fi(FS) - f2(FS)=0] The computation of FS involves solving the expression Z DH - ZRH = 0 fi(FS) - f2(FS) = 0 by an iterative procedure, either graph: celly (by Rand or analytically (on the computer). The procedure includes (1) assuming a trial value of FS, (2) calculating allowable values of φ' and c', (3) calculating ka and kp from φ', (4) substituting c', ka and kp into ZDH and ZRH, and (S) either	$ZR_{H} = H_{RW} + H_{PW} = V. ta$	$m\phi' + c' + 3$	2513 kp +	20.4014
EQUILIBRIUM OF APPLIED AND REACTION FORCES <u>CONSIDERING TRIAL FAILURE</u> SURFACE2 With [ZDH = fi(FS) AND ZRH = f2(FS)], the equilibrium relationship [ZDH = ZRH] becomes [fi(FS) = f2(FS)] which can be transposed to read [fi(FS) - f2(FS)=0] The computation of FS involves solving the expression ZDH - ZRH = 0 fi(FS) - f2(FS) = 0 by an iterative procedure, either graph: cally (by Rand or analytically (on the computer). The procedure includes (1) assuming a trial value of FS, (2) calculating allowable values of \$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$	= 14.3223 $\tan \phi' +$	14,80C1 + 3,251	$sk_p + zc$	D. 4. C√Kp
With [ZDn = fi(FS) AND ZRH = f2(FS)], the equilibrium relationship [ZDH = ZRH] becomes [fi(FS) = f2(FS)] which can be transposed to read [fi(FS) - f2(FS)=0] the computation of FS involves solving the expression ZDH - ZRH = D fi(FS) - f2(FS) = 0 by an iterative procedure, either graphically (by Rand) or analytically (on the computer). The procedure includes (1) assuming a trial value of FS, (2) calculating allowable values of \$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$	EQUILIBRIUM OF APPLIED A CONSIDERING TRIAL FAILU	AND REACTION F	-	
by an iterative procedure, either graph: call, (by hand or analytically (on the computer). The procedure includes (1) assuming a trial value of FS, (2) calculating allowable values of $\phi'$ and $c'$ , (3) calculating ka and kp from $\phi'$ , (4) substituting c', ka and kp into ZDy and ZRy, and (5) either	With $[ZD_{H} = f_{i}(FS) AND$ relationship $[ZD_{H} = ZR_{H}]$ which can be transposed the computation of $F$ $ZD_{H} - ZR$ P(FS) = P(H)	$ZRH = f_2(FS)], t$ ] becomes [filled to read [ S involves soled H = O	he equil FS) = fz fi(FS) - fz ving the	Librium (FS)] (FS)=0] Rypression
plotting $ZD_H = f_i(FS)$ and $ZR_H = f_2(FS)$ the point of intersection or solving for FS to make $ZD_H - ZR_H = 0$	by an iterative procedu or analytically (on the includes (1) assuming (2) calculating allowat (3) calculating ka and c', ka and kp into ZDM plotting ZDM = fi(FS) a the point of interse to make ZDM - ZR,	ure, either grap (on puter). The a trial values of $k_p$ from $\phi'$ , and $ZR_H$ , and $md ZR_H = f_2$ ( ection or 1 = 0	whice $d$ e proced $\phi'$ and (4) sub- (4) sub- (5) 4 (5) 4	(by Rand) ure S, c', stituting zither for FS

WES FORM NO. 1259 REV 02* 1968 1259

16

* 24

Su B.	ECT	r;	E	XH	111	31	т	J						CON	PUT	EDI	BY:			-	DAT	E			FILE	NO	
s s		PL	EN	6 ^F	L ( A	50 N /	DU	U A Y S	LL	•				СНЕ	CKE	D 89	<b>۲</b> ۰					E .			SHEI	TN	ი. <b>4</b>
EPH - EDH	35.29	20.44	13.71		4.40	2,64	1.16	0.04	0.03	0.02	0.0	0,0	- 0.01	- 0.06				0 * V							م		بببه
۳P	42.60	27.75	21.01	11.41	01.11	9.95	8.11	9 A A	ጉ ዓ ገ	1.47	1.96	2P.F	7.95	9.91				. FS)							t	,	
HPc	20,39	11-21	8.56	5.37	3.91	3.07	2.53	2.15	2.15	2-14	2.14	2.14	2,13	2.12	0			(	0 •						Cress	04	
Hege	6.63	5,26	4.69	4.14	3.90	3.16	3.61	3.61	3.61	3-61	3. 6 I	3.61	3.60	3.60	1 A =		r o c , c <b>4</b>	5 L ) 'F	M M						inele		
HRc	10.36	1 6.9	5.18	3.45	2.59	2.07	1.13	1.4B	- <b>*</b> 0	1:48	<b>[+</b> ]	1.4-1	141	1.46	<u>ب</u> بر		-	- 	W						A Pro x	5	133VS
HR¢	5.21	3.48	2.61	14	1.30	1.04	6 8 J	0.14	0, 1 <b>4</b>	0,74	0.74	0.14	0.14	0.73	A52UA		ممانعهم	1020	÷ ₩						-		י ד ד סב
Z DH	1.31	1.31	1.31	7.31	1.31	1.3I	1.63	1.44	1.14	1.95	1.15	21.95	٦.96	1.97	. م بو		4	Solution	with				ł				
HAL	- 7.72	- 5.12	-3.53	-1.13	-0.13	11.0-	0.32	0.64	0.64	0.64	0.65	0.65	0.65	19.0	, vall		force	4.42	1.0+								
, Ú	0.100	0.467	0.350	0.233	0.175	0.140	L11 .0	0.100	0.100	001.0	0.100	0.099	0.099	0.099	1gat i Ve	,	.+.00	idere	". S			ł	ied fue				3 TRU
V Kp	1.43	1.27	1.20	1-13	014	1. C &	1.06	1.05	1.05	1.05	501	1.05	105	1.05	ε ti		• Por	puter	ded			/	ed oppl				
لا لا	2.04	1.62	1,44	1.21	1.20	91-1	1-13	11-1	114	11-1		÷		1	*		- <del>-</del>	نغ ع ر	9:40		/		D₁ = N				
k A	0.490	0.618	0.696	0.185	0.834	0. 865	0.886	0.901	101.0	201.0	0.902	0.402	0.902	0.403				$\sim$	$\backslash$		,		W		İ		
tam o'	0.364	0.243	0.182	0.121	1 60.0	0.013	190.0	0.052	0.052	0.052	0.052	0.052	0.052	1 500	1	_	<u></u>		/		1	1 1	11	   			-
\ ف	20.0	13.64	10.31	5.92	5,20	4.16	3.41	2.98	2.97	2.97	2.96	3.96	2.96	2.93		\$	<u> </u>	13	ي ب لا	эп	ק רו /	/sd	17	2	אייי ארנו		
٢S	0 	ŝ	2.0	3.0	4.0	ې م	د ور	0. -	1°L	201	7,0 <b>3</b>	1.04	3.05	1.10													
RES F	ORM	NO.	12	253																							

J**-9** 

SUBJECT	EXHIBIT J SAMPLE FLOODWALL	COMPUTED BY:	DATE	FILE NO.
	SLIDING ANALYSIS	CHECKED BY:	DATE:	SHEET NO.

The total base-foundation interface of the T-wall may not be in compression with the foundation. If any part of the surface under consideration is along the base-foundation interface and is not in contact with the foundation, this portion should be neglected when obtaining the effective base area to resist sliding. However, if the assumed failure surface is not along the base-foundation interface but through the soil, no reduction in the area to resist sliding is made.

A portion of the base of A T-wall will not be in compression when the resultant falls outside the kern thus creating A crack which can result in an increase in uplift pressures. This condition will affect the sliding stability Analysis when the Assumed sliding plane acts along the soil - structure interface below the base of the wall (for this condition the program will have to recycle back through the line of creep calculations until the creep path Assumptions match the final part of the base that is in contact

SES FORM NO. 1253

J-10



SUBJECT	EXHIBIT J SAMPLE FLOODWALL	COMPUTED BY	DATE	FILEN
	SLIDING ANALYSIS	CHECKED BY	DATE:	SHEETING

Another reasons the utility forces are not affected for this considition is because they are forces inside the soil-structure free body and therefore, do not affect the overall sliding stability.

The  $\phi$  and C values should be consistent with the material being sheared. A plane of failure through the soil should use the  $\phi$  and C of the soil. For any of the failure planes along the soil-structure interface, use the  $\phi$  and C for slidius frictions at the interface.



711.57

K-1



1.1.1

WES FORM NO. 1253

ST STATIST

К-2

SAMPLE OVER	TURNING ERETAIN	ANALYSIS	б адала та 1 1 1			Letter 40.	2					
CALCULATE	LINE OF	CREE	P HYDR		C PRE	SSURES						
Note that t		ec IV P		P. J.H	erent	Prom the						
NOTE AND	hese pie	1		TE and								
pressure shown in Exhibits H, L, and & because												
the assumed line of creep is in a different location												
(along the soil structure interface).												
1. Lost head = cheadwater - tailwater, total distance 9.5 # 1.12 max. total distance 35:04												
2. Potential head = cheadwater - tailwater) - lost Read												
= 9.5 - lost head												
		= rema	Ining	Seep	age he	rd.						
7 Partin	4.001	- Vort		chanc	a L	te Queda	-					
S. OSITION	nend	= 227 x		1 A A	<del>د ۲</del> ۵	I I	5					
		F1205	tive if	Felow	+ + 1.1	water						
4 Uplift	read	= puter	ntial h	eed +	posit	tion head						
s Uplift p	ressure	= Upli	ft Rea	d × i	ut. of	water						
		·					.					
LOLUMN	2	3	4	5	6	7						
POINT INCLEAN	path t tota	lost	potential head	Position	head	pressure ps-f						
1 0	0	0	9.50	- 9.50	0	0						
<b>z</b> 0	0	ο	9.50	1.00	10.50	656						
4 9.2	9.2	2.49	7.01	10.20	17.21	1075.4						
5 1.5	10.7	2.90	6.60	10.20	16.80	1050						
6 5.14	16.44	4.46	5.04	4.50	9.54	596.5						

#F" F GM N 1 125.5

٦

9

14.1

4.5

30.54

35.04

8.28

9.50

K<del>-</del> 3

1.22

0

N

4.50

0

5.72

0

357.5

0

1
C.W.C. ..... EXHIBIT K SAMPLE OVERTURNING ANALYSIS FOR FLOOD WALLS & RETAINING WALLS · . · . CALCULATE ACTIVE EARTH PRESSURE  $k_{A_{\phi}} = \tan^{2}(45^{\circ} - \phi/2) = \tan^{2}(45^{\circ} - 20^{\circ}/2) = 0.490$  $P_A = \chi H k_A - 2C \sqrt{k_A} = (125 - 62.5)(9.2)(0.49) - 2(100)(\sqrt{0.49})$ = 281.15 - 980 = -698.25The soil will not pull on the wall in the active state; therefore, let Pa = 0.

WES FORM NO. 1253

•

SUBJECT E	PARCT EXHIBIT K COMPUTED BY		DATE	FI	LE NO.
SAMPLE	OVERTURNING ANALYSIS				_
FORCE AN	ID HOHENT SUHHARY	FORC	E	HOHENT	MOMENT
Item	FACTORS (ft, 14)		Nt	ARH , FT	+ "0"
C,	之(1.5+2.0)(15.5)(150)		4069	6.6	26854
٢.	1.5 (16.3)(150)		3668	8.15	29890
٢3	<u></u>		299	14.57	4360
C.	1.5 (5.7) (150)		1283	15.55	19943
(0	NCRETE SUBTOTALS		9319	8.70	81047
WE,	$2(125) \frac{8.7 + 8.732}{2}$		2179	11.942	26022
WEZ	$3(125) \frac{5(6+5)648}{2}$		2109	2.812	5931
Ww	1015 (8.932) (62.5)		5730	11.93	68382
above E,	10、5(0、169)(之)(62、5)		5 <i>S</i>	7.512	413
H	281,75 (9,2)(2)	1296 0		2.633	3412)
	- 980.0 (9.2)	- 9016 J		1.100	-9917.6)
HIA	656 (10.5) ( ¹ / ₂ )	3444		- 7.0	-24108
HiB	(1075.4-656)(9.2)(之)	1929		2-633	5080
Hic	656 (9.2)	6035		1.1	6639
2	UBTOTAL HI	11408		-1.086	-12389
Hz	359.5 (4.5) (Z)	- 804		-1.5	1207
UH,	596.5 (5.7)	-3400	Į	2.85	- 9690
UH2	(1050 - 596,5)(5,7)(之)	-1293		3.80	-4911
50	BTOTAL UH	-4693		3.11	-14601
U,	359.5 (16.3)		-5827	8.15	- 47492
U₂	(596.5 - 357.5)(14.1)(之)		-1685	9.4	-15839
U ₂	239 (2.2)		-526	15.2	- 7992
U4	(1050 - 596.5)(1.5)	Į	-680	15.55	-10578
Us	453.5 (0.7)(之)		-159	14.567	- 2312
U.	(1075・4 -1050)(1・5)(之)		- 19	15.8	-301
50	BTOTAL U. + + U6		-8896	9.50	-84514
SUBTON	TAL F W/O passive Resistance	5911	10 4 9 6		11,498

¢

UBJECT: EYUTRIT V COMPUTED BY:	DATE:	FILE NO.
	_	
SAMPLE OVERIVENING ANALTSIS		
FOR FLOOD WALLS & RETAINING WALLS CHECKED BY:	DATE:	SHEET NO. 6
DISCUSSION OF OVERTURNING ASSUMPTIO	NS C GEN	ERAL)
a. For overturning of a walk with a b	n, four	ded
	<b>d</b>	
on either a soil or ruck foundat	ion the	
Porison tel penili Louisant Rouse to r	the force	о <del>т</del> н
horizonnia equility from force to	-0134 14	* 2 11
Prices with a 1 Particular 1	• D	
forces will not the subjected to	any ri	mits.
	0	
This assumption assumes the wall	is so	te l
	,	
against sliding; the assumption	is cha	cked
0	•	
later. If the wall is not safe a	gainst	sliding
	Q., , ,	U min
the decises is inadequate and a main	1 tricl	1.5.90.
the design is inhacturate and i her		or sider
		ρ
is started. The equilibriant force	e may	have
a a ta a a a a a a a a a a a a a a a a	U	•
a distribution corresponding t	0 cm.	of
	Q	•
the four options as discussed in	para	4.4.2.
	1.1.1	1. 1. 2.
However the dependence will read	Droccure	
nowever, the defined equilibrium.	PIESSOIE	-
diction & the Proposition of the		
aistribution for flood walks c	orrespon	ds to
· · · · · · · · · · · · · · · · · · ·		1
option a as shown in figure	K-18.	The
default equilibriant pressure dis	stributi	om
	-	
for retaining walls correspond t	to optic	m'c"
for the former of the former of the		
as shallon in Figure Kall For	L. DDwg	1.500
us showing in Fyria 11 10: For	me From	
averally an everal of page	1-12	d a
example, as presented on page	S K-14	thru
	1	•
K-13 of this Exhibit, the Untalance	d sum	of
Korizontal Forces (ZH = 5911 pou	nds) w	ill be
resisted & an esuibriant force	e in	
	-	
-		

WES FORM NO. 1253

۲

SUBJECT FXHIBIT K	COMPUTED BY	DATE	FILE NO.
SAMPLE OVERTURNING ANALYSIS			
FOR FLOOD WALLS & RETAINING WALLS	CHECKED BY	DATE	SHEET NO.
Accordance with distri	bution option "a	" of Figu	nre
K-1B. For the retaining	wall example opt	ioN "C" Fig	ure
K-1B, is used.			
b. For overturning of a	a wall with how	-izontal b	ase
And without a key (	see Figure K-2A	), founded	ON
either a soil or rock.	foundation, the ho	prizontal	
equilibrant force to res	ist the EH forces	5 will Not	be
subjected to any limits	. This Assumption	Assumes	the
wall is safe agains	t sliding; the A	assumption	IS
checked later. If the u	vall is not safe a	gainst slic	lina
the design is inadequat	e and a new t	rial desig	N
is started. The equilibrant force will be broken			
into two parts:			
(1) A force (friction And	lor cohesion) Alo	ng the ba	se of
the wall, and			
(2) A force due to passiv	e pressure distrik	oution Accu	ording
to options shown	IN Figure K-1A	•	
For a T-wall with A horizontal base, the force along the			
base of the wall is to be assumed to be mobilized			
first but will be limited to a maximum value of			
$EV$ tan $\phi$ + CL (see )	Figure K-2A of this	s Exhibit)	
		·	
ES FORM NO. 1253			]

К-7





WES FORM NO. 1253

SAMPLE OVERTURNING ANALYSIS FOR FLOOD WALLS & RETAINING WALLS CHECKED BY	CATE : ATE	FILE NO. SMEET N 10
If any additional force is require satisfy equikibrium, it will passive pressure (Figure K-IA) distribute with no limit. It is recognized without a key with an inclined	red to come uted that o trase	from well
falls into a "gray area" of des respect to overturning, since this lehave in similar fashion as a (Fig. K-1c) Similar fashion as a a keya. Therefore, for overturning s- an option will be provided in the which will permit the user to de type of wall in similar fashion wall with a key as described in "a" of this Exhibit or as a wa a key as described in "b" of For the wall without a leg ( inc wall ) to be analygod for ou as if the wall has a key the	sign wall wall tabilit e prog sign n as n para ll wit this Ex lined verturan user	with could with d, ram this a -graph tout chibit. base img would

REV OCT 1918 1253

SUBJECT EXHIBIT K	COMPUTED BY	DATE	FILE NO.
FOR FLOOD WALLS & RETAINING WALLS	CHECKED BY	DATE	SHEET NO.
Simply input a key len the desired result. If the wall friction and base walls, see Figure first, the maximum	(see Fr ath of 0.01 foot the user choo cohesive forces e K-2B) be value of this	gure K-10 The obtin oses to (Incline developed force wi	:) 1et d 11 be
Ntan p + CL			
IF any additional the equilibrium, it will consider the figure K-2E). The rest of this Expandicl sections. She illustrate the procedure for vetain proceedure for vetain.	force is require ome fron: passin distributed with whibit is divided neets K-12 th ore for flood K-16 illustra ing walls.	d to sat ve press th no lin inito t rough k walls te the	risfy ure nit. wo -13 and

NET FORM & TESS REVISE 1968



WES FORM NO. 1253 REV OCT 1968 1253

К-12



WES FORM NG. 1253



WES FORM NO. 1253



WES FORM NO. 1253 REV OCT 1968 1253

SUBJELT EXHIBIT K COMPUTED BY	DATE	FILE NO.
SAMPLE OVERTURNING ANALYSIS FOR FLOOD WALLS & RETAINING WALLS CHECKED BY	DATE	SHELT NO 16
The seepage line-of-creep calculations for overturn	ing should	
meglect any part of the T-wall be	se which	n is
not in contact with the foundation.	IF	pert
of the concrete fose is not in	contact	ωίτι
the foundation the program will have	to rec	y le
back through the line line - of - cree	p calcu	lations
Until the creep path assumptions m	atch th	e
final part of the base that is in	contac	t with
the foundation. If the toe-side fac	e of t	he
leg falls within the projection of	the base	2
which is not in contact with the	founda	tion
two options exist as follows:	1	
a) JP passive pressure is not use	d the t	oe-side
face of the key should be	Omit	led
from the creep path calculation	S.	
b) If passive pressure is used the	toe - s	ide
face of the key should not I	e Omi	thed
from the creep path calculations		
The base pressure to be used for stru	ctural	
design/analysis will be that develope	d by th	e
locding used in the overturning an	algsis	and
· as described in detail in Section	-9, para	9.1.1.
WES F HM N°. 1253		

L-1

Edulit L:  
Possive Resistance For T-Walls reached 2 provident Relation For Coulomb Theory #ETL 1110-2-184  
The theoretical resistance of a partice wedge is  
defined in ETL 1110-2-184 dated 25 Feb 76 as:  

$$P_{p} = W \tan(q \pi d) \neq \frac{sA}{\cos \alpha} (1 - \tan \alpha \tan \alpha)$$
This general termula is applicable to sliding along  
a place surface in rack or earth. This formula is di-  
-recting related to the Coulomb equation for passive  
eurith force as shown below.  

$$h = 2 \phi [Sinds$$

$$els for (Sinds) = \frac{S}{16} = \frac{S$$

L-2







MICROCOPY RESOLUTION TEST CHART NATIONAL PUBLIC STANDARD THE STANDARD STAND

i.

## SUPPLEMENTARY

## INFORMATION



DEPARTMENT OF THE ARMY WATERWAYS EXPERIMENT STATION, CORPS OF ENGINEERS P. O. BOX 631 VICKSBURG, MISSISSIPPI 39180

REPLY TO ATTENTION OF:

WESKD

24 January 1983

A 4097 76.0

SUBJECT: Replacement Sheets for WES Report, Program Criteria Specifications Document for Computer Program TWDA

TO: All Recipients of Subject Report

1. Those sheets containing pages 8-8, 8-9, and 8-10 should be replaced with the attached sheets containing identically numbered revised pages.

2. We hope that a set of these changes will be replaced into every copy of the original documents in your office. Additional copies of these changes or of the basic documents can be obtained through informal request to Mrs. Rosemary Peck, Engineering Computer Programs Library, FTS: 542-2581. a fully passive earth pressure due to an earthquake is computed as follows:

$$\Delta P_{pe} = \frac{\gamma H^2}{2} (\Delta K_{pe})$$

where

 $\Delta K_{pe} = K_{pe} - K_{p}$ where

 $\frac{\sin^2 (\alpha - \phi + \theta)}{\cos \theta \sin^2 \alpha \sin (\alpha + \delta + \theta)} \left[ 1 - \sqrt{\frac{\sin (\phi - \delta) \sin (\phi + \beta - \theta)}{\sin (\alpha + \delta + \theta)}} \right]^2$ K_{pe} =

and

 $K_p$  = static passive pressure coefficient for the fully passive case (see 4.1.4b)

For the case where passive earth pressure is used as a stabilizing force, a reduction in the passive earth pressure due to an earthquake acceleration is assumed at the same instant the fill pressure behind the structure is increased. If a reduction in  $K_p$  has been used for computing an effective K for the static case, this same reduction in  $\Delta K$  is used.  $\Delta P_{pe}$  is applied at 2/3H above the base. The pressure distribution of  $\triangle P_{pe}$  is the same as assumed for the active earth pressure condition.

At Rest Earth Pressure Conditions - The increase in an at rest 8.5.3 earth pressure due to an earthquake is approximated by the Mononobe-Okabe method. The change in the active earth pressure coefficient  $\Delta K_{ae}$  is first computed as described in paragraph 8.5.1, and then multiplied by the ratio  $K_r/K_a$  to obtain the change in the at rest earth pressure coefficient  $\Delta K_{re}$ . The change in at rest earth pressure is then computed as follows:

$$\Delta \mathbf{P}_{\mathbf{re}} = \frac{\gamma \mathbf{H}^2}{2} (\Delta \mathbf{K}_{\mathbf{re}})$$

Only the horizontal component of the earthquake acceleration is considered.  $P_{re}$  is applied at two thirds the height of the fill above the base. The pressure distribution of  $\Delta P_{re}$  is the same as assumed for the active earth pressure condition.

8-7

## 8.6 LATERAL COHESIVE EARTH PRESSURES DUE TO EARTHQUAKES

The computation of the dynamic earth pressure for cohesive soils is beyond the scope of this computer program. A nonlinear finite element analysis to account for inelastic strains in the soil could possibly be used for critical cases.

٠.

## 8.7 WATER PRESSURE DUE TO EARTHQUAKES

8.7.1 Method - If the backfill over the heel is saturated to some level, the increase of soil force over the heel is determined by the Mononobe-Okabe theory described in paragraphs 8.5 and 8.5.1, using the saturated unit weight for submerged earth. When water exists above the soil top surface, the increase of force of the water above the soil top surface is computed by the Westergaard theory described in paragraph 8.7.2. Earthquake on water above the soil top surface on the heel side causes an increase in the total force when the acceleration is positive, a decrease when the acceleration is negative. Earthquake on water above the soil top surface on the toe side causes a decrease in the total force when the acceleration is positive and an increase when the acceleration is negative. The Westergaard theory yields a parabolic shape to the added-pressure diagram. All forces due to earth on the toe side of the stem are computed as a passive reaction to the summation of all other forces. A replacement figure 8-3 for page 8-10 is attached.

The heel-side increase in soil forces due to earthquake is placed in array EH for use in stability calculations and in array EHS for use in stem stress analysis, after the inversion shown in diagram (4) in Figure 8-3. The net dynamic pressure diagram of the heel-side Westergaard water pressures (a parabola) summed algebraically with the toe-side dynamic pressures (another parabola) is placed in array EFH. See Chapter 11 in the User's Reference Manual for more detail on these arrays and how the user can modify the computed values or substitute his own.

8.7.2 Westergaard Theory - By the Westergaard theory, the dynamic water pressure down to depth y below the surface for a total water depth h [D] is expressed by Equation 3 on page 5 of EM 1110-2-2200 as

$$P_{e_2} = \frac{2}{3} C_e^{\alpha y} + \overline{y}$$

The additional moment at depth v due to  $P_{e_2}$  is given by

$$M_e = \frac{4}{15} C_e xy^2 \sqrt{hy}$$

with

$$C_e = \sqrt{\frac{51}{1 - 0.72 \left(\frac{h}{1000 t_e}\right)^2}}$$

۰.

where g is acceleration of gravity  $(32.2 \text{ ft/sec}^2)$ .

**Sister**s



· .



8-10

- 04

Sec.

1.25

